

TUMWATER GENERAL GOVERNMENT COMMITTEE MEETING AGENDA

Online via Zoom and In Person at Tumwater City Hall, Council Conference Room, 555 Israel Rd. SW, Tumwater, WA 98501

Wednesday, July 09, 2025 8:00 AM

- 1. Call to Order
- 2. Roll Call
- <u>3.</u> 2025 Comprehensive Plan Periodic Update 2025 Development Code Update (Community Development Department)
- 4. Amended 5901 and 5917 Black Lake-Belmore SW 10 Percent Annexation Petitions (TUM-25-0453 and TUM-25-0454) (Community Development Department)
- Resolution No. R2025-011, 93rd Avenue SW and Case Road SW Annexation (TUM-25-0128) (Community Development Department)
- 2025 Comprehensive Plan Periodic Update Conservation Element (Community Development Department)
- 7. Additional Items
- 8. Adjourn

Meeting Information

All committee members will be attending remotely. The public are welcome to attend in person, by telephone or online via Zoom.

Watch Online

https://us02web.zoom.us/j/83025721090?pwd=kW3lUagmTxbUXnc0svG26J5Tanglvg.1

Listen by Telephone

Call (253) 215-8782, listen for the prompts and enter the Webinar ID 830 2572 1090 and Passcode 731678.

Public Comment

The public may submit comments by sending an email to <u>council@ci.tumwater.wa.us</u>, no later than 5:00 p.m. the day before the meeting. Comments are submitted directly to the Committee members and will not be read individually into the record of the meeting.

Post Meeting

Video of this meeting will be recorded and posted on our City Meeting page: <u>https://tumwater-wa.municodemeetings.com</u>.

Accommodations

The City of Tumwater takes pride in ensuring that people with disabilities are able to take part in, and benefit from, the range of public programs, services, and activities offered by the City. To request an accommodation or alternate format of communication, please contact the City's ADA Coordinator directly, call (360) 754-4129 or email <u>ADACoordinator@ci.tumwater.wa.us</u>. For vision or hearing impaired services, please contact the Washington State Relay Services at 7-1-1 or 1-(800)-833-6384.

TO:	General Government Committee
FROM:	Erika Smith-Erickson, Housing and Land Use Planner
DATE:	July 9, 2025
SUBJECT:	2025 Comprehensive Plan Periodic Update – 2025 Development Code Update

1) <u>Recommended Action</u>:

No action requested. This is an opportunity for a discussion about the draft amendments to meet the state requirements regarding certain development regulations and Design Review processes due December 31, 2025, and next steps.

2) <u>Background</u>:

On a ten-year cycle, the City must conduct a Growth Management Act periodic update of its Comprehensive Plan and related development regulations. For the current cycle, the City is required to complete work on the periodic update by December 31, 2025.

<u>2025 Comprehensive Plan Update | City of Tumwater, WA</u> has links to guidance material and information about the update.

The intent of this General Government Committee work session is to discuss the City's approach to meeting the state requirements for development code updates throughout Tumwater Municipal Code and Design Review.

3) <u>Policy Support</u>:

Goal LU-1: Ensure the Land Use Element is implementable and coordinated with all applicable City plans and the plans of other jurisdictions in the Thurston region.

Policy LU-1.9: Ensure consistency between the Land Use Element and the Tumwater Zoning Code.

4) <u>Alternatives</u>:

None.

5) Fiscal Notes:

This is primarily an internally funded work program task. The City used a Washington State Department of Commerce grant for \$75,000 to have consultant help staff to address the new state middle housing requirements.

- 6) <u>Attachments</u>:
 - A. Draft table for the 2025 Development Code Update Ordinance No. 02025-011

DUE WITH THE UPDATE BY DECEMBER 31, 2025:

- HB 1220 (2021) State-Projected Housing Needs and must allow emergency shelters, transitional housing, emergency housing and permanent supportive housing (PSH) (STEP) (RCW 36.70A.020, RCW 36.70A.070, RCW 35A.21.430, RCW 35A.21, and RCW 36.70A.390) COMPLETE. Addressed by Ordinance No. O2021-019 (1/18/22) and by Housing Element update (2025).
- ESSHB 1799 (2022) Organic materials management. (RCW 70A.205.040, RCW 70A.205.015, and others) COMPLETE. Addressed by Ordinance No. 02022-019 (12/6/22).
- SHB 1717 (2022) Tribal engagement NO DEVELOPMENT CODE CHANGES. Addressed in part by Thurston County update to County-Wide Planning Policies (2025).
- HB 1220 --- NO DEVELOPMENT CODE CHANGES. Addressed
- Bonus density on land owned by religious organizations (RCW 36.70A.545)
- May not regulate number of people in a household (RCW 35.21.682, RCW 35A.21.314, RCW 36.01.227)
- Limits on regulating homeless encampments on religious property (RCW 36.01.290, RCW 35.21.915, RCW 35A.21.360)
- Limit on amount of parking that may be required near transit (RCW 36.70A.620, ESSB 5184)
- Parking changes (RCW 36.70A.622)
- Allow 2 ADUs within urban growth areas (RCW 36.70A.680-2, HB 1337)
- Allow middle housing in 77 cities (RCW 36.70A.635, HB 1110)
- Clear objective design standards (RCW 36.70A.630 and 36.70B, HB 1293)

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
HB 5558 (Chapter 148, Laws of 2025) RCW 36.70A.130, RCW 36.70A.630, RCW 36.70A.635, RCW 36.70A.680,	Amendments to timelines for comprehensive plan and development regulations	 Moves the deadlines for some of the required development code updates up from June 30, 2026, to December 31, 2025. 	December 31, 2025
E2SHB 1110 RCW 36.70A.030, 36.70A.280, 43.21C.495, 43.21C.450, 36.70A, RCW; chapter 64.34 RCW; 64.32 RCW; 64.38 RCW; 64.90 RCW;.	Increasing middle housing in areas traditionally dedicated to single-family detached housing.	 At least 2 units per lot in residential zones (except lots <1,000 sf), unless higher-density zoning applies. At least 4 units per lot in residential zones (except lots <1,000 sf), unless higher-density zoning applies, if at 	December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 least one unit is affordable housing. (Note: it does not appear Tumwater has any major transit stops by the bill's definition) As an alternative to first two bullets above, meet their density requirements on 75% of city lots that are primarily dedicated to single-family. Also meet criteria for the other 25%. Allow at least 6 of the 9 types of middle housing in residential zones. Allow zero lot line short plats Limit design review for middle housing to administrative process and not apply any development standards that do not apply to single-family houses. Limit parking requirements for middle housing to 1 space on lots <6,000 sf and 2 spaces on lots>6,000 sf. Option to submit transportation safety study to Commerce. Actions on all the above bullets are exempt from appeal under SEPA Option for Commerce approval of 'substantially similar' plans and regulations to those required in this bill Option for Commerce to provide extension of timeline if will result in displacement or overburdened 	

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 infrastructure. CFP update can also be delayed if extension is granted. Actions to remove parking requirements for infill development in a UGA categorically exempt from SEPA Common Interest Communities (e.g., condominium or homeowners' associations) cannot prohibit implementation of this bill 	
HB 1220 (Chapter 254, Laws of 2021) RCW 36.70A.020, RCW 36.70A.070, RCW 35A.21.430, RCW 35A.21, and RCW 36.70A.390 ADU regulations in Section 7 of HB 1220 vetoed by Governor	State-Projected Housing Needs Emergency shelters, transitional housing, emergency housing and permanent supportive housing (PSH) (STEP) Indoor emergency Shelter, Transitional housing, Emergency housing	 Requires update to Housing Element to address: Housing allocation by income group Displacement analysis Reviewed definitions in Title 18 - emergency housing, emergency shelter, extremely low-income household, moderate income household, permanent supportive housing. Allow PSH and transitional housing in zones that allow residential uses and hotels Allow emergency housing and shelter in zones that allow hotels or allowed within at least 50% of zones within a mile of transit Siting restrictions for public health and safety Reasonable occupancy, spacing, and intensity of use requirements may be imposed to protect public health and safety. 	Completed. City Addressed by Ordinance No. O2021-019 (1/18/22) December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 Restrictions must still allow sufficient capacity to accommodate projected needs for these types of units (under RCW 36.70A.070(2)(a)(ii)). Consult with local STEP providers Write regulations that make it clear how housing providers can demonstrate consistency. Consider including provisions for alternative housing types. 	
RCW 36.70A.545	Bonus density on land owned by religious organizations	 Must allow an increased density bonus consistent for any affordable housing development of any single-family or multifamily residence located on real property owned or controlled by a religious organization provided that: The affordable housing is for or occupied exclusively by low-income households. The affordable housing development is part of a binding obligation that requires the development to be used exclusively for affordable housing for at least fifty years, even if the religious organization no longer owns the property. Housing does not discriminate or violate the federal fair housing amendments act of 1988. The religious organization development must pay 	December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
RCW 35.21.682, RCW 35A.21.314, RCW 36.01.227	May not regulate number of people in a household	 all fees, mitigation costs, and other charges required through the development of the affordable housing development. "Low-income household" means a single person, family, or unrelated persons living together whose adjusted income is less than eighty percent of the median family income, adjusted for household size, for the county where the affordable housing development is located; and TMC 5.75.010: "Dwelling unit" is defined under RCW <u>59.18.030</u>, as it exists or is hereinafter amended, and means a structure or part of a structure used as a home, residence, or sleeping place by one, two, or more persons maintaining a common household, including, but not limited to, single- family residences and multiplexes, apartment buildings, and manufactured or mobile homes. 	December 31, 2025.
RCW 36.01.290, RCW 35.21.915, RCW 35A.21.360	Limits on regulating homeless encampments on religious property	Must allow a density bonus (based on local needs) for affordable housing & STEP Few general guidelines Must allow hosting of the homeless	December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		Statute includes detailed provisions for local regulation	
ESSB 5184 RCW 36.70A.620 RCW 36.70A.622	Limit on amount of parking that may be required near transit	 Standards for specified housing types near transit Housing for seniors and people with disabilities, housing affordable to very- or extremely-low income households, and market-rate multifamily housing Development regulations related to parking area surfacing, design, type, and size Garages and carports may not be required Stalls may not be required to exceed 8' x 20' Must allow tandem parking spaces to count toward required parking 	December 31, 2025.
RCW 36.70A.622	Parking changes	 Garages and carports may not be required as a way to meet minimum parking requirements for residential development Parking spaces may not be required to exceed eight feet by 20 feet, except for required parking for people with disabilities Any county planning under this chapter, and any cities within those counties with a population greater than 6,000, may not require off-street parking as a condition of permitting a residential project if 	December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 compliance with tree retention would otherwise make a proposed residential development or redevelopment infeasible; and Parking spaces that consist of grass block pavers may count toward minimum parking requirements. Existing parking spaces that do not conform to the requirements of this section by June 6, 2024, are not required to be modified or resized, except for compliance with the Americans with disabilities act. Existing paved parking lots are not required to change the size of existing parking spaces during resurfacing if doing so will be more costly or require significant reconfiguration of the parking space locations. 	
HB 1337 RCW 36.70A.680-2	Allow 2 ADUs within urban growth areas	Must allow conversion of existing structures, even if nonconforming to current setback or lot coverage requirements. RCW 36.70A.681(1)(j) •May not require the owner to occupy the property •Must allow an ADU of at least 1,000 square feet •Parking requirements based on distance from transit and lot size •May not charge more than 50% of impact fees charged for the principal unit	December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 May not prohibit sale as independent units May be restricted in: Within areas designated as critical areas In zones with very low densities within UGAs In areas designated to protect a critical area In shoreline areas designated under a shoreline master program On lots in a watershed used for potable water Within a mile radius of SeaTac airport Where no sewer service exists Where prohibited by HOA covenants & restrictions 	
		 ADU: Can adopt regulations ADU's cannot be used for short term rental. Minimum gross floor area (1,000 sq. feet) Must allow roof decks, balconies, and porches may encroach into setbacks must apply abutting lot setbacks to accessory dwelling units on lots abutting zones with lower setback requirements 	

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 must establish an amnesty program to help owners of unpermitted accessory dwelling units to obtain a permit Must allow ADU's on lots that meet minimum size requirements for principal unit Must allow conversion of nonconforming structures 1,000-1,200 sq or 60% of principle structure for min size Cannot require public street improvements May require a new or separate utility connection when necessary to be consistent with water availability/water system plans – if so connection fees and charges must be proportionate to the burned of the proposed accessory dwelling unit on the utility system. (sewer?) 	
HB 1293 RCW 36.70A.630 and 36.70B	Clear objective design standards- Design review process	 Only clear and objective development regulations governing the exterior design of new development. Must be clear and easy to determine if a building design is permissible May not result in a reduction in density, height, bulk, or scale below the generally applicable development regulations for a 	December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 development proposal in the applicable zone. 	
HB 1998 RCW 36.70A.535	Allow co-living where 6 units per lot are allowed	Update language and reference for co- living. Currently viewed as rooming houses. Update definitions. A residential development with sleeping units that are independently rented and lockable and provide living and sleeping space. Residents share kitchen facilities with other sleeping units in the building. Also called: Single-room occupancy, rooming/boarding house, residential suites •Allow co-living in areas that allow 6 or more residential units This includes mixed-use zones, <i>but not</i> <i>areas that allow 6 units only after an</i> <i>incentive, such as an affordable housing</i> <i>bonus</i> •Require no more than 0.25 off-street parking spaces per unit No off-street parking within ½ mile walking distance of a major transit stop •Require standards and review and notice procedures that are no more restrictive than those for similar residential uses in the same zone	December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 This includes design standards, shoreline regulations, and requirements for public meetings •May not require co-living housing to have room dimensions larger than required by building code, provide a mix of unit sizes or bedrooms, or to include other uses •May not exclude co-living housing from affordable housing incentive programs •Must treat a sleeping unit as one-quarter of a dwelling unit (or less) when calculating unit density •Must treat a sleeping unit as one-half of a dwelling unit (or less) when calculating fees for sewer connections 	
RCW 36.70A.635	Minimum Residential Density	Min zoning: 2 du/lot If within ¼ mile walking distance of major transit stop, allow at least: 4 du/lot Affordability bonus, allow at least: 4 du/lot, if at least one (1) is affordable	December 31, 2025.
SSB 5818 (Chapter 246, Laws of 2022) WAC 197-11-800(1)(b), WAC 197-11- 800(1)(d), and others.	SEPA Amendments	 Added an exemption for the construction or location of four <u>attached or</u> detached single family residential units to WAC 197-11-800(1)(b). Changed Flexible Exemption Thresholds in WAC 197-11-800(1)(d) for: Single family residential project types with a total square 	December 31, 2025.

Bill / RCW	Short Description	Effect on Tumwater	Deadlines
		 footage of fewer than 1,500 square feet - 100 Units Multifamily residential - 200 units Requires outreach to WSDOT 	

Notes:

*Model ordinances: Preempt/supersede local development regulations if a city has not passed HB 1110 development regulations by the due date and remains in effect until the city takes all actions to implement RCW 36.70A.635.

TO:General Government CommitteeFROM:Brad Medrud, Community Development DirectorDATE:July 9, 2025SUBJECT:Amended 5901 and 5917 Black Lake-Belmore SW 10 Percent Annexation Petitions
(TUM-25-0453 and TUM-25-0454)

1) <u>Recommended Action</u>:

Place the amended 10 Percent Annexation Petition on the July 15, 2025, City Council consideration calendar for the City Council to decide whether they will accept the proposed annexation and whether the City Council will require the assumption of existing City indebtedness by the area to be annexed.

2) <u>Background</u>:

The owner of a property currently in Tumwater's urban growth area in unincorporated Thurston County filed petition applications under Chapter 35A.14 RCW for annexation to the City on April 25, 2025, that were deemed complete on April 29, 2025, after payment of the necessary review fees.

The applicant had submitted petition applications for two parcels and paid the necessary review fees. City staff inadvertently left the petitions for the second parcel out of the May 14, 2025, General Government Committee packet and June 3, 2025, City Council packet, so this item contains the complete application package for both parcels.

Because the property owner owns all the property included in the annexation area, they filed petitions representing both the ten percent and sixty percent of the value of the properties proposed for annexation.

The property owner has asked that their property be annexed into the City of Tumwater and indicated their willingness to assume their fair share of the City's indebtedness and are willing to accept the City's Comprehensive Plan.

3) Policy Support:

Goal LU-2: Ensure development takes place in an orderly and cost-efficient manner in order to best utilize available land and public services, conserve natural resources, protect critical areas, preserve open space, and reduce sprawl.

Policy LU-2.4 Ensure new annexations adhere to the goals and policies of the City's Annexation Policy.

- 4) <u>Alternatives</u>:
 - None.
- 5) <u>Fiscal Notes</u>:

None.

6) <u>Attachments</u>:

- A. Staff Report
- B. Presentation
- C. 5901 Black Lake-Belmore Road SW Ten and Sixty Percent Petitions
- D. 5901 Black Lake-Belmore Road SW Application Exhibits
- E. 5917 Black Lake-Belmore Road SW Ten and Sixty Percent Petitions
- F. 5917 Black Lake-Belmore Road SW Application Exhibits
- G. Black Lake-Belmore Road SW Application Map

STAFF REPORT

Date: May 5, 2025To: General Government CommitteeFrom: Brad Medrud, Deputy Community Development Director



6

5901 & 5917 Black Lake-Belmore SW Ten Percent Annexation Petitions (TUM-25-0453 and TUM-25-0454)

The owner of the properties at 5901 and 5917 Black Lake-Belmore Road SW currently in Tumwater's urban growth area in unincorporated Thurston County is requesting to be annexed into Tumwater. The property owner filed ten percent and sixty percent annexation petition applications with the City on April 25, 2025, which were deemed complete on April 29, 2025, after payment of the necessary application review fees.

The property owner has indicated a willingness to assume their fair share of the City's indebtedness and is willing to accept the City's Comprehensive Plan. Because the property owner owns all the property included in the annexation area, they filed annexation petitions representing both ten percent and sixty percent of the value of the properties proposed for annexation.

The applicant had submitted petition applications for two parcels and paid the necessary review fees. City staff inadvertently left the petitions for the second parcel out of the May 14, 2025, General Government Committee packet and June 3, 2025, City Council packet, so this item contains the complete application package for both parcels.

At their meeting on July 9, 2025, the General Government Committee will be asked to place the complete ten percent annexation petitions on the July 15, 2025, City Council consideration calendar for the City Council to decide whether it will accept the proposed annexation and whether the City Council will require the assumption of existing City indebtedness by the area to be annexed.

Contents

1 – Background	<u>)</u>
2 – Growth Management Act and Annexations	1
3 – General Guidelines for Evaluating Proposed Annexations	1
4 – Sixty Percent Annexation Method Process	5
A. Initiation of the Sixty Percent Petition Annexation	

5901 & 5917 Black Lake-Belmore SW Ten Percent Annexation Petitions (TUM-25-0453 and TUM-25-0454)

В.	Meeting on the Annexation Proposal	6		
C.	Contents of the Petitions	6		
D.	Filing of Sixty Percent Petitions	7		
E.	Legal Sufficiency of the Petitions	7		
F.	Hearing on Petitions	7		
G.	Decision	8		
Н.	Review by the Boundary Review Board	8		
١.	Effective Date of Annexation	8		
J.	Notice of Annexation	9		
5 – St	aff Recommendation and Request9			
Appendix A – Draft Property Tax Comparison11				

<u>1 – Background</u>

As a non-charter Code City, Tumwater follows the annexation processes outlined in Chapter 35A.14 RCW. Cities such as Tumwater that plan under the Growth Management Act may annex only property that is located within their designated urban growth areas.

The two parcels in the proposed annexation area are located northwest of the intersection of Black Lake-Belmore Road SW and 60th Avenue SW and are 10.75 acres in size. The proposed annexation area is west of the current city limit and has a Single Family Low Density Residential district designation:

1. The first annexation petition is for the property at 5901 Black Lake-Belmore Road SW, Thurston County Assessor's parcel number 12705220200, which is approximately 9.72 acres and is owned by Terrence Hess. The red line is the current City limits.



2. The second annexation petition is for the property at 5917 Black Lake-Belmore Road SW, Thurston County Assessor's parcel number 12706110201, which is approximately 1.03 acres and is owned by Terrence Hess. The red line is the current City limits.



3

The properties are undeveloped.

The ten percent annexation petitions request that the City Council set a date not later than sixty days after the filing of the ten percent annexation petitions for a meeting with the property owner to determine:

- Whether the City Council will accept, reject, or geographically modify the proposed annexation;
- Whether the City Council will require adoption of a comprehensive plan for the area to be annexed [This has been completed already through the adopted Tumwater Thurston County Joint Plan by the City and County.]; and
- Whether the City Council will require the assumption of existing City indebtedness by the area to be annexed.

2 – Growth Management Act and Annexations

The Growth Management Act (GMA) establishes the framework for Tumwater's annexation authority. A major goal of GMA is to reduce urban sprawl by encouraging development in urban areas where adequate public facilities already exist or where such facilities can be more efficiently provided (RCW 36.70A.020(1) and (2)). To help implement this goal, GMA requires that counties designate urban growth areas "within which urban growth shall be encouraged and outside of which growth can occur only if it is not urban in nature" (RCW 36.70A.110(1)).

Tumwater's urban growth area includes land sufficient to accommodate the 20-year population growth projected for the City. In addition, GMA states that "it is appropriate that urban government services be provided by cities, and urban government services should not be provided in rural areas" (RCW 36.70A.110(3)). Consistent with the goal of controlling the spread of urban growth, GMA limits the territory Tumwater may annex to that which lies within its urban growth area (RCW 35A.14.005). The Tumwater Thurston County Joint Plan provides the necessary Comprehensive Plan policies dealing with land use, housing, capital facilities, utilities, and transportation in Tumwater's urban growth area.

3 – General Guidelines for Evaluating Proposed Annexations

The Municipal Research and Services Center of Washington prepared the updated guidebook *Annexation by Washington Cities and Towns* in 2020.

The Guidebook suggested consideration of the following basic principles when considering an annexation to ensure that the chances of a successful completion of the annexation would be increased and the effect upon the City would be positive.

• The boundaries of the annexation area should be drawn in accordance with the ability (both from a geographic and economic standpoint) of the city to provide services. The need for services should be taken into account.

4

- The population and assessed valuation of the area should be sufficient to allow the area to pay its fair share of the cost of providing services.
- The area should contribute to the logical growth pattern of the city and should encourage orderly growth. Where possible, irregular boundaries should be avoided.
- It should be no larger than what the city is able to service adequately with capital improvements and services within a reasonable time.
- The area should be adaptable to anticipated expansion requirements of the city for residential or commercial/industrial purposes.
- The boundaries of an area should be drawn to include residents who are generally favorable toward annexation or where annexation can be demonstrated to be advantageous to the residents of both the fringe area and the city.
- In drawing boundaries of an annexation area, due regard should be given to special districts in the area (see Special Districts).

4 – Sixty Percent Annexation Method Process

The most frequently used method of annexing unincorporated territory is by petition of the owners of at least sixty percent of the property value in the area, computed according to the assessed valuation of the property for general taxation purposes. The steps in the sixty percent annexation method process are as follows:



A. Initiation of the Sixty Percent Petition Annexation

Under RCW 35.14.120, prior to circulating a petition for annexation, the owners of property representing not less than ten percent of the assessed value of the property for which annexation is sought must give written notice to the City Council of their intention to commence annexation proceedings. This process was completed on April 29, 2025, when the ten percent petition application was deemed completed after the applicant paid the necessary application review fees.

B. Meeting on the Annexation Proposal

Following RCW 35A.14.120, after being notified of the proposed annexation, the City Council is to set a date not later than sixty days after the filing of the ten percent petitions for a meeting with the initiating parties to determine:

- Whether Tumwater will accept, reject, or geographically modify the proposed annexation;
- Whether it will require the simultaneous adoption of a proposed zoning regulation, if such a proposal has been prepared and filed (as provided for in RCW 35A.14 330 and 35A.14.340); and
- Whether it will require the assumption of all or any portion of existing City indebtedness by the area to be annexed.

This meeting with the City Council is proposed for June 3, 2025. If the City Council requires the adoption of a proposed zoning regulation and/or the assumption of all or any portion of indebtedness as conditions to annexation, it is to record this action in the minutes of the meeting. City Council acceptance of the proposed annexation is a condition precedent to circulation of the petitions. There is no appeal from the City Council's decision.

C. Contents of the Petitions

Under RCW 35A.14.120, if the City Council accepts the initial annexation proposal, the sixty percent petitions may be drafted and circulated. The sixty percent petitions must:

- Describe the property according to government legal subdivisions or legal plats.
- Be accompanied by a map that outlines the boundaries of the property sought to be annexed.
- If the City Council has required the assumption of all or any portion of City indebtedness and/or the adoption of a proposed zoning regulation for the area to be annexed, set forth these facts clearly, together with a quotation of the minute entry of that requirement.
- Be signed by the owners of not less than sixty percent of the assessed value of the property for which annexation is petitioned.
- Comply with the rules for petitions in RCW 35A.01.040 (RCW 35A.14.130).

The applicant has submitted complete sixty percent annexation petitions. So if the City Council approves the ten percent petitions, then this step will have been completed.

D. Filing of Sixty Percent Petitions

The completed sixty percent annexation petitions are to be filed with the City Council (RCW 35A.14.120). Although there is no time limit specified in the annexation statutes as to when a petition needs to be filed with the City Council after it has begun circulating for signatures, the signatures on a petition are valid only if signed no later than six months prior to the filing date. Any signatures older than six months are to be stricken from the petition by the officer certifying petition sufficiency (RCW 35A.01.040(8)).

The applicant has submitted a complete sixty percent petitions. So if the City Council approves the ten percent petitions, then this step will have been completed.

E. Legal Sufficiency of the Petitions

The annexation petitions must be certified as sufficient (i.e., as having valid signatures representing the required sixty percent of property value). Within three working days of the filing of the sixty percent petitions, the officer with whom the petitions are filed must transmit the petitions to the county assessor, who makes the determination of the sufficiency of the petitions. The county officer whose duty it is to determine petition sufficiency must file with the officer receiving the petitions for filing a certificate stating the date the determination of sufficiency was begun. The officer determining petition sufficiency must do so "with reasonable promptness" (RCW 35A.01.040(4)).

This will be the next step if the City Council approves the ten percent petitions.

F. Hearing on Petitions

Under RCW 35A.14.130, when legally sufficient annexation petitions are filed, the City Council may consider it and:

- Fix a date for a public hearing, and
- Provide notice specifying the time and place of the hearing and inviting interested people to appear and voice approval or disapproval of the annexation. The notice is to be:
 - o Published in one or more issues of a newspaper of general circulation in the City; and
 - Posted in three public places within the territory proposed for annexation.

There are no statutory requirements concerning the actual hearing, other than to give proponents and opponents an opportunity to speak.

G. Decision

Because Tumwater may not annex territory without prior Thurston County Boundary Review Board approval, an annexation ordinance passed following a hearing but before board review cannot yet be effective. Consequently, the City Council will be asked to approve a resolution of intent to annex. If the review board approves the annexation, the City Council will adopt the formal annexation ordinance.

H. Review by the Boundary Review Board

Under RCW 36.93.090 and .100, Tumwater must file a "notice of intention" with the Thurston County Boundary Review Board within 180 days of when the annexation is proposed. For the purposes of sixty percent annexation petition, an annexation is "proposed" when the annexation petition is filed with the Tumwater.

If the proposal includes annexation of the territory of a fire district, in this case the McLane Black Lake Fire District, Tumwater must provide notice to the district of the proposed annexation simultaneously when notice of the proposed annexation is provided to the Thurston County Boundary Review Board.

The board may assume jurisdiction over the annexation if, within forty-five days of filing the notice of intention, a request for review is made by:

- Thurston County or any other affected governmental unit; or
- Petition of registered voters or property owners.

If jurisdiction is not invoked within forty-five days, the proposed annexation is deemed approved by the Thurston County Boundary Review Board.

The board must act within 120 days of the review request unless the board and the annexation initiators agree to an extension. If no decision is made within 120 days and no extension is granted, the proposal is deemed approved.

I. Effective Date of Annexation

The annexation, together with any provision relating to application of a proposed zoning regulation, is effective on the date fixed in the annexation ordinance. The relevant statute, RCW 35A.14.150, does not specify any date by which the annexation must be made effective. Note, however, that there are important timing issues as to when an annexation occurs with respect to when the City's property tax levy can be effective in the newly annexed area and with respect to receipt of state-shared revenues, sales tax, and, if applicable, sales tax equalization payments.

J. Notice of Annexation

1) Notice to State (Office of Financial Management Certification)

Tumwater must submit an annexation certificate and additional supporting documents to the state Office of Financial Management within thirty days of the effective date of annexation specified in the annexation ordinance (RCW 35A.14.700).

The Office of Financial Management files the approved annexation certificates on a quarterly basis. Filing dates are the last working days of November, February, May, and August. Annexations are not approved and filed until all of Office of Financial Management's requirements are met. Revenues may be lost as a result of problems in the certification process because revenue distributions are not backdated.

2) Notice to County, Light and Power and Gas Distribution Businesses, and, Where Applicable, to Fire Protection and Library Districts

At least sixty days before the effective date of the annexation, Tumwater is required by RCW 35A.14.801 to provide to the county treasurer and assessor and to light and power and gas distribution businesses, by certified mail or electronic means, notice of the annexation that includes a list of annexed parcel numbers and street addresses.

If Tumwater annexes territory within a fire district, which in this case is the McLane Black Lake Fire District, it is required to provide the same notice to the district. The county treasurer is required to remit to the City only those road taxes and, where applicable, fire district and library district property taxes collected sixty days or more after receipt of the notice. Light and power businesses and gas distribution businesses are only required to remit to the City those utility taxes collected sixty days or more after receipt of the notice.

3) Notice to Department of Revenue

Sales tax changes may take effect only on January 1, April 1, or July 1. The term "sales tax changes," for purposes of this legislation, includes changes resulting from annexation (RCW 82.14.055(4)). The City must provide notice to the Department of Revenue at least seventy-five days before the change takes place.

4) Notice to City Departments.

Although any annexation will impact some City departments more than others, all will be advised of the annexation.

5 – Staff Recommendation and Request

- Accept the proposals to annex, with no modifications.
- Do not propose any amendments to the existing land use designations or zone districts.
- Require the annexed area to assume City indebtedness.

• Ask the General Government Committee to place the ten percent annexation petitions on the July 15, 2025, City Council consideration calendar for the City Council to discuss with the applicant.

Appendix A – Draft Property Tax Comparison

Property Tax Rates are per \$1,000 of the assessed value of your property	Tax Rate for your area	Taxes for \$500,000 home - Based on Average Value for your area	Tax Rate after annexation to City of Tumwater	Taxes for \$500,000 home after annexation
County Road Tax	\$0.72	\$360.00	\$-	\$-
McLane Black Lake Fire Dept.	\$1.42	\$710.00	\$-	\$-
McLane Black Lake Fire Dept. Excess Levy	\$0.77	\$385.00	\$0.77	\$385.00
City of Tumwater General Tax	\$-	\$-	\$1.75	\$875.00
Tumwater Metropolitan Park District	\$-	\$-	\$0.49	\$245.00
Public Utility District #1	\$0.01	\$5.00	\$0.01	\$5.00
Medic One	\$0.13	\$65.00	\$0.13	\$65.00
Medic One Levy Lid Lift	\$0.21	\$105.00	\$0.21	\$105.00
Port of Olympia	\$0.13	\$65.00	\$0.13	\$65.00
Timberland Regional Library	\$0.23	\$115.00	\$0.23	\$115.00
Tumwater School District Excess Levy	\$3.88	\$1,940.00	\$3.88	\$1,940.00
County General Tax	\$0.85	\$425.00	\$0.85	\$425.00
State General Tax	\$2.26	\$1,130.00	\$2.26	\$1,130.00
TOTAL	\$10.61	\$5,305.00	\$10.71	\$5,355.00

Note: McLane Black Lake Fire excess levy runs 1/1/24 through 12/31/27

28

5901 & 5917 Black Lake-Belmore Road SW Ten Percent Annexation Petitions (TUM-25-0453 and TUM-25-0454)



General Government Committee, July 9, 2025

Intent

- Present the Ten Percent Annexation Petitions
- Explain the Sixty Percent Annexation Method Process
- Discuss Next Steps



Background

- The owner of two properties on Black Lake-Belmore Road SW in Tumwater's urban growth area is requesting to be annexed
- A ten percent annexation petition applications were filed with the City on April 25, 2025, and were deemed complete on April 29, 2025, after payment of application review fees
- Because the property owner owns all the property included in the annexation area, they filed annexation petitions representing both ten percent and sixty percent of the value of the property proposed for annexation



Background

- The property owner has indicated their willingness to assume their fair share of the City's indebtedness and are willing to accept the land use designation and zone district for their property
- Staff is unaware of any pending development applications for the property
- Staff inadvertently left the petitions for the second parcel out of the May 14, 2025, General Government Committee packet and June 3, 2025, City Council packet, so this item contains the complete application package for both parcels





Proposed Annexation Area

- The two parcels in the proposed annexation area are located northwest of the intersection of Black Lake-Belmore Road SW and 60th Avenue SW and are 10.75 acres in size
- The proposed annexation area is west of the current city limit and has a Single Family Low Density Residential district designation



Proposed Annexation Area

- 5901 Black Lake-Belmore Road SW
- Parcel #12705220200

- Owned by Terrence Hess
- Approximately 9.72 acres





Proposed Annexation Area

- 5917 Black Lake-Belmore Road SW
- Parcel #12706110201 • Approximately 1.03 acres 58THLN SW BLACK LAKE-BELMORE RD SW 59TH AVE SW JODY 2 SM GOTH AVE SW MS GIST CT SW CORTEZ DR SW RD NORE



• Owned by Terrence Hess

Initial Request of City Council

The ten percent annexation petitions request that the City Council set a date not later than 60 days after the filing of the ten percent petitions for a meeting with the property owner to determine:

- 1. Whether the City Council will accept, reject, or geographically modify the proposed annexation;
- 2. Whether the City Council will require adoption of a new land use designation for the area to be annexed; and


Sixty Percent Annexation Method Process



CITY OF

37

Staff Recommendation

- Accept the proposal to annex, with no modifications
- Do not propose any amendments to the existing land use designations or zone districts
- Require the annexed area to assume City indebtedness



Initial Request of City Council

 The General Government Committee will be asked to place the ten percent annexation petitions on the July 15, 2025, City Council consideration calendar for the City Council to discuss with the applicants



Future Steps

- If the Council decides to accept the annexation proposal either as proposed or modified – then a sixty percent petition will be drafted and circulated to the affected property owners
- Approval by property owners representing sixty percent of the assessed value of the annexation area is required
- Final approval would only occur after review and approval of the Boundary Review Board, followed by a public hearing and a vote of City Council



Attachment C

- Contraction	CITY OF TUMWATER	TUM - 25	DATE STAMP
	555 ISRAEL RD. SW, TUMWATER, WA 98501 Email: cdd@ci.tumwater.wa.us (360) 754-4180	0453	04-25-2025
CITY OF Z	PETITION FOR ANNEXATION TO THE CITY OF TUMWATER	Kerri RCVD BY	

TO THE HONORABLE MAYOR AND CITY COUNCIL OF THE CITY OF TUMWATER

We, the undersigned, being the owners of not less than sixty percent (60%) in value of the real property herein described, do hereby petition that such territory be annexed to and made a part of the City of Tumwater.

The territory proposed to be annexed is within Thurston County, Washington, and the boundaries are outlined on the plat accompanying petition. The property is described as follows:

> (Legal Description) See "Exhibit A"

The City Council of the City of Tumwater met with the initiating parties at a Council meeting on:

. . .

______ and did determine that the City would accept the proposed annexation. At said meeting, the City Council did also determine that:

It would require the simultaneous adoption of a comprehensive plan for the area to be annexed; and

It would require the assumption of existing City indebtedness by the area to be annexed, as disclosed by the minute entry regarding the same in the minutes of said Council meeting.

Petition for Annexation to the City of Tumwater

This page is one of a number of identical pages forming one petition seeking the annexation of territory to the City of Tumwater, Washington, as above stated, and may be filed with other pages containing additional signatures.

The petitioners subscribing hereto agree that all property within the territory hereby sought to be annexed shall be assessed and taxed at the same rate and on the same basis as other property within the City of Tumwater, including assessments or taxes in payment of any bonds issued or debts contracted prior to or existing at the date of annexation.

	(Names of the petitioners should be	in identical form as the same appear of
	record in the splain of the to the	e real estate. All owners must sign.)
	Tenny I Idem	Terrence Hess
	OWNER'S SIGNATURE	PRINTED NAME
ADDRESS:	5901 Black Lake Belmore Rd SW,	Olympia, WA 98512

ASSESSOR'S PARCEL #:127	05220200	DATE:	11/07/2024
APPROXIMATE # OF ACRES:	9.72		

41

-	CITY OF TUMWATER	TUM - 25-	DATE STAMP
	555 ISRAEL RD. SW, TUMWATER, WA 98501 Email: cdd@ci.tumwater.wa.us (360) 754-4180	0453	04-25-2025
TUMARTIER	PETITION FOR ANNEXATION TO THE CITY OF TUMWATER	Kerri RCVD BY	

TO THE HONORABLE MAYOR AND CITY COUNCIL OF THE CITY OF TUMWATER

We, the undersigned, being the owners of not less than sixty percent (60%) in value of the real property herein described, do hereby petition that such territory be annexed to and made a part of the City of Tumwater.

The territory proposed to be annexed is within Thurston County, Washington, and the boundaries are outlined on the plat accompanying petition. The property is described as follows:

> (Legal Description) See "Exhibit A"

The City Council of the City of Tumwater met with the initiating parties at a Council meeting on:

_____ and did determine that the City would accept the proposed annexation. At said meeting, the City Council did also determine that:

It would require the simultaneous adoption of a comprehensive plan for the area to be annexed; and

It would require the assumption of existing City indebtedness by the area to be annexed, as disclosed by the minute entry regarding the same in the minutes of said Council meeting.

Petition for Annexation to the City of Tumwater

This page is one of a number of identical pages forming one petition seeking the annexation of territory to the City of Tumwater, Washington, as above stated, and may be filed with other pages containing additional signatures.

The petitioners subscribing hereto agree that all property within the territory hereby sought to be annexed shall be assessed and taxed at the same rate and on the same basis as other property within the City of Tumwater, including assessments or taxes in payment of any bonds issued or debts contracted prior to or existing at the date of annexation.

(Names of the petitioners should be in ia	lentical form as the same appear of
record in the shain of title to the real	l estate. All owners must sign.)
Jenny Liden	Terrence Hess
OWNER'S SIGNATURE	PRINTED NAME

WNER'S SIGNATURE

ADDRESS:

ASSESSOR'S PARCEL #: 12705220200

APPROXIMATE # OF ACRES:	9.72

DATE: ____ 11/07/2024

Updated 06/18/2018

42

(Carlor)	CITY OF TUMWATER	TUM -25-	DATE STAMP
	555 ISRAEL RD. SW, TUMWATER, WA 98501 (360) 754-4180 (360) 754-4126 (FAX) Email: cdd@ci.tumwater.wa.us	0453	04-25-2025
TUMMATTER	NOTICE OF INTENT TO COMMENCE ANNEXATION PROCEEDINGS	Kerri RCVD BY	

TO THE CITY COUNCIL OF THE CITY OF TUMWATER

We, the undersigned, being owners of not less than ten percent in value of real property herein described for which annexation is sought, hereby notify the City Council of the City of Tumwater that it is our desire to commence annexation proceedings.

The property herein referred to is described in Exhibit "A" attached hereto and is depicted in Exhibit "B" further attached hereto.

It is requested that the City Council of the City of Tumwater set a date not later than sixty (60) days after the filing of this request for a meeting with the undersigned to determine:

Whether the City Council will accept the proposed annexation;

Whether the City Council will require adoption of a comprehensive plan for the area to be annexed; and

Whether the City Council will require the assumption of existing City indebtedness by the area to be annexed.

Said property is now in Thurston County. We wish to have this property annexed into the City of Tumwater and are willing to assume our fair share of the City's indebtedness and are willing to accept the City's Comprehensive Plan.

This page is one of a group of pages containing identical text material and is intended by the signers of this Notice of Intent to be presented and considered as one Notice of Intent.

(Names of the petitioners should be in identical form as the same appear on record in the chain of title to the real estate. All owners must sign.)

OWNER'S SIGNATURE

Terrence Hess

PRINTED NAME

ADDRESS: 5901 Black Lake Belmore RD SW Olympia WA 98512

DATE: 11/7/24

ASSESSOR'S PARCEL#: 12705220200

LEGAL DESCRIPTION OF ENTIRE ANNEXATION AREA:

See "Exhibit A"

APPROXIMATE # OF ACRES: 9.72

Item 4.

Attachment D

EXHIBIT "A" - LEGAL DESCRIPTION

See "Exhibit A"

EXHIBIT "B" – MAP





Data for Parcel No. 12705220200

Owner(s): HESS, TERRENCE E

Address:	5937 BLACK LAKE BELMORE RD SW
City:	OLYMPIA
State:	WA, 98512
Site Address:	
Site City:	
Site Zip:	
Section:	S05172W
Abbreviated Legal:	Section 05 Township 17 Range 2W S2 N2 NW NW LESS .22A
	RW
Usecode:	91
Tax Code Area:	450
Taxable:	Yes
Annual Tax:	View Property Taxes for Parcel
Property Type:	LND
Total Acres:	9.72
Land Value:	View Assessor's Data for Parcel
Building Value:	View Assessor's Data for Parcel
Total Value:	View Assessor's Data for Parcel
Current Use:	N
Exemptions:	None
Wetlands:	Unknown
Flood Zone:	OUT
Flood of 1999:	Unknown
Winter Flooding of 1996:	Unknown
High Groundwater Flood Hazards:	Unknown
Zoning:	SFL, Single-Family Low Density Residential
Commissioner District:	4
Historic Site:	No
Permitting Jurisdiction:	COUNTY
Jurisdiction of Influence:	TUMUGA
No Shooting Zone:	NO
Animal Control:	Ordinance No. 12989, Contact Animal Services (360-352-2510).
Weed Containment Zone:	NO
Landslide Hazard Review Area - Slope A:	Yes - Check with Pullding Development Center
Landslide Hazard Review Area - Slope B.	Net manped by GooData
Cround Water Sensitive Areas	Not mapped by GeoData
Cround Water Sensitive Aleas.	
Dink Natural Hentage Data.	Ves
Mazama Bocket Copher Soils:	Less Preferred More Preferred
Mazama Pocket Copher Solis.	
Mazama Pocket Copher:	OPG (Olympia Pocket Gonber)
Oregon Vesner Sparrow:	Unknown
Taylor's Checkerspot Butterfly:	Unknown
Oregon Spotted Frog:	Unknown
Marine Riparian Review Area - 300':	Unknown

DEC 12'19 744700

When recorded return to: Mr. Terrence E Hess 3448 7th Ave. SW Olympia, WA 98502

Filed for Record at Request of Foundation Escrow, Inc. Escrow Number: 19-4492-JB Thurston County Treasurer, 10

Real Estate Excise. 'ax Pai By

CHICAGO TITLE

Statutory Warranty Deed

NO. 100033425

THE GRANTOR Seaport Assets, Inc, a California Curporation for and in consideration of TEN DOLLARS AND OTHER GOOD AND VALUABLE CONSIDERATION in hand paid, conveys and warrants to Terrence E Hess, an unmarried individual the following described reat estate. situated in the County of Thurston. State of Washington, a married man as his separate estate

Abbreviated Legal:

Ptn NW ½ NW ½ 5-17-2W Tax Account No.: 12705220200

Tax Parcel Number(s): 12705220200

Full Legal Description Attached to Exhibit "A" SUBJECT TO: Special Exceptions shown on Attached Exhibit "B"

Dated November 26, 2019 .

Seaport Assets, Inc

By Matthew Wayne Walker, Chief Executive Officer

STATE OF California COUNTY OF San Dieg SS:

I certify that I know or have satisfactory evidence

Maithew Wayne Walker is/are the person(s) who appeared before

me, and said person(s) acknowledge he signed this instrument, on oath stated he is/are authorized to execute the instrument and acknowledge that as the Chief Executive Officer of Seaport Assots, inc

to be the free and voluntary act of such party(ies) for the uses and purposes mentioned in this instrument.

Dated: 11 26 2019

ABE ALBANNA COMM. #2285218 NOTARY PUBLIC-CALIFORNIA SAN DIEGO COUNTY My Comm. Expires May 11, 2023

ape alle

Notary Public in and for the	State of California
Residing at San Dirgo	(oun ty)
My appointment expires:	May (1, 2023

LPB 10-05(1-0) Page 1 of 1

Item 4.

EXHIBIT A

The South half of the North half of the Northwest quarter of the Northwest quarter of Section 5, Township 17 North, Range 2 West, W.M.; EXCEPT the East 30 feet for county road known as Belmore-Black Lake Road, as conveyed to Thurston County by deed recorded April 10, 1953 under Auditor's File No. 522295;

EXCEPTING any portion thereof lying South of an agreed boundary established as the South line thereof which is described as follows:

Commencing at a concrete monument marking the Northwest corner of Section 5, Township 17 North, Range 2 West, W.M.; Thence South 00° 00' 37" East along the Section line between said Section 5 and Section 6 for a distance of 651.38 feet; Thence North 89° 38' 41" East for a distance West to and extending through the shorelands lying in front of, adjacent to, and abutting upon Government Lot 8, Section 6, Township 17 North, Range 2 West, W.M.;

EXCEPTING ALSO that portion thereof, if any, lying North of the following described line:

Commencing at a concrete monument marking the Northwest corner of Section 5, Township 17 North, Range 2 West W.M.; Thence South 00° 00' 37" East along the section line between said Section 5 and Section 6 for a distance of 325.38 feet; Thence North 89° 39' 41" East for a distance of 1276.90 feet, more or less, to the West right-of-way line of the Belmore-Black Lake County Road and the Eastern terminus and point of beginning of this description; Thence South 89° 28' 41" West to and extending through the shorelands lying in front of, adjacent to, and abutting upon Government Lot 8, Section 6, Township 17 North, Range 2 West, W.M.; ALSO EXCEPTING county road known as Belmore-Black Lake Road along the East line of the Northwest quarter of the Northwest quarter of Section 5;

In Thurston County, Washington

Item 4.			
Ro	CITY OF TUMWATER	TUM - 25	DATE STAMP
	555 ISRAEL RD. SW, TUMWATER, WA 98501 Email: cdd@ci.tumwater.wa.us (360) 754-4180	0-154	04-25-2025
TUMMENDER	PETITION FOR ANNEXATION TO THE CITY OF TUMWATER	Kerri RCVD BY	

Attachment E

TO THE HONORABLE MAYOR AND CITY COUNCIL OF THE CITY OF TUMWATER

We, the undersigned, being the owners of not less than sixty percent (60%) in value of the real property herein described, do hereby petition that such territory be annexed to and made a part of the City of Tumwater.

The territory proposed to be annexed is within Thurston County, Washington, and the boundaries are outlined on the plat accompanying petition. The property is described as follows:

> (Legal Description) See "Exhibit A"

The City Council of the City of Tumwater met with the initiating parties at a Council meeting on:

_____ and did determine that the City would accept the proposed annexation. At said meeting, the City Council did also determine that:

It would require the simultaneous adoption of a comprehensive plan for the area to be annexed; and

It would require the assumption of existing City indebtedness by the area to be annexed, as disclosed by the minute entry regarding the same in the minutes of said Council meeting.

Petition for Annexation to the City of Tumwater

This page is one of a number of identical pages forming one petition seeking the annexation of territory to the City of Tumwater, Washington, as above stated, and may be filed with other pages containing additional signatures.

The petitioners subscribing hereto agree that all property within the territory hereby sought to be annexed shall be assessed and taxed at the same rate and on the same basis as other property within the City of Tumwater, including assessments or taxes in payment of any bonds issued or debts contracted prior to or existing at the date of annexation.

(Names of the petitioners should be in record in the shain of title to the r	n identical form as the same appear of real estate. All owners must sign.)
Jenny L'dem	Terrence Hess
OWNER'S SIGNATURE	PRINTED NAME

ADDRESS: 5917 B	ack Lake Belmore Rd Sw, Olympia, WA 98512		
ASSESSOR'S PARCEL #:	12706110201	DATE:	11/07/2024
APPROXIMATE # OF AC	RES: 1.03		

Par	CITY OF TUMWATER	TUM - 25-	DATE STAMP		
	555 ISRAEL RD. SW, TUMWATER, WA 98501 (360) 754-4180 (360) 754-4126 (FAX) Email: cdd@ci.tumwater.wa.us	0454	04-25-2025		
CITY OF Z	NOTICE OF INTENT TO COMMENCE ANNEXATION PROCEEDINGS	Kerri RCVD BY			

TO THE CITY COUNCIL OF THE CITY OF TUMWATER

We, the undersigned, being owners of not less than ten percent in value of real property herein described for which annexation is sought, hereby notify the City Council of the City of Tumwater that it is our desire to commence annexation proceedings.

The property herein referred to is described in Exhibit "A" attached hereto and is depicted in Exhibit "B" further attached hereto.

It is requested that the City Council of the City of Tumwater set a date not later than sixty (60) days after the filing of this request for a meeting with the undersigned to determine:

Whether the City Council will accept the proposed annexation;

Whether the City Council will require adoption of a comprehensive plan for the area to be annexed; and

Whether the City Council will require the assumption of existing City indebtedness by the area to be annexed.

Said property is now in Thurston County. We wish to have this property annexed into the City of Tumwater and are willing to assume our fair share of the City's indebtedness and are willing to accept the City's Comprehensive Plan.

This page is one of a group of pages containing identical text material and is intended by the signers of this Notice of Intent to be presented and considered as one Notice of Intent.

(Names of the petitioners should be in identical form as the same appear on record in the chain of title to the real estate. All owners must sign.)

OWNER'S SIGNATURE

Terrence Hess

PRINTED NAME

ADDRESS:5917 Black Lake Belmore RD SW, Olympia, WA 98512 DATE: 11/7/24

ASSESSOR'S PARCEL#: 12706110201

LEGAL DESCRIPTION OF ENTIRE ANNEXATION AREA: _

See "Exhibit A"

APPROXIMATE # OF ACRES: 1.03

Item 4

Attachment F



50



HESS, TERRENCE & HESS, SARA J

Owner(s):

Address:	4709 BRECH ST SE
City:	TUMWATER
State:	WA, 98512
Site Address:	5917 BLACK LAKE BELMORE RD SW
Site City:	OLYMPIA
Site Zip:	98512
Section:	S06172W
Abbreviated Legal:	6-17-2W L8 COM NE SEC COR S 0-00-37 E 651.38F TO POB
Usecode:	91
Tax Code Area:	450
Taxable:	Yes
Annual Tax:	View Property Taxes for Parcel
Property Type:	LND
Total Acres:	1.03
Land Value:	View Assessor's Data for Parcel
Building Value:	View Assessor's Data for Parcel
Total Value:	View Assessor's Data for Parcel
Current Use:	N
Exemptions:	None
Wetlands:	Unknown
Flood Zone:	OUT
Flood of 1999	Unknown
Winter Flooding of 1996	Unknown
High Groundwater Flood Hazards:	Unknown
Zoning	SFL, Single-Family Low Density Residential
Commissioner District	4
Historic Site:	No
Permitting Jurisdiction:	COUNTY
Jurisdiction of Influence:	TUMUGA
No Shooting Zone:	No
Animal Control:	Ordinance No. 12989. Contact Animal Services (360-352-2510).
Weed Containment Zone:	No
Landslide Hazard Review Area - Slope A:	Yes - Check with PBuilding Development Center
Landslide Hazard Review Area - Slope B:	Unknown
Landslide Hazard Review Area - Slope C. D. F.	Not mapped by GeoData
Ground Water Sensitive Areas:	No
DNR Natural Heritage Data:	Unknown
Prairie Indicator Soils:	Unknown
Mazama Pocket Gopher Soils:	Less Preferred
Mazama Pocket Gopher:	Unknown
Mazama Pocket Gopher Service Area:	OPG (Olympia Pocket Gopher)
Oregon Vesper Sparrow:	Unknown
Taylor's Checkerspot Butterfly:	Unknown
Oregon Spotted Frog:	Unknown
Marine Riparian Review Area - 300':	Unknown
Stream Riparian Review Area - 300'	Unknown
A sent the full for the for the sect	

08/30/2022 01:42 PM D Thurston County, Washington JENSEN MILNER | CAP CITY LAW PS

After recording return document to:

Item 4

Cap City Law PS 2401 Bristol Court SW, Suite A-103 Olympia, WA 98502

> Thurston County Treasurer E028610 \$0.00 08/30/2022 Disate the A Walker

Document Title: Related Documents: Grantors: **Grantees:** Legal Description: Tax Parcel Number:

QUITCLAIM DEED 3330319 **Terrence Hess and Sara J. Hess Terrence Hess and Sara J. Hess** 6-17-2W L8 COM NE SEC COR S 0-00-37 E 651.38F TO POB 12706110201

QUITCLAIM DEED

The Grantors, Terrence Hess and Sara J. Hess, formerly married persons, in consideration of a divorce action between Terrence Hess and Sara J. Hess, hereby convey and quitclaim to the Grantees, Terrence Hess, a single person, and Sara J. Hess, a single person, as joint tenants with rights of survivorship, the following-described real estate, and any after-acquired interest therein, situate in Thurston ' County, in the State of Washington:

SEE EXHIBIT A

Terrence Hess Sara J; 688

Page 1 of 4 Pages PLEASE MAKE NO MARK IN THE MARGIN SPACE - RESERVED FOR COUNTY AUDITOR'S USE

STATE OF WASHINGTON

THURSTON COUNTY, SS.

August 19, 2022

I certify that I know or have satisfactory evidence that **Sara J. Hess** personally appeared before me and acknowledged signing this instrument as a free and voluntary act for the uses and purposes mentioned in the instrument.

(U) HBefore me WHIIIIII :SSICA Jessica McKeegan-Jensen Notary Public for the State of Washington 2401 Bristol Court SW, Suite A-103 Olympia, Washington 98502 Commission Expires 6/16/2023

STATE OF WASHINGTON THURSTON COUNTY, SS.

August 19, 2022

I certify that I know or have satisfactory evidence that **Terrence Hess** is the person who appeared before me, and acknowledged signing this instrument as a free and voluntary act for the uses and purposes mentioned in the instrument.

Before me: Notary Public for the State of Washington > Printed Name: Jessica McKeegan Jensen SICA Residing at: Olympia, WA 98502 Commission Expires: 6/16/2023

Page 2 of 4 Pages

PLEASE MAKE NO MARK IN THE MARGIN SPACE - RESERVED FOR COUNTY AUDITOR'S USE ONLY.

EXHIBIT A

PARCEL A:

THAT PORTION OF GOVERNMENT LOT 8, SECTION 6, TOWNSHIP 17 NORTH, RANGE 2 WEST, W.M., DESCRIBED AS FOLLOWS:

COMMENCING AT THE NORTHEAST CORNER OF SAID SECTION 6, THENCE SOUTH 00° 00' 37" EAST 651.38 FEET TO A POINT ON THE EAST LINE OF SAID GOVERNMENT LOT 8 WHICH IS THE TRUE POINT OF BEGINNING; THENCE SOUTH 89° 38' 41: WEST 150 FEET; THENCE NORTH 00° 00' 37" WEST 326 FEET; THENCE NORTH 89° 38' 41" EAST 150 FEET TO THE EAST LINE OF SAID GOVERNMENT LOT 8; THENCE SOUTH 00° 00' 37" EAST 326 FEET TO THE TRUE POINT OF BEGINNING; EXCEPT THE SOUTH 30 FEET.

Page 3 of 4 Pages

PLEASE MAKE NO MARK IN THE MARGIN SPACE - RESERVED FOR COUNTY AUDITOR'S USE ONLY.





Item 5

TO:General Government CommitteeFROM:Brad Medrud, Community Development DirectorDATE:July 9, 2025SUBJECT:Resolution No. R2025-011, 93rd Avenue SW and Case Road SW Annexation (TUM-
25-0128)

1) <u>Recommended Action</u>:

Place Resolution No. R2025-011 for the August 19, 2025, City Council meeting to allow the City Council to conduct a public hearing on the proposed annexation petitions.

2) <u>Background</u>:

The owners of two properties in Tumwater's urban growth area at the northwest corner of 93rd Avenue SW and Case Road SW in unincorporated Thurston County filed petition applications under Chapter 35A.14 RCW for annexation to the City on February 5, 2025, that were deemed complete on March 18, 2025, after payment of the necessary review fees.

Because the two property owners own all the property included in the annexation area, they filed petitions representing both the ten percent and sixty percent of the value of the properties proposed for annexation. The property owners have asked that their properties be annexed into the City of Tumwater and indicated their willingness to assume their fair share of the City's indebtedness and are willing to accept the City's Comprehensive Plan.

The General Government Committee was briefed on the proposed annexation petitions at their April 15, 2025, meeting. The City Council met the petitioners and agreed to accept the proposed annexation for further processing and determined that the City will require the assumption of existing City indebtedness by the area to be annexed at their May 6, 0225 meeting. The Thurston County Assessor certified that the sixty percent petitions were legally sufficient (i.e., as having valid signatures representing the required 60 percent of the property value) on June 3, 2025.

Under RCW 35A.14.130, when a legally sufficient annexation petition is filed, the City Council may consider it by doing the following:

- Fix a date for a public hearing, and
- Provide notice specifying the time and place of the hearing and inviting interested people to appear and voice approval or disapproval of the annexation. The notice is to be:
 - \circ $\;$ Published in one or more issues of a newspaper of general circulation in the City; and
 - Posted in three public places within the territory proposed for annexation.

There are no statutory requirements concerning the actual hearing, other than to give proponents and opponents an opportunity to speak.

The General Government Committee is asked to set the date for the public hearing on the

annexation petition for the City Council's August 19, 2025, meeting.

3) <u>Policy Support</u>:

Goal LU-2: Ensure development takes place in an orderly and cost-efficient manner in order to best utilize available land and public services, conserve natural resources, protect critical areas, preserve open space, and reduce sprawl.

Policy LU-2.4 Ensure new annexations adhere to the goals and policies of the City's Annexation Policy.

4) <u>Alternatives</u>:

None.

5) Fiscal Notes:

An initial review of the fiscal impact of the proposed annexations was determined in advance of the May 6, 2025, City Council meeting.

- 6) <u>Attachments</u>:
 - A. Resolution No. R2025-011

RESOLUTION NO. R2025-011

A **RESOLUTION** of the City Council of the City of Tumwater, Washington, establishing intent to annex certain contiguous property located adjacent to 93rd Avenue SW and Case Road SW in Thurston County, Washington, to the City of Tumwater.

WHEREAS, on March 18, 2025, pursuant to RCW 35A.14.120, Nathan Waunch and H. John and Diane Elwin, the owner of not less than ten percent (10%) in value according to the assessed valuation for general taxation of the property hereinafter described, notified the Tumwater City Council of their intent to commence proceedings for the annexation to the City of Tumwater of said property; and

WHEREAS, on April 15, 2025, the General Government Committee was briefed on the request for annexation and scheduled and placed the ten percent annexation petitions on the May 6, 2025, City Council consideration calendar; and

WHEREAS, on May 6, 2025, at a regularly scheduled meeting, the City Council met with the initiating parties and accepted the proposal to annex, determined the proposed annexation boundaries, and determined that properties within the newly annexed area will assume a proportionate share of indebtedness; and

WHEREAS, on June 3, 2025, the Thurston County Assessor certified that the petition to annex meets the sufficiency requirements of RCW 35A.01.040, including the requirement that it be signed by owners of more than sixty percent (60%) in value according to the assessed valuation for general taxation of said property; and

WHEREAS, on July 9, 2025, said petition was thereafter filed with the City Council; and

WHEREAS, on July 9, 2025, the General Government Committee scheduled a public hearing with the City Council on the proposed annexation for August 19, 2025; and

WHEREAS, the Tumwater City Clerk caused Notice of Public Hearing on the petition for annexation to be posted and published in the manner provided by law; and

WHEREAS, the annexation is consistent with the City of Tumwater adopted annexation policies, the County-wide Planning Policies, and the Joint Plan adopted by Thurston County and the City of Tumwater; and

WHEREAS, on August 19, 2025, a public hearing on said petition for annexation was held in the City Council Chambers of the City Hall; and the City Council, having

Resolution No. R2025-011 – Page 1 of 6

determined that the annexation of the hereinafter described property to the City of Tumwater would be in the public interest and for the public welfare and in the best interest of the City of Tumwater and the residents thereof;

NOW, THEREFORE, BE IT RESOLVED BY THE CITY COUNCIL OF THE CITY OF TUMWATER AS FOLLOWS:

<u>Section 1</u>. The Tumwater City Council hereby declares its intent to annex the property described in Exhibit A and depicted in Exhibit B attached hereto and by this reference incorporated herein as if fully set forth,

<u>Section 2</u>. It is hereby further declared that upon annexation, the property described in Section 1 above shall be subject to existing indebtedness of the City of Tumwater. Zoning and land use designations shall be applied which are consistent with the Tumwater Comprehensive Plan. The Airport Overlay Zone shall be applied pursuant to the Tumwater Municipal Code and the Tumwater Comprehensive Plan. The Aquifer Protection Overlay Zone shall also be applied to the property described in Section 1 above.

<u>Section 3</u>. Upon the Thurston County Boundary Review Board's approval of this annexation, the city attorney is directed to prepare an ordinance for council consideration annexing said property to the City of Tumwater.

<u>Section 4</u>. <u>Severability</u>. The provisions of this Resolution are declared separate and severable. The invalidity of any clause, sentence, paragraph, subdivision, section, or portion of this Resolution or the invalidity of the application thereof to any person or circumstance, shall not affect the validity of the remainder of the Resolution, or the validity of its application to other persons or circumstances.

<u>Section 5</u>. <u>Effective Date</u>. This Resolution shall become effective immediately upon adoption and signature as provided by law.

RESOLVED this _____ day of _____, 20____.

CITY OF TUMWATER

Debbie Sullivan, Mayor

ATTEST:

Resolution No. R2025-011 – Page 2 of 6

Melody Valiant, City Clerk

APPROVED AS TO FORM:

Karen Kirkpatrick, City Attorney

Exhibit "A"

Property #1

1020 93rd Avenue, SW, Tumwater, WA Thurston County Assessor Number: 12715420203 Approximate Number of Acres: 4.87 (10.74 total annexation area)

LEGAL DESCRIPTION

THE WEST 300 FEET OF THE SOUTH 726 FEET OF THE SOUTHWEST QUARTER OF THE SOUTHEAST QUARTER OF SECTION 15, TOWNSHIP 17 NORTH, RANGE 2 WEST, W.M.; EXCEPTING THEREFROM THE SOUTH 30 FEET FOR COUNTY ROAD KNOWN AS LATHROP ROAD.

Property #2

93rd Avenue, SW, Tumwater, WA Thurston County Assessor Number: 12715430400 Approximate Number of Acres: 5.87 (10.74 total annexation Area

LEGAL DESCRIPTION

THAT PORTION OF THE SOUTHEAST QUARTER OF SECTION 15, TOWNSHIP 17 NORTH, RANGE 2 WEST, W.M., SHOWN AS TRACT 1 OF SURVEY RECORDED MARCH 20, 1979 UNDER FILE NO. 1071074, IN THURSTON COUNTY, WASHINGTON. SITUATE IN THE COUNTY OF THURSTON, STATE OF WASHINGTON.

Exhibit "B"



Resolution No. R2025-011 – Page 5 of 6 $\,$



Resolution No. R2025-011 – Page 6 of 6

TO: General Government Committee
FROM: Dana Bowers, Associate Planner, and Erika Smith-Erickson, Housing and Land Use Planner
DATE: July 9, 2025
SUBJECT: 2025 Comprehensive Plan Periodic Update – Conservation Element

1) Recommended Action:

This is a discussion item about the Conservation Element for the 2025 Comprehensive Plan periodic update.

2) <u>Background</u>:

On a ten-year cycle, the City is required to conduct a Growth Management Act periodic update of its Comprehensive Plan and related development regulations. For the current cycle, the City is required to complete work on the periodic update by December 31, 2025. Work on the periodic update started last fall.

The updated Comprehensive Plan will address diversity, equity, and inclusion throughout the Plan. <u>2025 Comprehensive Plan Update | City of Tumwater, WA</u> contains links to guidance material and information about the update.

The intent of this briefing is to continue the presentation of the complete draft Conservation Element for discussion by the General Government Committee from the June 11, 2025 meeting.

3) Policy Support:

Comprehensive Plan Goal C-1: Recognize the significant role played by natural features and systems in determining the overall environmental quality and livability of Tumwater.

4) <u>Alternatives</u>:

None.

5) Fiscal Notes:

This is primarily an internally funded annual work program task.

- 6) <u>Attachments</u>:
 - A. Staff Report
 - B. Presentation
 - C. Conservation Element, Part 1 Goals, Policies, and Implementation Actions
 - D. Conservation Element, Part 2 Technical Information

- E. Conservation Element, Part 2 Technical Information Appendices
 F. Conservation Element, Part 2 Technical Information Maps

MEMORANDUM

Date: July 9, 2025

- To: General Government
- From: Dana Bowers, Long Range Planner, and Erika Smith-Erickson, Housing and Land Use Planner



2025 Comprehensive Plan Update – Conservation Element

On a ten-year cycle, Tumwater is required to conduct a Growth Management Act periodic update of its Comprehensive Plan and related development regulations. For the current cycle, Tumwater is required to complete work on the periodic update by December 31, 2025.

The updated Comprehensive Plan will address diversity, equity, and inclusion throughout the Plan and incorporate many of the State required changes addressing housing, climate change, and other topics.

The intent of the General Government Committee meeting on Wednesday, July 9, 2025, is to continue the discussion of the complete draft Conservation Element from the June 11, 2025, meeting. The Element consist of two parts.

- **Part 1** contains the Conservation Element's goals, policies, and draft implementation actions. The General Government Committee will be asked to consider whether the proposed goals, policies, and draft implementation actions are appropriate and whether the information in the Element is presented effectively.
- **Part 2** contains the Conservation Element's technical information used to update the Element. The General Government Committee will be asked to consider whether the information in the Element is presented in an understandable format.

Below are questions for your consideration as you review the element:

- 1. Do you have any general or specific comments about the goals?
- 2. Are there shifts in action timelines to that you would suggest?
- 3. Do you foresee any unintended consequences or burdens associated with policies and actions?
- 4. Which could impact you or the things you care about most? What do you expect that impact might be?
- 5. Do you have any additional thoughts you'd like to share related to the goals and their associated policies and actions?

Contents

1 – Growth Management Act – Conservation Goals	2
2 – Conservation Element	3
3 – Structure of the Part 1 of the Element	3
4 – Structure of the Part 2 of the Element	4
5 – Goals, Policies, and Draft Implementation Actions Review	5
Appendix A – Guidance	7
Appendix B – Current Conservation Element Goals, Policies, and Actions	8

1 – Growth Management Act – Conservation Goals

The state Growth Management Act (Chapter 36.70A Revised Code of Washington (RCW)) requires that Tumwater demonstrate that each Element in its Comprehensive Plan meets the relevant fifteen planning goals contained within the Act. The fifteen goals guide the development and adoption of the Comprehensive Plan and development regulations.

The following is a summary of how the updated Conservation Element meets the two goals related to natural resource industries and environment.

8. **Natural resource industries.** Maintain and enhance natural resource-based industries, including productive timber, agricultural, and fisheries industries. Encourage the conservation of productive forestlands and productive agricultural lands, and discourage incompatible uses.

The Conservation Element has specific guidelines and policies that ensure the viability of natural resource industries and activities. Additionally, the Conservation Element ensures the viability of natural resource industries in Tumwater through the identification of such lands in the Conservation Element text and maps.

While Tumwater has limited natural resource lands as defined by the Growth Management Act, it does have limited lands that are used for mineral resources, forestry, and urban agriculture that will change uses as Tumwater develops as required by state law into an urban area. While it does not meet the state definitions of agricultural resource lands, Tumwater supports urban agriculture with the Conservation Element's goals and policies.

10. *Environment.* Protect and enhance the environment and enhance the state's high quality of life, including air and water quality, and the availability of water.

The Conservation Element contains specific policies relating to air and water quality, water availability, and protection and preservation of critical areas and addresses how to

enhance the environment. If conflict occurs in the implementation of urban growth policy and development regulations, protecting critical areas will be the priority.

2 – Conservation Element

A – Background

As required by the Growth Management Act, the Conservation Element addresses both natural resource lands and critical areas in Tumwater. Natural resources lands, such as agricultural, forestry and mineral resource lands, are typically designated outside of cities and urban growth areas as those types of uses are intended for rural areas under the Growth Management Act. Critical areas include environmentally sensitive spaces such as wetlands, critical aquifer recharge areas, frequently flooded areas, geologically hazardous areas, and fish and wildlife habitat conservation areas.

Areas that are within Tumwater's urban growth area are addressed through the Tumwater and Thurston County Joint Plan in accordance with adopted County-Wide Planning Policies.

The Conservation Element is separated into two parts which are described below.

Part 1 contains the Element's draft goals, policies, and draft implementation actions that will guide the implementation of the Element over the next 20 years. The goals, policies, and actions of the 2016 Conservation Element are found in Appendix B of this staff report.

Part 2 contains an inventory of the natural resource lands and critical areas in Tumwater. Natural resource lands are identified based on state designation listed by the county assessor. Critical areas are identified based criteria provided at the state and federal levels. For example floodways are determined by FEMA and Washington State Department of Ecology provides guidance for wetland identification.

The Conservation Element's maps in Attachment F show where resource lands are designated and the general extent of critical areas in Tumwater. The actual boundaries of critical areas are determined on a project level as directed by the regulations in TMC Title 16 *Environment*.

B – Link to Current Conservation Element

https://www.ci.tumwater.wa.us/departments/community-developmentdepartment/tumwater-comprehensive-plan

3 - Structure of the Part 1 of the Element

Part 1 – Goals, Policies, and Implementation Actions of the Element is structured similarly to the Part 1 of the Housing and Land Use Elements. The intent of separating the goals, policies, and draft implementation actions from the technical information for each Element is to make it easier for policymakers and the community to use the document.

Chapter 1 Introduction

Chapter 1 provides a short background to the purpose of the Conservation Element and an explanation of how to read the Element.

Chapter 2 Growth Management Act – Element Goals

Chapter 2 discusses the Element's connection to the conservation goals of the state Growth Management Act.

Chapter 3 County-Wide Planning Policies

Chapter 3 discusses the Element's connection to the Thurston County-Wide Planning Policies.

Chapter 4 Element Goals and Policies

Chapter 4 presents the Conservation Element's goals and policies in detail with an explanation the importance of each goal, what Tumwater department is responsible for implementation, and timeline for those actions. Comments are provided that discuss the source of each of the goals and policies.

The Conservation Element's goals and policies are the policy basis for the draft implementation actions in the Element and those future actions that will be developed over the next 20 years which will be the foundation for Tumwater's annual work programs to address natural resource lands and critical areas.

Appendix A Draft Implementation Actions

Appendix A contains the draft implementation actions, which are intended to be a source of annual work program items that serve to implement the goals and policies of the Element.

The annual work programs will further refine the draft implementation actions prior to their being put into practice. It is expected that draft implementation actions will be further amended, added, or subtracted as needed over the course of the 20 year Comprehensive Plan as new opportunities arise to meet the intent of the Element's goals and policies.

4 – Structure of the Part 2 of the Element

Part 2 – Technical Information of the Conservation Element consists of the following chapters.

Chapter 1 – Introduction

Discusses the State requirements for the Element, how to read the Element, best available science, and how shoreline of the state are addressed in the Comprehensive Plan.

Chapter 2 – Natural Resources

Provides a summary of urban agriculture, forestry and mineral resource lands and uses in Tumwater.

Chapter 3 – Critical Areas

Provides an overview of all the Tumwater managed critical areas including wetlands, critical aquifer recharge areas, frequently flooded areas, geologically hazardous areas, and fish and wildlife habitat areas.

Appendix A – Foundational Documents

Provides a list of the documents used to create the Conservation Element's Technical Summary.

Appendix B – Open Space Taxation Act Summary

Provides basic information about the State criteria, benefits, and penalties associated with forest and agriculture designation.

Appendix C – Tumwater Soils Report

Provides a map, list, and detailed description of all of the soils within Tumwater.

5 – Goals, Policies, and Draft Implementation Actions Review A – Introduction

Goals and policies describe how Tumwater proposes to address identified needs. Goals are statements of desired outcomes or intended achievements. Policies are specific statements that guide actions and provide a framework for future decision-making. Actions are specific implementations of goals and policies.

Example from the draft Conservation Element:

Goal C-1: Recognize the significant role played by natural features and systems in determining the overall environmental quality and livability of Tumwater

Policy

C-1.2 Use adopted plans to inform critical area policies, regulations, and implementation actions.

Draft Implementation Action

C-1.2.1 Implement the actions identified in the Thurston Climate Mitigation Plan.

How key terms are used in goals, policies, and actions:

- "Shall" means implementation of the policy is mandatory and imparts a higher degree of substantive direction than "should."
- "Should" means implementation of the policy is expected but its completion is not mandatory.
- "May" means the actions described in the policy are either advisable or are allowed.

- "Ensure" means actions described in the policy are guaranteed.
- "Must" means implementation of the policy is an obligation.
- "Require" means implementation of the policy is compulsory.
- "Support" means to advocate for implementation of the policy.
- "Promote" means to help bring about implementation of the policy.
- "Encourage" means to foster or help implementation of the policy.
- "Consider" means to take into account.
- "Coordinate" means to bring into a common action, movement, or condition.
- "Implement" means to carry out or accomplish.
- "Integrate" means to form, coordinate, or blend into a functioning or unified whole.
- "Make" means to enact or establish.
- "Engage" means to do or take part in something.

B – Policy Strength Continuum

When developing goals and policies, it is important to understand the policy strength continuum. The Puget Sound Regional Council developed the following example.

Passive	Policy Strength	Active
Statements of Inclination	Statements of Principle	Statements of Impact
Conveys intent, but establishes no target or definition of success	Describes clear targets or conditions of success	Go further, describing specific situations where protecting critical areas is a priority
Example	Example	Example
Tumwater shall encourage the creation of a new City Center.	Tumwater shall endeavor to designate 100-acres for a new City Center.	Work with the development community and local agencies to create a new City Center based on framework established by Tumwater Center Plan.

For an example of how policies can be written to be more active and how implementation strategies can be established for policies, include identifying who will be responsible for implementing the policy and the timeframes to do so.

Appendix A – Guidance

The State Department of Commerce has provided guidance specific to the periodic update on their Periodic Update webpage.

https://www.commerce.wa.gov/serving-communities/growth-management/periodicupdate/

www.commerce.wa.gov/serving-communities/growth-management/growth-managementtopics

In addition, the Puget Sound Regional Council is conducting a series of workshops on a variety of topics related to the periodic update.

www.psrc.org/our-work/passport-2044-comprehensive-plan-workshop-series)

The Municipal Research Services Center has a Comprehensive Planning webpage.

https://mrsc.org/getdoc/d7964de5-4821-4c4d-8284-488ec30f8605/Comprehensive-Planning.aspx
<u>Appendix B – Current Conservation Element Goals, Policies, and</u> <u>Actions</u>

- 4.1 Conservation Goals, Policies, and Actions
- Goal C-1 Recognize the significant role played by natural features and systems in determining the overall environmental quality and livability of Tumwater

Policy Action

- C-1.1 Protect the ecological integrity of the natural environment while allowing for compatible growth and development
- C-1.2 Promote conservation of natural resources and the environment in cooperation with residents, business owners, schools, affected jurisdictions, and tribes.
- C-1.3 Encourage and support active measures to protect and enhance Tumwater's natural environment.
- C-1.4 Implement the mitigation goals, objectives, and initiatives contained in the most recent version of the adopted Natural Hazards Mitigation Plan for Thurston County.
- C-1.5 Maximize retention of a healthy tree cover and native vegetation and encourage restoration, replacement, and enhancement of unhealthy trees and disturbed vegetation.
- C-1.6 Reduce communitywide greenhouse gas emissions 45 percent below 2015 levels by 2030 and 85 percent below 2015 levels by 2050 to ensure that local communities do their part to keep the global average temperature from rising more than 2°C.
- C-1.7 Implement the strategies contained in the most recent version of the accepted Thurston Climate Mitigation Plan
- Goal C-2 Designate and protect critical areas including wetlands, critical aquifer recharge areas, frequently flooded areas, geologically hazardous areas, and fish and wildlife habitat conservation areas in accordance with the Growth Management Act to protect the functions and values of these areas as well as to protect against threats to health, safety, and property.

Policy Action

- C.2.1 Include best available science in developing policies and development regulations to protect the functions and values of critical areas and consider conservation or protection measures necessary to preserve or enhance anadromous fisheries, consistent with the Growth Management Act.
- C.2.2 Use incentive programs, acquisition, appropriate regulations, and other techniques to preserve critical areas as permanent open space where

development may pose hazards to health, property, or important ecological functions.

- C.2.3 Require that prior to any development, critical areas are identified and protected.
- C.2.4 Ensure the effectiveness of critical area mitigation by requiring adequate critical area studies and mitigation plans, the application of mitigation sequencing, financial assurances from developers to ensure mitigation success, and by improving City oversight of maintenance and monitoring of mitigation sites.
- C.2.5 Require and enforce mitigation to ensure no net loss of critical area functions.
- C.2.6 Support restoration of river and stream channels and associated wetland and riparian areas to enhance water quality, improve fish and wildlife habitat, and mitigate flooding and erosion.
- C.2.7 Allow public access to wetlands, streams, and lakes for scientific, educational, and recreational use, provided the public access is carefully sited, sensitive habitats and species are protected, and hydrologic continuity is maintained.
- C.2.8 Protect wetlands not as isolated units, but as ecosystems, and essential elements of watersheds.
- C.2.9 Protect the quality and quantity of groundwater used for public water supplies.
- C.2.10 Prevent land alterations that would increase potential flooding and minimize the alteration of natural surface water features that retain or carry floodwaters, such as wetlands, floodplains, rivers, streams, and lakes.
- C.2.11 Require mitigation for adverse environmental impacts from engineered flood control measures.
- C.2.12 Work cooperatively to meet regulatory standards for floodplain development as these standards are updated for consistency with relevant federal requirements including those related to the Endangered Species Act.
- C.2.13 Regulate development intensity, site coverage, and vegetation removal in geologically hazardous areas in order to minimize drainage problems, soil erosion, siltation, and landslides.
- C.2.14 Minimize soil disturbance and maximize retention and replacement of native vegetative cover for any land uses permitted in erosion and landslide hazard areas.
- C.2.15 Encourage special building design and construction measures in areas with severe seismic hazards to minimize the risk of structural damage, fire, and injury to occupants during a seismic event and to prevent post-seismic collapse.
- C.2.16 Protect and preserve habitats for species, which have been identified as endangered, threatened, or sensitive by the state or federal government, giving special consideration: to conservation or protection measures necessary to preserve or enhance anadromous fisheries.
- C.2.17 Maintain habitats that support the greatest diversity of fish and wildlife through

conservation and enhancement of critical areas.

- C.2.18 Implement salmon habitat protection and restoration priorities in approved Water Resource Inventory Area 13 and 23 plans.
- C.2.19 Coordinate with adjacent jurisdictions and tribes to identify, protect, and develop enhancement plans and actions for habitat networks and wetlands that cross-jurisdictional lines.
- C.2.20 Promote the enhancement or restoration of streams, rivers, lakes, and wetlands as adjacent development activities occur.
- C.2.21 Protect wildlife corridors to minimize habitat fragmentation, especially along existing linkages and in patches of native habitat by enhancing vegetation composition and structure, and incorporating indigenous plant species compatible with the site.
- Goal C-3 In accordance with the Growth Management Act, designate and protect natural resource lands including agricultural, forest, and mineral lands that have long-term significance to conserve and protect these areas.

Policy Action

- C-3.1 Recognize the importance of farmland conservation and local food production in maintaining the quality of life and long-term sustainability of Tumwater.
- C-3.2 Zone designated agricultural lands at very low densities to ensure the conservation of the resource for continued agricultural use.
- C-3.3 Limit non-agricultural development within designated agricultural areas to nonprime farmland soils where possible.
- C-3.4 Work with community groups to support the continued viability of agriculture and encourage community support for it.

C-3.4.1 Support the efforts of the Thurston Food System Council to develop a vibrant food system through access to healthy, local, affordable, culturally appropriate, sustainably produced food to assist the community in having reliable access to sufficient quantity of affordable nutritious food.

- C-3.5 Ensure that harvesting for conversion to other uses occurs in a manner compatible with land uses of the surrounding area and maintenance of water quality and environmentally critical areas.
- C-3.6 Allow mineral extraction industries to locate where prime natural resource deposits exist.
- C-3.7 Conserve designated mineral resource lands of long-term commercial significance for mineral extraction, and the use of adjacent lands should not interfere with the continued use of the designated mining sites that are being operated in accordance with applicable best management practices and other laws and regulations.

C-3.8 Restore mineral extraction sites as the site is being mined. The site should be restored for appropriate future use and it should blend with the adjacent landscape and contours.

76

2025 Comprehensive Plan Update Conservation Element

Balancing Nature and Community: Tumwater's Path to Sustainable Growth



General Government Committee, July 9, 2025

- Review format of Parts 1 and 2
- Discuss the draft Conservation Element
- Take feedback on goals, policies, and draft implementation actions
- Explain next steps





Format for Part 1- Goals, Policies, and Actions



Chapter 1- Introduction



Chapter 2- GMA & Element Goals



Chapter 3- County-Wide Planning Policies



Chapter 4- Element Goals and Policies





Appendix A- Draft Implementation Actions

Format for Part 2- Technical Information



Chapter 1- Introduction

Chapter 2- Natural Resources



Chapter 3- Critical Areas



Appendix A- Foundational Documents and Best Available Science





Appendix B- Open Space Taxation Act Summary



Appendix C- Tumwater Soils Report



Natural Resources

Urban Agriculture



Introduction/Overview

Sustainable Agriculture

Classification

Identification

Current Agricultural Uses

Small Scale Urban Agriculture



Forest Lands









Forest Lands in the Urban Area





Urban Forestry Management Plan

Mineral Resource Lands



Introduction Classification Identification Protection



Item 6.

Working Together







Critical Areas

Wetlands



- 1. Introduction
- 2. Values and Benefits
- 3. Policies, Regulations and Inventories
- 4. Protection and Identification
- 5. Enhancement



Critical Aquifer Recharge Areas



- 1. Introduction
- 2. Classification and Concerns
- 3. Current Inventory
- 4. Protection



Frequently Flooded Areas



- 1. Introduction
- 2. Classification and Concerns
- 3. Protection
- 4. Mapping
- 5. Groundwater



Geologically Hazardous Areas



- 1. Introduction
- 2. Classification
- 3. Identification
- 4. Protection



Fish and Wildlife Habitat Conservation Areas



- 1. Introduction
- 2. Classification
- 3. Identification
- 4. Protection



Questions to Consider as Reviewing Element

- 1. Any general or specific comments about the goals?
- 2. Are there shifts in action timelines to suggest?
- 3. Do you foresee any unintended consequences or burdens associated with policies and actions?
- 4. Which could impact you or the things you care about most? What do you expect that impact might be?
- 5. Do you have any additional thoughts you'd like to share related to the goals and their associated policies and actions?





Overarching Environmental Goals

Recognize the significant role played by natural features and systems in determining the overall environmental quality and livability of Tumwater

- Natural systems make Tumwater livable
- Protect and enhance environment
- Use adopted plans



Goal C-2

Promote conservation of natural resources and the protection of the environment in cooperation with residents, property owners, other jurisdictions, and tribes

Highlights:

• Support education programs





Natural Resource Areas Goals

Item 6.

Goal C-3

Support urban agriculture for access to local food production

- Work with partners to ensure agricultural viability
- Support local food production



Goal C-4

Support urban forestry

- Canopy retention
- Conversions compatible with surrounding development pattern



Protect mineral resource lands

- Allow extraction where it will not cause degradation
- Protect sites from incompatible uses
- Restoration extraction sites





Critical Areas Goals

Goal C-6

Designate and protect critical areas in accordance with the Growth Management Act to protect functions and safeguard against threats to health, safety and property.

- Use best available science
- Protect critical areas and allow for compatible development
- Require identification and protection of critical areas prior to development



Item 6.

Protect and enhance water quality

- Enhance ecological functions
- Support restoration of stream channels and wetlands
- Allow public access for monitoring and education
- Balance habitat, water supply, recreation, and all needs,



Item 6.

Improve natural drainage systems

- Develop watershed management plans and fish conservation measures
- Improve drainage systems for water quality
- Use best management practices to control erosion through construction and redevelopment



Improve air quality

- Support and entrust state programs
- Require appropriate vegetation retention
- Reduce vehicle miles traveled



Protect and enhance rivers, streams, and lakes

Highlights:

- Protect, enhance, and restore water channels
- Require mitigation for alterations



Item 6.

Protect and enhance wetlands

Highlights:

- Protect wetlands
- Use buffers to preserve functions
- Allow reasonable use
- Maintain stormwater treatment facilities and flow control



Item 6.

Goal C-12

Protect groundwater

- Protect aquifers, recharge areas, and wellheads
- Protect waterways



Protect geologically hazardous areas

- Regulate development intensity, site coverage, and vegetation
- Minimize soil disturbance
- Require erosion control throughout construction


Preserve floodplains and floodways

Highlights:

- Minimize changes that impact flow
- Require mitigation for engineered flood control measures
- Emphasize non- structure flood control
- Allow low-impact development only



Item 6.

Protect and enhance wildlife habitat

Highlights:

- Minimize fragmentation
- Protect habitat for listed species
- Encourage vegetated buffer areas
- Control invasive species



Next Steps in the Review Process

- Complete final round of stakeholder input and engagement
 - Late Spring 2025
 - Review draft goals, policies, and actions
 - Ensure input and feedback was incorporated
- Comprehensive Plan Adoption Process Fall 2025 Winter 2026
- Comprehensive Plan Ordinance Briefing Planning Commission – October 28, 2025



Submitting Comments or Questions

Written comments or questions are welcome at any time during the periodic update process

- Update website: <u>2025 Comprehensive Plan Update</u>
- Periodic update email: compplan@ci.tumwater.wa.us
- City of Tumwater contact:

 Community Development Department
 555 Israel Road SW
 Tumwater, WA 98501
 Phone: 360-754-4180
 Email: compplan@ci.tumwater.wa.us



Goals, Policies, and Implementation Actions

City of Tumwater 2025 Comprehensive Plan Balancing Nature and Community: Tumwater's Path to Sustainable Growth

DRAFT VERSION MAY 20, 2025

December 2025 Ordinance No. O2025-0XX





Part 1 – Goals, Policies, and Implementation Actions

Table of Contents

4
5
θ
θ
θ
7
g

Abbreviations Used in Document

CDD – Comr	nunity Dev	velopment	Department
------------	------------	-----------	------------

PRD – Parks,	Recreation,	and Facilities	Department
---------------------	-------------	----------------	------------

- **RCW** Revised Code of Washington
- TED Transportation & Engineering Department
- TMC Tumwater Municipal Code
- WRS Water Resources & Sustainability Department

2 | Page

Part 1 – Goals, Policies, and Implementation Actions

1. Introduction

A. Background

The Conservation Element is part of Tumwater's Comprehensive Plan. It was created to meet the State Growth Management Act (Chapter 36.70A RCW) requirements to identify and protect critical environmental areas and valuable natural resources.

Tumwater's Vision, Mission, and Belief Statements that provide overarching direction for the future of Tumwater are found in the Comprehensive Plan Summary.

The Conservation Element aligns and is consistent with the other elements of the Comprehensive Plan.

Part 1 – Goals Policies, and Implementation Actions establishes Tumwater's goals and policies to set forth a direction to identify, protect, and conserve critical environmental areas and valuable natural resources in Tumwater.

B. How to Read this Part of the Element

In Part 1, Chapters 2 and 3 discuss the Conservation Element's connection to the goals of the state Growth Management Act and the Thurston County-Wide Planning Policies.

Chapter 4 presents each goal with an explanation of how to read the tables and then presents each of Tumwater's conservation goals in detail with an explanation of the importance of each goal.

Appendix A contains the draft implementation actions, which are intended to be a source of annual work program items that serve to The goals and policies of the Conservation Element are guided by the state Growth Management Act and the Thurston County-Wide Planning Policies and the vision of a Tumwater that protects and enhances natural resources and the environment. The Element's goals and policies are coordinated with the other Elements and regional plans.

The Conservation Element's goals and policies are the policy basis for the draft implementation actions in Conservation Element and those future actions that will be developed over the next 20 years which will be the foundation for Tumwater's annual work programs to address natural resources and the environment.

Part 2 – Technical Information provides an analysis of natural resource lands appropriate to Tumwater and its critical areas to support the twenty-year growth projections for Tumwater.

implement the goals and policies of the

The annual work programs will further refine the

implementation actions prior to their being put

into practice. It is expected that implementation actions will be further amended, added, or

subtracted as needed over the course of the 20-

year Comprehensive Plan as new opportunities arise to meet the intent of the Conservation

Conservation Element.

Element's goals and policies.

Commented [BM1]: This will be the document that has the Comprehensive Plan wide goals and policies.

3 | Page



Part 1 – Goals, Policies, and Implementation Actions



2. Growth Management Act - Element Goals

Chapter 36.70A RCW requires that Tumwater show how the Conservation Element meets the relevant planning goals contained within the Act. The following is a listing of the two applicable goals for the Conservation Element and an analysis of how the Element addresses these goals:

 Natural resource industries. Maintain and enhance natural resource-based industries, including productive timber, agricultural, and fisheries industries. Encourage the conservation of productive forestlands and productive agricultural lands and discourage incompatible uses.

The Conservation Element has specific guidelines and policies that ensure the viability of natural resource industries and activities. Additionally, the Conservation Element ensures the viability of natural resource industries in Tumwater through the identification of such lands in the Conservation Element text and maps.

While Tumwater has limited natural resource lands as defined by the Growth Management

Act, it does have mineral resources, forestry, and agriculture lands. Since these parcels are within Tumwater, it is difficult to determine their long-term significance. Tumwater supports urban agriculture with goals, policies, and implementation actions.

10. Environment. Protect and enhance the environment and enhance the state's high quality of life, including air and water quality, and the availability of water.

The state legislature updated the Growth Management Act environment goal in 2023 to require enhancement of the environment. The Conservation Element contains specific policies relating to air and water quality, water availability, and protection and preservation of critical areas and addresses how to enhance the environment.

If conflict occurs in the implementation of policy and development regulations, the priority of protecting critical areas will be superior to other uses of natural resources.

4 | Page

Part 1 - Goals, Policies, and Implementation Actions

TUMWÄTE

3. County-Wide Planning Policies

The Growth Management Act requires that Tumwater's Comprehensive Plan be consistent with Thurston County's County-Wide Planning Policies, which were last amended in 2025.

The following is a list of the relevant policies that apply to this Element of the Comprehensive Plan. All County-Wide Planning Policies are adopted as Appendix B to the Comprehensive Plan. The relevant sections of the County-Wide Planning Policies to this Element are cited below.

II. Urban Growth Areas

The resource land and critical area chapters in the Element describe what kinds of development are compatible with the resource or area that chapter covers.

III. Promotion of Contiguous and Orderly Development, Provision of Urban Services, and Protection of Rural Areas The critical aquifer recharge area policies in the Element support this policy.

VII. Economic Development and Employment

Urban agriculture is discussed in the Element. The critical areas chapters address the protection of crucial areas for both environmental and public health reasons.

X. Environmental Quality

The Conservation Element is based upon the theme of the importance of natural systems and resources to human uses. The portion of the Element related to water resources cover the issues surrounding the balance between meeting future needs for water and protecting water resources.

5 | P a g e

Part 1 – Goals, Policies, and Implementation Actions

4. Element Goals and Policies

How to Read These Tables A.

The Conservation Element's goals and policies are not in priority order.

The nt's goals and policies are associated with one of the three areas of conservation goals:

- 1. Overarching Environmental Goals
- 2. Natural Resource Areas Goals

Appendix A provides a list of the draft implementation actions by goal and policy that will be considered when developing annual work programs for implementing the Conservation Element's goals and policies.

1) **Department Leads**

As noted in the tables below, implementation of the Conservation Element's goals and policies are associated with four different Tumwater departments:

Overarching Environmental Goals Β.

Goal C-1 Recognize the significant role played by natural features and systems in determining the overall environmental quality and livability of Tumwater.

Natural features and systems perform key environmental functions that enhance the

livability of Tumwater and protects its residents from hazards.

Community Development

Parks, Recreation, and Facilities

Transportation & Engineering

Each of the Conservation Element's policies is

associated with estimated start dates, length of

time to complete, and target completion dates, if appropriate based on adequate funding for

staff and resources. Most policies are ongoing

Water Resources & Sustainability

Department

Department

Department

Department

with no set target completion date.

CDD

PRD

TED

WRS

Period

2)

Policies	Lead	Period
C-1.1 Support active measures to protect and enhance Tumwater's natural environment	CDD	Term of Plan

Commented [BM2]: Old Goal C-1.

Commented [BM3]: Modified old Policy C-1.1.

6 | Page

	1.		.,	-		-	
Con	serv	/at	ior	۱E	le	me	en

- 3. Critical Areas Goals



Part 1 – Goals, Policies, and Implementation Actions

	Policies	Lead	Period
C-1.2	Use adopted plans to inform critical area policies, regulations, and implementation actions.	CDD WRS	Term of Plan

Goal C-2 Promote conservation of natural resources and the protection of the environment in cooperation with residents, property owners, other jurisdictions, and tribes.

Natural resources are the building blocks to produce all the things we need. Some resources need Tumwater's protection and management to ensure they can continue to provide materials and services into the future. Human activity has a large impact on the environment, changing the way the landscape meets the needs of others. Collaboration is key to ensuring that everyone's needs can be met in the future.

Policies	Lead	Period
C-2.1 Support education programs in the community that outline the need for natural resource conservation and protection of critical areas and create opportunities for community action.	CDD WRS	Term of Plan

Commented [BM5]: New policy.

Commented [BM4]: New policy to replace current

Policies C-1.3 and C-1.7.

C. Natural Resource Areas Goals

Goal C-3 Support urban agriculture for access to local food production.

Access to nutritious food is becoming more important as Tumwater residents face issues with the cost of living. Supporting urban agriculture like farmstands, community gardens, and urban farms helps provides residents with access to fresh locally grown products, decreases the cost of transporting goods, protects cultural heritage, and provides open space. As goals and policies are developed which support urban agriculture, farmers have increased access to markets, as well as opportunities to diversify their products and services. These benefits help make farms more viable at a scale that is compatible with urban environments.

7 | Page

Goal C-4 Support urban forestry.

Urban forestry is the care and management of the vegetation on public and private properties in Tumwater. It includes native plants and ornamental plants, understory, and canopy. Plants in Tumwater provide benefits to those who live, work, and play in the area.

Trees mitigate pollution, provide habitat for wildlife, sequester carbon, reduce stormwater runoff, and provide socioeconomic and aesthetic benefits. Managing trees to ensure safety and avoiding conflicts with infrastructure is an important part of urban forestry practices. The City Council adopted the Tumwater Urban Forestry Management Plan in 2021 to guide code development and protect trees and improve public safety.

	Policies	Lead	Period	
C-4.1	Maximize retention of a healthy tree cover and native vegetation and encourage restoration, replacement, and enhancement of unhealthy trees and disturbed vegetation as recommended in the adopted <i>Tumwater Urban Forestry Management Plan</i> consistent with the Growth Management Act and the requirements of protected habitat.	WRS	Term of Plan	Commented [BM8]: Update of current Policy C-1.5.
C-4.2	Ensure that harvesting for conversion to other uses occurs in a manner compatible with land uses of the surrounding area and maintenance of water quality and <mark>critical areas</mark> .	WRS CDD	Term of Plan	Commented [BM9]: Update of current Policy C-3.5.

Goal C-5 Protect mineral resource lands.

Modern industry relies on mineral resources. The most abundant mineral resource in Washington State is aggregate which includes sand, gravel, and crushed rock. These products are vital to road construction and other projects that require soil stabilization and additional drainage.

Local mineral sources are especially valuable since they reduce the shipping costs associated with transporting heavy materials. Since

Part 1 – Goals, Policies, and Implementation Actions

TUMWÄTER

minerals are a non-renewable resource, protecting existing sources in Tumwater is

important to ensure they are available locally into the future.

Policies	Lead	Period
C-5.1 Allow mineral extraction industries to locate where prime natural resource deposits exist and where extraction will not cause the degradation of sensitive areas	CDD	Term of Plan
C-5.2 Conserve designated mineral resource lands of long-term commercial significance for mineral extraction, and the use of adjacent lands should not interfere with the continued use of the designated mining sites that are being operated in accordance with applicable best management practices and other laws and regulations.	CDD	Term of Plan
C-5.3 Restore mineral extraction sites as the site is being mined. The site should be restored for appropriate future use, and it should blend with the adjacent landscape and contours.	CDD	Term of Plan

D. Critical Areas Goals

Goal C-6 Designate and protect critical areas including wetlands, critical aquifer recharge areas, frequently flooded areas, geologically hazardous areas, shorelines of the state, and fish and wildlife habitat conservation areas in accordance with the Growth Management Act to protect their functions and values as well as safeguard against threats to health, safety, and property.

Unwise development of critical areas jeopardizes environmental resource functions and values, puts species at risk of extinction, triggers regulatory burdens, and places people and property in unsafe conditions.

If critical area functions are not protected, attempting to enhance or restore them in the future is likely to be costly, if not impossible.

Policies	Lead	Period
C-6.1 Base critical areas designations and regulations on best available science to protect and enhance the functions and values of these areas.	CDD	Term of Plan

9 | Page

City of Tumwater 2025 Comprehensive Plan Balancing Nature and Community: Tumwater's Path to Sustainable Growth **Commented [BM13]:** Add tooltip to Comprehensive Plan glossary of terms.

-3.6.

-3.7.

-3.8.

Commented [BM14]: Updated current Policy C-2.1.

TMC 16.24.030 defines Critical aquifer recharge areas the same way the RCW defines it; could add this to the Comprehensive Plan glossary.

RCW states "Areas with a critical recharging effect on aquifers used for potable water."

Part 1 – Goals, Policies, and Implementation Actions

	Policies	Lead	Period
C-6.2	Protect critical areas while allowing for compatible growth and development.	CDD	Term of Plan
C-6.3	Require that prior to any development, critical areas are identified and protected.	CDD	Term of Plan

Commented [BM15]: Update of current policy C-1.1.

.6.

Goal C-7 Protect and enhance water quality.

Clean water is necessary for life. Everyday activities such as cleaning and driving can have a negative impact on water quality. Reducing the quantity of pollutants on Tumwater's roads and yards decreases the likelihood that they will be washed into waterways or carried into groundwater. Since most sources of pollution are all around Tumwater's landscape, it is also important to maintain systems that can remove pollution from water.

	Policies	Lead	Period	
C-7.1	Protect water quality by preserving and enhancing the ecological functions of water features through land use plans, development regulations, and public education.	CDD WRS	Term of Plan	Commented [BM17]: New policy.
C-7.2	Support restoration of river and stream channels and associated wetland and riparian areas to enhance water quality, improve fish and wildlife habitat, and mitigate flooding and erosion.	CDD WRS	Term of Plan	Commented [BM18]: Update of current Policy C-2
C-7.3	Allow public access to wetlands, streams, and lakes for scientific, educational, and recreational use, provided access is carefully sited, sensitive habitats and species are protected, and hydrologic continuity is maintained.	CDD	Term of Plan	Commented [BM19]: Update of current Policy C-2
C-7.4	Manage water resources to preserve ecosystem services, while addressing fish and wildlife habitat, flood protection, water supply, recreation, and open space.	CDD PRD WRS	Term of Plan	Commented [BM20]: New policy.

10 | Page

Commented [BM16]: Current Policy C-2.3.

Part 1 – Goals, Policies, and Implementation Actions

Policies	Lead	Period
C-7.5 Work with Olympia, Thurston County, and other affected entities to enhance and protect water quality in the region guided by the Deschutes River Study.	WRS	Term of Plan

Goal C-8 Improve natural drainage systems.

Storm and surface waters are a primary concern for urban environments. As impervious surfaces replace the permeable land cover, water travelling across the land accumulates pollutants. As development converts more land, less water is filtered through soil and more pollutants are accumulated. Drainage systems are also critical to protect the built environment from flood damage. With extreme amounts of stormwater entering the stormwater system within short periods of time, it is imperative that Tumwater expand the system at the rate of land conversion to meet the ever-changing needs. Commented [BM21]: New policy.

	Policies	Lead	Period	
C-8.1	Develop Watershed Management Plans in partnership with other jurisdictions who are also part of the same watershed.	WRS	Term of Plan	Commented [BM22]: New policy.
C-8.2	Develop and implement conservation measures necessary to preserve and enhance anadromous fish habitat and fisheries and other state and federally protected species, consistent with the Growth Management Act.	CDD	Term of Plan	Commented [BM23]: New policy.
C-8.3	Protect and enhance natural drainage systems to maintain and improve water quality and reduce public costs.	CDD WRS	Term of Plan	Commented [BM24]: New policy.
C-8.4	Prevent environmental degradation by using best management practices and current stormwater treatment and flow control standards on new and redevelopment projects.	CDD TED WRS	Term of Plan	Commented [BM25]: New policy.
C-8.5	Use current State approved stormwater treatment and flow control standards on development and construction projects.	CDD TED WRS	Term of Plan	Commented [BM26]: New policy.

11 | Page

Part 1 – Goals, Policies, and Implementation Actions

	Policies	Lead	Period
C-8.6	Require resource industries to use management practices that prevent erosion and sedimentation and pollutants from entering ground or surface waters following State regulations.	CDD	Term of Plan

Commented [BM27]: New policy.

Goal C-9 Improve air quality.

Poor air quality can have serious impacts on health for both people and the environment. Ensuring that air is appropriately filtered by retaining vegetation and reducing the amount of air pollution at the source are two strategies to manage air quality for the next 20 years.

Policies	Lead	Period	
C-9.1 Continue to support and rely on the various State, federal, and local programs to protect and enhance air quality.	CDD	Term of Plan	(
C-9.2 Require appropriate tree and vegetation retention and landscaping to provide filtering of suspended particulates.	CDD	Term of Plan	(
C-9.3 Coordinate with the other Elements in the Comprehensive Plan on actions to reduce vehicle miles traveled, greenhouse gas emissions, and other locally generated air pollutants.	CDD WRS TED	Term of Plan	(

Goal C-10 Protect and enhance rivers, streams, and lakes.

Surface waters like rivers, streams and lakes provide necessary transit for water as it flows across the landscape. Measures to protect waterways mitigate erosion and pollution that are ubiquitous in urban areas. Surface water also provides necessary habitat for wildlife. Protecting water quality is important for preserving Tumwater's environments to continue the beneficial ecosystem functions necessary for life.

Policies	Lead	Period
C-10.1 Protect, enhance, and restore natural stream channels for their hydraulic, ecological, and aesthetic functions through development regulations; land dedications, easements, and acquisitions; incentives; and restoration planning.	CDD WRS	Term of Plan

Commented [BM31]: New policy.

12 | Page

Part 1 – Goals, Policies, and Implementation Actions

	Policies	Lead	Period
C-10.2	Require any alteration of rivers, streams, and lakes include mitigation and maintenance which address water quality; floodplain protection; fish and wildlife habitat; channel stability and vegetative cover; maintenance of instream flows; and impacts to downstream property owners.	CDD	Term of Plan

Commented [BM32]: New policy.

Goal C-11 Protect and enhance wetlands.

Wetlands play a vital role in the overall natural environment. Wetlands protect water quality, reduce flooding, provide aquifer recharge for drinking water and other uses, and provide critical habitat for fish and wildlife as well as carbon sequestration. Wetlands also provide many social values including recreational opportunities, aesthetic benefits, places for research and education, and cultural resources important to tribes.

	Policies	Lead	Period
C-11.1	Where possible, protect wetlands as ecosystems that provide essential services and functionality within watersheds.	CDD	Term of Plan
C-11.2	Preserve and enhance wetlands from new developments by providing buffers around wetlands adequate to protect natural functions.	CDD	Term of Plan
C-11.3	Allow reasonable use of property containing existing wetlands to avoid a regulatory taking following State guidance.	CDD	Term of Plan
C-11.4	Maintain stormwater treatment and flow control sites associated with wetlands in a manner that ensures the ecological functions of the wetland to minimize impacts to critical area habitat.	CDD WRS	Term of Plan

Goal C-12 Protect groundwater.

Tumwater's water comes from underground aquifers, which are sand and gravel formations below the ground that "hold" water. These

aquifers are replenished by rain and snow infiltrated through the ground and filtered by the soil and trillions of microbes. The many layers of

13 | Page

City of Tumwater 2025 Comprehensive Plan Balancing Nature and Community: Tumwater's Path to Sustainable Growth Commented [BM37]: Update of current Policy C-2.9.

Commented [BM33]: Update of current Policy C-2.8.

Commented [BM34]: New policy.

Commented [BM35]: New policy.

Commented [BM36]: New policy.

Part 1 – Goals, Policies, and Implementation Actions

soil, rock, and microbes that sit above the aquifers act to cleanse the water as it passes and helps prevent contamination from the surface. Groundwater contributes to the flow of streams

and rivers, providing base flows that are essential for fish and other aquatic life, especially during dry periods.

	Policies	Lead	Period
C-12.1	Protect aquifers, aquifer recharge areas, and wellhead protection areas from <mark>contamination.</mark>	CDD WRS	Term of Plan
C-12.2	Protect streams, wetlands, and lakes that serve to recharge aquifers from contamination.	CDD WRS	Term of Plan



Commented [BM39]: New policy.

Goal C-13 Protect geologically hazardous areas.

Protecting geologically hazardous areas in Tumwater is important because they pose significant threats to public safety and infrastructure. Identifying these areas allows for better planning, mitigation, and emergency response. Washington state is particularly more vulnerable due to its geological location, active faults, and volcanic activity.

	Policies	Lead	Period	
C-13.1	Regulate development intensity, site coverage, and vegetation removal in geologically hazardous areas to prevent property damage and environmental degradation; minimize soil erosion, siltation, and landslides; and enhance greenbelt and wildlife habitat values,	CDD	Term of Plan	Commented [BM40]: Modification of current Policy C-
C-13.2	Minimize soil disturbance and maximize retention and replacement of native vegetative cover, such as trees, for land use permitted in erosion and landslide hazard areas.	CDD	Term of Plan	Commented [BM41]: Current Policy C-2.14.
C-13.3	Decrease development intensity as slopes increase to mitigate problems of drainage, erosion, siltation, and landslides.	CDD	Term of Plan	Commented [BM42]: New policy.
C-13.4	Require erosion and sedimentation prevention best management practices be used on construction projects by the Development Guide and Drainage Design & Erosion Control Manual	CDD	Term of Plan	Commented [BM43]: New policy.

14 | Page

Part 1 – Goals, Policies, and Implementation Actions

	Policies	Lead	Period
C-13.5 Rec con dan pos	quire appropriate engineering, building design, and Istruction measures to minimize the risk of structural nage and fire and injury to occupants, and to prevent It-seismic collapse in areas with severe <mark>seismic hazards</mark> .	CDD TED	Term of Plan

Commented [BM44]: Current Policy C-2.15.

Goal C-14 Preserve flood plains and floodways.

Floodplains are important for flood protection and ecosystems. Natural floodplains provide flood risk reduction benefits by slowing runoff, storing floodwater, and reducing erosion. Floodplains frequently contain wetlands and provide fish and wildlife habitats. It is important to protect the functions of floodplains and reduce risk to people and property by limiting development in frequently flooded areas.

	Policies	Lead	Period
C-14.1	Minimize land alterations that would increase potential flooding and changes to natural surface water features that retain or carry floodwaters, such as wetlands, floodplains, rivers, streams, and lakes.	CDD	Term of Plan
C-14.2	Require mitigation for adverse environmental impacts from engineered flood control measures.	CDD	Term of Plan
C-11.3	Meet state regulatory standards for floodplain development as these standards are updated for consistency with relevant federal requirements including those related to the Endangered Species Act	CDD	Term of Plan
C-14.4	Emphasize non-structural methods in planning for flood prevention and damage reduction. Require new developments or land modifications in 100-year floodplains maintain natural flood storage functions and minimize hazards.	CDD	Term of Plan

Part 1 – Goals, Policies, and Implementation Actions

	Policies	Lead	Period
C-14.5	Protect 100-year floodplains by restricting development; encouraging low-impact uses such as open space, trails, and parks; locating infrastructure above the flood level, and requiring new development replace existing flood storage capacity lost due to filling.	CDD	Term of Plan
C-14.6	Restrict permanent structures within the floodway due to risks associated with deep and fast-flowing waters unless appropriate flood control measures have been taken.	CDD	Term of Plan
C-14.7	Restrict land uses in a floodway that would divert water from the floodway, change flood elevation or obstruct natural flow, unless appropriate flood control measures have been taken such that there are no additional offsite impacts and no degradation of water quality.	CDD	Term of Plan
C-14.8	Restrict development in the floodway fringe that would reduce the existing level of flood storage.	CDD	Term of Plan

Goal C-15 Protect and enhance wildlife habitat.

Wildlife habitats, such as prairies, provide essential resources for wildlife and humans. Resources include food, water, shelter, and breeding grounds, ensuring the continuation of diverse species and the health of ecosystems. Wildlife habitat offers recreation opportunities for the community and has economic benefits as well. Integrating wildlife habitat protection into land use planning decisions can help minimize development impacts on wildlife populations.

Protecting habitats also helps mitigate climate change, conserve biodiversity conservation, and provide health and economic benefits.

Policies	Lead	Period
C-15.1 Identify, protect, and enhance fish and wildlife habitat corridors to minimize habitat fragmentation, especially along existing linkages by enhancing vegetation composition and structure and incorporating compatible indigenous plant species.	CDD	Term of Plan

Commented [BM53]: Modification of current Policy C-2.21.

16 | Page

Cons Part 1 – C	Goals, Policies, and Implementation Actions		TUMWÄTER	
	Policies	Lead	Period	
C-15.2	Protect and preserve habitats for species identified as endangered, threatened, or sensitive by the State or federal government, giving special consideration to measures necessary to preserve or <mark>enhance suitable habitat.</mark>	CDD	Term of Plan	Commented [BM54]: Modification of current Policy C-
C-15.3	Maintain habitats that support the greatest diversity of fish and wildlife through conservation and enhancement of <mark>critical areas</mark> .	CDD	Term of Plan	2.16. Commented [BM55]: Current Policy C-2.17.
C-15.4	When developing forested property adjacent to steep slopes, wetlands, stream ravines, or stream corridors, encourage development to provide additional buffer areas to provide wildlife and fish habitat.	CDD	Term of Plan	Commented [BM56]: New policy.
C-15.5	Restore native vegetation and control invasive species to preserve and enhance fish and wildlife habitat.	CDD	Term of Plan	Commented [BM57]: New policy.

17 | Page

Part 1 - Goals, Policies, and Implementation Actions



Appendix A – Draft Implementation Actions

Each of the Conservation Element goals and policies in Chapter 4 will require Tumwater to take specific actions to implement over the course of the 20-year term of the Comprehensive Plan.

The draft implementation actions in the tables below were developed in coordination with the stakeholders, the community, and Tumwater staff. The draft implementation actions in the tables below are intended to serve as the start for developing annual Tumwater work programs to address conservation actions.

1. How to Read These Tables

The Conservation Element goals, policies, and the draft implementation actions associated with them are not in priority order.

The draft implementation actions that can only be undertaken by other entities are not included in this plan.

The Conservation Element's goals, policies, and draft implementation actions are associated with one of the three areas of conservation goals:

- 1. Overarching Environmental Goals
- 2. Natural Resource Areas Goals
- 3. Critical Areas Goals

1) Department Leads

As noted in the tables below, implementation of the Conservation Element's policies and draft implementation actions are associated with four different Tumwater departments: As the Comprehensive Plan is put into action over the next twenty years, the draft implementation actions will change as new, unforeseen opportunities emerge: new ones may be added, proposed ones may be modified or replaced by other actions.

The draft implementations actions below are intended to be draft work program items that serve to implement the goals and policies of the Element. The draft implementation actions will need further refinement before they are incorporated into annual Tumwater work programs.

CDD	Community Development Department
PRD	Parks, Recreation, and Facilities Department
TED	Transportation & Engineering Department
WRS	Water Resources & Sustainability Department

2) Period

Each of the Conservation Element's policies and draft implementation are associated with estimated start dates, length of time to complete, and target completion dates, if appropriate based on adequate funding for staff and resources. Most policies are ongoing with no set target completion date.

18 | Page

Part 1 – Goals, Policies, and Implementation Actions

2. Overarching Environmental Goals

Goal C-1 Recognize the significant role played by natural features and systems in determining the overall environmental quality and livability of Tumwater.

	I			
	Policies and Implementation Actions	Lead	Period	
C-1.1	Support active measures to protect and enhance Tumwater's natural environment	CDD	Term of Plan	Commented [BM59]: Modified old Policy C-1.1.
C-1.2	Use adopted plans to inform critical area policies, regulations, and implementation actions.	CDD WRS	Term of Plan	Commented [BM60]: New policy to replace current Policies C-1.3 and C-1.7.
C-1.2	2.1 Implement the actions identified in the Thurston Climate Mitigation Plan.	CDD WRS	Term of Plan	Commented [BM61]: Based on current Land Use policy.
C-1.	2.2 Implement the actions identified in the most recent version of the Natural Hazards Mitigation Plan for Thurston County.	CDD WRS	Term of Plan	Commented [BM62]: Based on current Land Use policy.
C-1.2	2.3 Implement the actions identified in the Tumwater Urban Forestry Management Plan.	CDD WRS	Term of Plan	Commented [BM63]: Based on current Land Use policy.

Commented [BM58]: Old Goal C-1.

Goal C-2 Promote conservation of natural resources and the protection of the environment in cooperation with residents, property owners, other jurisdictions, and tribes.

	Policies and Implementation Actions	Lead	Period	
C-2.1	Support education programs in the community that outline the need for natural resource conservation and protection of critical areas and create opportunities for community action	CDD WRS	Term of Plan	Commented IRM641: New policy
	cifical areas and create opportainties for community action.			Commented [BW04]. New policy.
C-2.2	1.1 Support the work of the Stream Team	WRS	Term of Plan	Commented [BM65]: New implementation action.
C-2.:	1.2 Support the work for the Tumwater Green Team.	WRS	Term of Plan	Commented [BM66]: New implementation action.

19 | Page

Part 1 – Goals, Policies, and Implementation Actions			TUMWÄTER	
	Policies and Implementation Actions	Lead	Period	
C-2.1.3	Support water conservation messaging and outreach.	WRS	Term of Plan	Commented [BM67]: New implementation action.
C-2.1.4	Support water quality messaging and outreach for wellhead and source protection.	WRS	Term of Plan	Commented [BM68]: New implementation action.

3. Natural Resource Areas Goals

Goal C-3 Support urban agriculture for access to local food production.

Policies and Implementation Actions		Lead	Period	
C-3.1 Su ar	upport local food production to maintain the quality of life nd long-term sustainability of Tumwater <mark>.</mark>	CDD	Term of Plan	Commented [BM69]: Modification of current Policy C-
C-3.1.1	Implement and keep the Food System Plan updated.	CDD	Term of Plan	3.1. Commented [BM70]: New implementation action.
C-3.1.2	Collaborate with local organizations to promote food system resources to improve participation.	CDD	Term of Plan	Commented [BM71]: New implementation action.
C-3.2 W via co	ork with community groups to support the continued ability of agriculture and encourage neighborhood and mmunity support.	CDD	Term of Plan	Commented [BM72]: Modification of current Policy C-
C-3.2.1	Support the efforts of organizations to develop a vibrant food system through access to healthy, local, affordable, culturally appropriate, sustainably produced food to assist the community in having reliable access to enough affordable putritious food	CDD	Term of Plan	3.4.
				Implementation Action C-3.4.1.

20 | Page

Part 1 – Goals, Policies, and Implementation Actions

Goal C-4 Support urban forestry.

Policies and Implementation Actions
C-4.1 Maximize retention of a healthy tree cover and native vegetation and encourage restoration, replacement, and enhancement of unhealthy trees and disturbed vegetation as recommended in the adopted <i>Tumwater Urban Forestry Management Plan</i> consistent with the Growth Management Act and the requirements of protected habitat.
C-4.1.1 Implement the actions in the Tumwater Urban Forestry Management Plan.
C-4.2 Ensure that harvesting for conversion to other uses occurs in a manner compatible with land uses of the surrounding area and maintenance of water quality and critical areas.

Goal C-5 Protect mineral resource lands.

	Policies and Implementation Actions	Lead	Period	
C-5.1	Allow mineral extraction industries to locate where prime natural resource deposits exist and where extraction will not cause the degradation of sensitive areas.	CDD	Term of Plan	Commented [BM77]: Current Policy C-3.6.
C-5.2	Conserve designated mineral resource lands of long-term commercial significance for mineral extraction, and the use of adjacent lands should not interfere with the continued use of the designated mining sites that are being operated in accordance with applicable best management practices and other laws and regulations.	CDD	Term of Plan	Commented [BM78]: Current Policy C-3.7.
C-5.3	Restore mineral extraction sites as the site is being mined. The site should be restored for appropriate future use, and it should blend with the adjacent landscape and contours.	CDD	Term of Plan	Commented [BM79]: Current Policy C-3.8.

21 | Page

Part 1 – Goals, Policies, and Implementation Actions

4. Critical Areas Goals

Goal C-6 Designate and protect critical areas including wetlands, critical aquifer recharge areas, frequently flooded areas, geologically hazardous areas, shorelines of the state, and fish and wildlife habitat conservation areas in accordance with the Growth Management Act to protect their functions and values as well as safeguard against threats to health, safety, and property.

	Policies and Implementation Actions	Lead	Period		
C-6.1	Base critical areas designations and regulations on best available science to protect and enhance the functions and values of these areas	CDD	Term of Plan		Commented [BM80]: Add tooltip to Comprehensive Plan glossary of terms.
C-6.	1.1 Update critical area regulations as needed to reflect current best available science and state guidance to minimize impacts from new development.	CDD	Term of Plan		Commented [BM81]: Updated current Policy C-2.1. TMC 16.24.030 defines Critical aquifer recharge areas the same way the RCW defines it; could add this to the Comprehensive Plan glossary.
C-6.2	Protect critical areas while allowing for compatible growth and development.	CDD	Term of Plan		RCW states "Areas with a critical recharging effect on aquifers used for potable water."
					Commented [BM83]: Update of current policy C-1.1.
C-6.	2.1 When updating development regulations, review in conjunction with critical areas regulations to ensure compatibility.	CDD	Term of Plan		Commented [BM84]: New implementation action.
C-6.	2.2 Use incentive programs, acquisition, appropriate regulations, and other techniques to preserve critical areas as permanent open space where development may pose hazards to health, property, or important	CDD	Term of Plan		
	ecological functions.				Commented [BM85]: New implementation action based on current Policy C-2.2.
C-6.	2.3 Apply mitigation sequencing to avoid and minimize				Commented [BM86]: Add tooltip to Comprehensive Plan
	impacts to critical areas, and, if necessary, require and	CDD	Term of Plan		glossary of terms.
	functions		-		Commented [BM87]: New implementation action based
					Commented (DM001: New implementation action based

Commented [BM88]: New implementation action based on current Policy C-2.5.

22 | Page

Conserv Part 1 – Goals,	/ation Element Policies, and Implementation Actions		TUMWÄTER	
	Policies and Implementation Actions	Lead	Period	
C-6.2.4	Ensure the effectiveness of critical area mitigation by requiring adequate critical area studies and mitigation plans, the application of mitigation sequencing, financial assurances to ensure mitigation success, and by improving Tumwater oversight of maintenance and monitoring of mitigation.	CDD	Term of Plan	Commented [BM89]: New implementation action based
C-6.3 Req ide	uire that prior to any development, critical areas are ntified and protected.	CDD	Term of Plan	on current Policy C-2.4. Commented [BM90]: Current Policy C-2.3.
C-6.3.1	In reviewing development proposals that may impact a critical area, supplement staff review with a qualified third-party professional as needed to assess potential impacts and require development alternatives or mitigation.	CDD	Term of Plan	Commented [BM91]: New implementation action.

Goal C-7 Protect and enhance water quality.

	Policies and Implementation Actions	Lead	Period	
C-7.1	Protect water quality by preserving and enhancing the ecological functions of water features through land use plans, development regulations, and public education.	CDD WRS	Term of Plan	Commented [BM92]: New policy.
C-7.1	1 Require adequate stormwater treatment and flow control for new development, including low impact development techniques.	CDD	Term of Plan	Commented [BM93]: New implementation action.
C-7.1	2 Update stormwater regulations as needed to reflect current state guidance, including low impact development provisions.	CDD WRS	Term of Plan	Commented [BM94]: New implementation action.
C-7.1	3 Support water quality messaging and outreach for wellhead and source protection.	WRS	Term of Plan	

Conservation Element Part 1 – Goals, Policies, and Implementa	tion Actions		TUMWÄTER	
Policies and Imple	mentation Actions	Lead	Period	
C-7.2 Support restoration of rive associated wetland and rip quality, improve fish and w flooding and erosion.	r and stream channels and parian areas to enhance water vildlife habitat, and mitigate	CDD WRS	Term of Plan	Commented [BM95]: Update of current Policy C-2.6.
C-7.3 Allow public access to wet scientific, educational, and is carefully sited, sensitive protected, and hydrologic	lands, streams, and lakes for l recreational use, provided access habitats and species are continuity is <mark>maintained.</mark>	CDD	Term of Plan	Commented [BM96]: Update of current Policy C-2.7.
C-7.3.1 Work with the Tumwa LOTT, and other comr public education on n	ater and Olympia School Districts, nunity-based organizations on naintaining water quality.	WRS	Term of Plan	Commented [BM97]: New implementation action.
C-7.4 Manage water resources to while addressing fish and v water supply, recreation, a	o preserve ecosystem services, wildlife habitat, flood protection, nd <mark>open space</mark> .	CDD PRD WRS	Term of Plan	Commented [BM98]: New policy.
C-7.4.1 Update the Salmon C	reek Basin Plan	CDD TED WRS	Term of Plan	Commented [BM99]: New implementation action.
C-7.4.2 Monitor storm drain of pursue corrective active	outfalls on a regular basis and ions, <mark>as necessary.</mark>	WRS	Term of Plan	Commented [BM100]: New implementation action.
C-7.5 Work with Olympia, Thurs entities to enhance and pr guided by the Deschutes R	ton County, and other affected otect water quality in the region iver Study <mark>.</mark>	WRS	Term of Plan	Commented [BM101]: New policy.
C-7.5.1 Coordinate implementa Olympia and Thurston	ation strategies and regulations with County.	CDD TED WRS	Term of Plan	Commented [BM102]: New implementation action.

Goal C-8 Improve natural drainage systems.

Policies and Implementation Actions	Lead	Period
C-8.1 Develop Watershed Management Plans in partnership with other jurisdictions who are also part of the same watershed.	WRS	Term of Plan

Commented [BM103]: New policy.

24 | Page

Cons Part 1 – C	Goals,	Policies, and Implementation Actions		TUMWÄTER	
		Policies and Implementation Actions	Lead	Period	
C-8.	.1.1	Work with Olympia and Thurston County to ensure that regulations regarding surface water management provide consistent surface water management.	WRS	Term of Plan	Commented [BM104]: New implementation acti
C-8.	.1.2	Continue involvement with watershed planning efforts through participation in the Salmon Creek Basin and WRIA 13 watershed planning efforts.	WRS	Term of Plan	Commented [BM105]: New implementation acti
C-8.2	Dev pres stat Gro	elop and implement conservation measures necessary to serve and enhance anadromous fish habitat and other e and federally protected species, consistent with the wth Management Act	CDD	Term of Plan	Commented [BM106]: New policy.
C-8.3	Prot and	tect and enhance natural drainage systems to maintain improve water quality and reduce public costs.	CDD WRS	Term of Plan	Commented [BM107]: New policy.
C-8.4	Prev mar and	vent environmental degradation by using best nagement practices and current stormwater treatment flow control standards on new and redevelopment	CDD TED WRS	Term of Plan	Commented (BM108): New policy
C-8.5	Use	current State approved stormwater treatment and flow trol standards on development and construction projects.	CDD TED WRS	Term of Plan	Commented [BM109]: New policy.
C-8.	.5.1	Update stormwater regulations as needed to reflect current best available science and state guidance.	CDD TED WRS	Term of Plan	Commented [BM110]: New implementation acti
C-8.6	Req that ente regu	uire resource industries to use management practices prevent erosion and sedimentation and pollutants from ering ground or surface waters following State ulations.	CDD	Term of Plan	Commented [BM111]: New policy.
C-8.	.6.1	Update regulations as needed to meet state standards to minimize the amount of erosion, sedimentation, and water pollutants created by resource industries.	CDD	Term of Plan	Commented [BM112]: New implementation acti

25 | Page

Part 1 – Goals, Policies, and Implementation Actions

Goal C-9	Improve air quality.			
	Policies and Implementation Actions	Lead	Period	
C-9.1 Co and	ntinue to support and rely on the various State, federal, d local programs to protect and enhance air quality.	CDD	Term of Plan	Commented [BM113]: New policy.
C-9.1.1	Work with the Olympic Region Clean Air Agency and with Federal and State agencies to ensure that air quality is improved and protected within Tumwater.	CDD	Term of Plan	Commented [BM114]: New implementation action.
C-9.2 Real	quire appropriate tree and vegetation retention and adscaping to provide filtering of suspended particulates.	CDD	Term of Plan	Commented [BM115]: New policy.
C-9.2.1	Update regulations as needed to reflect the current best available science and state guidance that allow the use of existing vegetated areas for biofiltration.	CDD	Term of Plan	Commented [BM116]: New implementation action.
C-9.3 Co Pla gas	ordinate with the other Elements in the Comprehensive an on actions to reduce vehicle miles traveled, greenhouse s emissions, and other locally generated air pollutants.	CDD WRS TED	Term of Plan	Commented [BM117]: New policy.

Goal C-10 Protect and enhance rivers, streams, and lakes.

	Policies and Implementation Actions	Lead	Period
C-10.1	Protect, enhance, and restore natural stream channels for their hydraulic, ecological, and aesthetic functions through development regulations; land dedications, easements, and acquisitions; incentives; and restoration planning.	CDD WRS	Term of Plan
C-10.2	Require any alteration of rivers, streams, and lakes include mitigation and maintenance which address water quality; floodplain protection; fish and wildlife habitat; channel stability and vegetative cover; maintenance of instream flows; and impacts to downstream property owners.	CDD	Term of Plan
C-10	D.2.1 Rehabilitate degraded river and stream channels and banks using public and private programs, where conditions permit.	CDD WRS	Term of Plan

26 | Page

Item 6

Conservation	Element
--------------	---------

Part 1 – Goals, Policies, a	and Implementation Actions
-----------------------------	----------------------------

	Policies and Implementation Actions	Lead	Period
C-10.2.2	Work with tribes, nonprofit organizations, and other public agencies to implement programs to rehabilitate rivers, streams, and lakes.	WRS	Term of Plan

Goal C-11 Protect and enhance wetlands.

protection areas from contamination.

	Policies and Implementation Actions	Lead	Period	
C-11.1	Where possible, protect wetlands as ecosystems that provide essential services and functionality within watersheds.	CDD	Term of Plan	Commented [BM122]: Update of current Policy C-2.8.
C-11.2	Preserve and enhance wetlands from new developments by providing buffers around wetlands adequate to protect natural functions.	CDD	Term of Plan	Commented [BM123]: New policy.
C-11.3	Allow reasonable use of property containing existing wetlands to avoid a regulatory taking following State guidance.	CDD	Term of Plan	Commented [BM124]: New policy.
C-11.4	Maintain stormwater treatment and flow control sites associated with wetlands in a manner that ensures the ecological functions of the wetland to minimize impacts to critical area habitat.	CDD WRS	Term of Plan	Commented [BM125]: New policy.
C-11	.4.1 Ensure the water level fluctuations within wetland areas due to stormwater treatment and flow control objectives are maintained like natural conditions as part of new development.	CDD WRS	Term of Plan	Commented [BM126]: New implementation action.
Goal C	12 Protect groundwater.			Commented [BM127]: Update of current Policy C-2.9.
	Policies and Implementation Actions	Lead	Period	
C-12.1	Protect aquifers, aquifer recharge areas, and wellhead	CDD	Term of	

27 | Page

City of Tumwater 2025 Comprehensive Plan Balancing Nature and Community: Tumwater's Path to Sustainable Growth

WRS

Plan

Commented [BM128]: New policy.

Item 6.

Conserv Part 1 – Goals,	Vation Element Policies, and Implementation Actions		TUMWÄTER	
	Policies and Implementation Actions	Lead	Period	
C-12.1.1	Work with the State Department of Ecology and others to update the delineation of aquifer recharge areas and regulations.	CDD WRS	Term of Plan	Commented [BM129]: New implementation action.
C-12.1.2	Update the wellhead protection plan as new source locations are considered or implemented.	WRS	Term of Plan	
C-12.2 Prot aqu	tect streams, wetlands, and lakes that serve to recharge ifers from contamination.	CDD WRS	Term of Plan	Commented [BM130]: New policy.

Goal C-13 Protect geologically hazardous areas.

Policies a	nd Implementation Actions	Lead	Period
C-13.1 Regulate develo vegetation remo prevent propert minimize soil en greenbelt and w	pment intensity, site coverage, and val in geologically hazardous areas to y damage and environmental degradation; osion, siltation, and landslides; and enhance ildlife habitat values.	CDD	Term of Plan
C-13.2 Minimize soil di replacement of land use permit	sturbance and maximize retention and native vegetative cover, such as trees, for ed in erosion and landslide hazard areas,	CDD	Term of Plan
C-13.3 Decrease develo mitigate probler landslides.	pment intensity as slopes increase to ns of drainage, erosion, siltation, and	CDD	Term of Plan
C-13.4 Require contrac prevention best projects by the I Erosion Control	cors to use erosion and sedimentation management practices on construction Development Guide and Drainage Design & Manual <mark>.</mark>	CDD	Term of Plan
C-13.5 Require approprion construction me damage and fire post-seismic col	iate engineering, building design, and asures to minimize the risk of structural and injury to occupants, and to prevent apse in areas with severe seismic hazards.	CDD TED	Term of Plan

28 | Page

area.

art 1 – Goals, Policies, and Implementat	ion Actions	
Policies and Impler	mentation Actions	Lead
C-13.5.1 Update building and fi current state standard	ire codes as needed to reflect Is to account for the severity and	CDD

frequency of seismic activity in the south Puget Sound

Goal C-14 Preserve flood plains and floodways.

	Policies and Implementation Actions	Lead	Period
C-14.1	Minimize land alterations that would increase potential flooding and changes to natural surface water features that retain or carry floodwaters, such as wetlands, floodplains, rivers, streams, and lakes.	CDD	Term of Plan
C-14.2	Require mitigation for adverse environmental impacts from engineered flood control measures.	CDD	Term of Plan
C-14	1.2.1 Prior to authorizing engineered flood control measures, encourage non-structural methods of flood control utilizing best available science.	CDD	Term of Plan
C-14.3	Meet state regulatory standards for floodplain development as these standards are updated for consistency with relevant federal requirements including those related to the Endangered Species Act	CDD	Term of Plan
C-14.4	Emphasize non-structural methods in planning for flood prevention and damage reduction. Require new developments or land modifications in 100-year floodplains maintain natural flood storage functions and minimize hazards.	CDD	Term of Plan
C-14.5	Protect 100-year floodplains by restricting development; encouraging low-impact uses such as open space, trails, and parks; locating infrastructure above the flood level, and requiring new development replace existing flood storage capacity lost due to filling.	CDD	Term of Plan

CDD

Period

Term of Plan

Commented [BM136]: New implementation action.

29 | Page

Cons Part 1 – G	ervation Element bals, Policies, and Implementation Actions		TUMWATER	
	Policies and Implementation Actions	Lead	Period	
C-14	5.1 Update regulations as needed to reduce development in floodplains following Federal Emergency Management Agency standards.	CDD	Every five years. Next update 2028.	Commented [BM143]: New implementation action.
C-14.6	Restrict permanent structures within the floodway due to risks associated with deep and fast-flowing waters unless appropriate flood control measures have been taken.	CDD	Term of Plan	Commented [BM144]: New policy.
C-14.7	Restrict land uses in a floodway that would divert water from the floodway, change flood elevation or obstruct natural flow, unless appropriate flood control measures have been taken such that there are no additional offsite impacts and no degradation of water quality.	CDD	Term of Plan	Commented IRM1451: New policy
C-14.8	Restrict development in the floodway fringe that would reduce the existing level of flood storage.	CDD	Term of Plan	Commented [BM146]: New policy.

Goal C-15 Protect and enhance wildlife habitat.

	Policies and Implementation Actions	Lead	Period	
C-15.1	Identify, protect, and enhance fish and wildlife habitat corridors to minimize habitat fragmentation, especially along existing linkages by enhancing vegetation composition and structure and incorporating compatible indigenous plant species.	CDD	Term of Plan	Commented [BM147]: Modification of current P
C-15.2	Protect and preserve habitats for species identified as endangered, threatened, or sensitive by the State or federal government, giving special consideration to measures necessary to preserve or enhance suitable habitat.	CDD	Term of Plan	2.21. Commented [BM148]: Modification of current P
C-15	5.2.1 Implement salmon habitat protection and restoration priorities in approved Water Resource Inventory Area 13 and 23 plans.	CDD WRS	Term of Plan	Commented [BM149]: New implementation acti

30 | Page

Conserv Part 1 – Goals,	Policies, and Implementation Actions		TUHWATER	
	Policies and Implementation Actions	Lead	Period	
C-15.2.2	Implement suitable habitat protections and restoration priorities for threatened and endangered aquatic and terrestrial plant and animal species.	CDD	Term of Plan	Commented [BM150]: New implementation action.
C-15.2.3	Complete and implement the Bush Prairie Habitat Conservation Plan.	CDD	Term of Plan	Commented [BM151]: New implementation action.
C-15.3 Mai and criti	ntain habitats that support the greatest diversity of fish wildlife through conservation and enhancement of cal areas	CDD	Term of Plan	Commented [BM152]: Current Policy C-2.17.
C-15.3.1	Coordinate with adjacent jurisdictions and tribes to identify, protect, and develop enhancement plans and actions for habitat networks and wetlands that cross jurisdictional lines.	CDD	Term of Plan	Commented [BM153]: New implementation action
C-15.3.2	Promote the enhancement or restoration of streams, rivers, lakes, and wetlands as adjacent development activities occur.	CDD	Term of Plan	based on current Policy C-2.19. Commented [BM154]: New implementation action.
C-15.4 Wh wet dev wild	en developing forested property adjacent to steep slopes, lands, stream ravines, or stream corridors, encourage elopment to provide additional buffer areas to provide llife and fish habitat.	CDD	Term of Plan	Commented [BM155]: New policy.
C-15.5 Res	tore native vegetation and control invasive species to serve and enhance fish and wildlife habitat	CDD	Term of Plan	Commented [BM156]: New policy.
C-15.5.1	Develop regulations requiring all new developments to remove invasive plants and use native vegetation in restoration of buffers around wetlands, streams, creeks, and steep slope areas.	CDD	Term of Plan	Commented [BM157]: New implementation action.

31 | Page

Part 2 – Technical Information

City of Tumwater 2025 Comprehensive Plan

Balancing Nature and Community: Tumwater's Path to Sustainable Growth

DRAFT VERSION MAY 20, 2025

December 2025

Ordinance No. O2025-0XX


Part 2 – Technical Information



Table of Contents

1.	Introduct	tion	4
A	. Backgr	ound	4
В.	How to	o Read this Part of the Element	5
C.	Best A	vailable Science	6
D	. Shorel	ines of the State	6
2.	Natural F	Resources	7
A	Backgr	ound	7
В.	Urban	Agriculture	7
C.	Forest	Lands 1	0
C.	Minera	al Resource Lands	4
3.	Critical A	reas1	8
A	. Backgr	ound1	8
В.	Wetlar	nd Areas1	8
C.	Critica	I Aquifer Recharge Areas 2	3
D	. Freque	ently Flooded Areas	8
E.	Geolog	gically Hazardous Areas	0
F.	Fish ar	nd Wildlife Habitat Conservation Areas3	2
Арр	endix A	Foundational Documents and Best Available Science	6
1.	Genera	al Policy	6
2.	Critica	I Areas	7
3.	Resou	rce Lands	0
Арр	endix B	Open Space Taxation Act Summary 4	1
Appendix C		Tumwater Soils Report	2

List of Tables

Table C-1.	State Private Forest Land Grades	11
Table C-2.	Forest Resource Land Identification.	13
Table C-3.	Forest Lands Designation Considerations.	13
Table C-4.	Mineral Resource Land Identification.	16

Part 2 – Technical Information



Table C-5. Considerations for Mineral Lands Designation in Tumwater.	17
Table C-5. General Policy Foundational Documents for the Conservation Element	36
Table C-6. Critical Areas Foundational Documents for the Conservation Element.	37
Table C-7. Resource Lands Foundational Documents for the Conservation Element.	40

List of Figures

Figure C-1. Agricultural Soil Classifications in Tumwater.	
0	-

List of Maps

- Map C-1. Forestry Lands Designation Map
- Map C-2. Mineral Resource Lands Map
- Map C-3. Critical Aquifer Recharge Areas Map
- Map C-4. Frequently Flooded Areas Map
- Map C-5. Geologically Hazardous Areas Map
- Map C-6. Shorelines of the State Map
- Map C-7. Wetlands Map

Abbreviations Used in Document

- **CDD** Community Development Department
- RCW Revised Code of Washington
- TMC Tumwater Municipal Code
- WAC Washington Administrative Code
- WRS Water Resources & Sustainability Department

Part 2 – Technical Information

1. Introduction

A. Background

The Conservation Element is part of Tumwater's Comprehensive Plan. It was created to meet the State Growth Management Act (Chapter 36.70A RCW) requirements to identify and protect critical environmental areas and valuable natural resources.

The Element addresses:

- 1. Natural Resource Lands Conservation
 - Agricultural Lands
 - Forest Lands
 - Mineral Resource Lands
- 2. Critical Areas Protection
 - Wetland Areas
 - Critical Aquifer Recharge Areas
 - Frequently Flooded Areas
 - Geologically Hazardous Areas
 - Fish and Wildlife Habitat Conservation Areas

The Growth Management Act requires that Tumwater demonstrate that each Element in its Comprehensive Plan meets the relevant fifteen planning goals contained within the Act. The fifteen goals in turn guide the development and adoption of Tumwater's Comprehensive Plan and development regulations.

The Conservation Element addresses the two Growth Management Act goals related to the environment and natural resources:



CONSERVATION

State requirements (WAC 365-196-405) that the Comprehensive Plan must meet:

- Designation of the proposed general distribution and general location and extent of the uses of land, where appropriate, for agricultural, timber, and mineral production of long-term commercial significance.
- The general location of any known critical areas that limit the suitability of land for development.
- Provisions for the protection of the quality and quantity of ground water used for public water supplies.
- A review of drainage, flooding, and stormwater runoff in the area covered by the plan and nearby jurisdictions, and guidance for corrective actions to mitigate or cleanse those discharges that pollute waters of the state, including Puget Sound or waters entering Puget Sound.
- Influences or threats to the quality and quantity of ground water used for public water supplies.
- 8. Natural resource industries. Maintain and enhance natural resource-based industries, including productive timber, agricultural, and fisheries industries. Encourage the conservation of productive forestlands and productive

City of Tumwater 2025 Comprehensive Plan Balancing Nature and Community: Tumwater's Path to Sustainable Growth

4 | Page

Part 2 – Technical Information



agricultural lands, and discourage incompatible uses.

The Conservation Element has specific guidelines and policies that ensure the viability of natural resource industries and activities. Additionally, the Conservation Element ensures the viability of natural resource industries in Tumwater through the identification of such lands in the Conservation Element text and maps.

While Tumwater has limited natural resource lands as defined by the Growth Management Act, it does have mineral resources, forestry, and agriculture lands. Tumwater supports urban forestry and agriculture appropriate and compatible with other land use goals, policies, and implementation actions.

10. Environment. Protect and enhance the environment and enhance the state's high quality of life, including air and water quality, and the availability of water. The State Legislature updated the Growth Management Act environment goal in 2023 to require enhancement of the environment.

The Conservation Element contains specific policies relating to air and water quality, water availability, and protection and preservation of critical areas and enhance the environment. Areas of environmental sensitivity are designated as open space or a lower intensity land use designation than other areas of Tumwater.

If conflict occurs in the implementation of planning and development regulations, the priority of protecting critical areas will be superior to other uses of natural resources.

The Conservation Element and implementing ordinances were developed with public input as described in the Public Outreach Plan required by the Growth Management Act. The Element is based on the updated list of additional supporting plans, documents, and best available science found in Appendix A.

B. How to Read this Part of the Element

Part 2 of the Conservation Element consists of the following chapters.

- Chapter 2 Natural Resources: Provides a summary of Tumwater's current mineral resource lands and discusses urban forestry and urban agriculture.
- Chapter 3 Critical Areas: Provides a summary of critical areas in Tumwater, including wetlands, critical aquifer recharge areas, frequently flooded areas,

geologically hazardous areas, and fish and wildlife conservation areas. Each critical area has its own section, discussing regulations, classifications, and locations within Tumwater.

 Appendix A – Foundational Documents and Best Available Science: Provides a list of the documents used to create the Conservation Element's Technical Summary.

Part 2 – Technical Information

C. Best Available Science

RCW 36.70A.172 and WAC 365-195-900 through WAC 365-195-925 require Tumwater to use best available science in revising or adopting new policies and regulations related to critical areas.

Utilization of best available science is particularly important to wetland and riparian area protection, floodplain areas and salmon recovery efforts required under the Endangered

D. Shorelines of the State

For shorelines of the state, the goals, and policies of the Shoreline Management Act (RCW 90.58.020) were added as one of the goals of the Growth Management Act (RCW 36.70A.020) without creating an order of priority among the fifteen goals. The goals and policies of Tumwater's Shoreline Master Program approved under RCW 90.58 shall be considered an element of the Comprehensive Plan.

The shorelines of the state as identified by the Shoreline Management Act within Tumwater include the Deschutes River, and Black Lake Drainage area as well as Trosper Lake, Barnes Lake, Lake Susan, and Munn Lake. Shorelines of Species Act. Best available science is a process to assist jurisdictions in ascertaining what science is appropriate for use in basing policy and regulatory decision-making.

Tumwater uses best available science in all revisions and additions to critical areas policies and regulations to protect the functions and values of critical areas.

the state within Tumwater's urban growth area include Black Lake.

Shorelines of the state also include the upland or shorelands that extend 200 feet landward from the edge of these waters, and any wetlands, floodways, and floodplain areas associated with such waters.

Tumwater's updated Shoreline Master Program was adopted in 2019 following review and approval from the State Department of Ecology. The Shoreline Master Program incorporated the existing critical area regulations from TMC Chapter 16.20, TMC Chapter 16.28, and TMC Chapter 16.32.



Part 2 – Technical Information

2. Natural Resources

A. Background

Natural resources are the materials that occur in nature and are used by humans to meet their needs. The European settlement of Tumwater was based on its location adjacent an abundant water source for industry and transport. The trees around Tumwater were harvested and used for construction, wood pulp, and other products.

Today, natural resources are still a part of Tumwater's urban environment. Having access to locally grown fresh food, forests for wood products, and mineral resources for transportation and other infrastructure reduces the cost of transporting these essential goods from far away. It also creates jobs in harvesting and manufacturing these raw materials into goods.

Natural Resources serve many benefits besides

B. Urban Agriculture

1) Introduction

Protecting agricultural resource lands in rural areas is prioritized in the Growth Management Act. While Tumwater is developing to urban levels to protect the long-term sustainability of agricultural resource lands outside of the urban area, there are limited areas of Tumwater that currently contain agricultural uses.

Access to healthy food choices is an important public health issue. Lack of healthy food choices contributes to health problems such as obesity, diabetes, heart disease, and cancer. Access to healthy food and local food production are use. Trees sequester carbon, filter air, provide shade, absorb water, and provide habitat. Agriculture improves soil fertility, manages water infiltration, reduces soil erosion, and provides habitat.

Harvesting natural resources can also have impacts on neighboring ecosystems, properties, and human health. Air pollution, erosion, and habitat loss are some of the more detrimental disturbances.

Identifying lands where humans should extract and use resources that will have less impact on the landscape and on other humans is an important part of planning. This section considers natural resources in Tumwater, their value for residents, and considers which lands are most appropriate for natural resource use in an urban area.

clearly part of planning for a vital, healthy community.

The Tumwater City Council's Strategic Plan has several goals and policies directly related to environmental sustainability and increasing the availability of healthy, locally grown food. Long distance transportation consumes an enormous amount of fossil fuel and generates a great deal of greenhouse gases. Increased local food production has a direct beneficial effect on the environment by reducing greenhouse gas emissions. Transportation costs are much lower for local food producers.



Part 2 – Technical Information

TUMWÄTER

In addition, a direct benefit to the community is the provision of fresh, healthy, locally grown food. Encouraging a wide range of local food production options compatible with an urban environment are important policy decisions in furthering the sustainability goals of Tumwater.

2) Sustainable Urban Agriculture

Sustainable urban agriculture can take a variety of forms, some of which are listed below.

- 1. Urban Farm. An urban farm is where plants and/or some animals are grown for sale of the plants and animals or their products, and in which the plants and animals or their products are sold either on the lot where they are grown or off site, or both. Examples may include flower and vegetable raising, orchards and vineyards. Urban farms are small-scale agricultural uses that are appropriate for an urban area and compatible with other urban land uses.
- 2. **Community Garden.** A community garden means land managed by a public or nonprofit organization, or a group of individuals, which is used to grow plants and harvest food or ornamental crops from them for donation or use by those cultivating the land and their households.
- 3. Individual Home Garden. A home garden is a garden grown on a residential lot as an accessory use to the primary use for consumption by the occupants.
- 4. **Farmers Market.** A farmers market consists of a group of individual venders primarily selling locally grown produce and products drawn from the region. This

use typically is seasonal and may be temporary. Some examples are set up on closed streets or on portions of sites used for other primary uses.

3) Agricultural Lands Classification

The Conservation Element's classification and identification of agricultural lands of long-term significance is based upon the land-capability classification system of the U.S. Department of Agriculture Handbook No. 210, which utilizes soil characteristics to determine capacity. The classes of agricultural lands are based upon consideration of growing capacity, productivity, and soil composition of the land.

The reference standard for defining categories of agricultural lands of long-term significance is the use of prime and unique farmland soils classifications as mapped by the Natural Resource Conservation Service in the Web Soil Survey.

The Conservation Element recognizes that under the Growth Management Act, the prime agricultural lands in Tumwater and its urban growth areas have developed for urban uses in large part to protect the agricultural lands of long-term significance in rural Thurston County from development pressure.

These circumstances do not allow for a classification of agricultural lands of long-term significance to be applied to Tumwater or its urban growth area, which is intended for urban growth under WAC 365-190-050(3)(a).

4) Agricultural Lands Identification

The U.S. Department of Agriculture updates the soil classification and surveys annually. The

Part 2 – Technical Information



Natural Resource Conservation Service soil survey report for Tumwater is included as Appendix A.

The Natural Resource Conservation Service Soil Survey Report for Tumwater and its urban growth area provides information on the various soils as surveyed in the area. Classifications for soil are contained in the report as well as information about the characteristics of the soil and suitability for land use types. The percentage of prime soil, soils of state significance, and other agriculture land uses classes are provided in Table C-1 below.

While this information is valuable for open space and parcels without development, most of the soil

Figure C-1. Agricultural Soil Classifications in Tumwater.

in Tumwater is or will be covered by urban land uses. Information on soil is developed from the Natural Resource Conservation Service Soil Survey of Thurston County. Soils in the classifications including where drained and where irrigated were not included as those actions do not align with other conservation goals within this section.

Urban soils are separately identified as they have different characteristics than natural soils. Part of the reason for this is the impact of human activity from adding fill material to make it more suitable for specific development or to modify the landscape for specific activities.



Soil Classifications in Tumwater

- Prime Farmland (7%)
- Farmland of Statewide Importance (16%)
- Not Prime Farmland (7%)
- Prime Farmland if drained (12%)
- Prime Farmland if irrigated (58%)

Source: Web Soil Survey, Natural Resources Conservation Service, U.S. Department of Agriculture. Accessed April 2025.

5) Current Agriculture Uses

Currently within Tumwater there are several agriculture operations identified by the State Department of Agriculture. Several landowners produce hay, pasture, lawn and turf, nursery, and other crops. One of two local farmstand and corn maze agritourism businesses and a grocery retailer focused on local food are located within Tumwater.

Due to the current use, there are parcels within Tumwater that are designated as agriculture and

Part 2 – Technical Information



This incentive-based program promotes the retention of lands used for agriculture by reducing the tax rate for these parcels and restricting the use to agriculture. Future landowners are deterred from changing the land use as doing so results in a penalty. Map C-1 shows the properties enrolled in these programs within Tumwater.

6) Small Scale Urban Agriculture

There are other land use factors that impact small scale agriculture in urban areas and should be considered when regulating food systems. Encouraging small-scale urban farms and community gardens is one way Tumwater can play a role in ensuring food access, food security, and overall environmental sustainability into the future.

While lands for agriculture can be found within Tumwater, the cost of land and the economic pressure for development make it difficult to sustain large scale agriculture businesses within Tumwater.

Existing Tumwater support for small scale agriculture includes:

- The Right-to-Farm Ordinance (Ordinance No. 1276, effective 1991) protected legally established agricultural facilities.
- The Urban Agriculture Ordinance

C. Forest Lands

1) Introduction

Protecting forest lands in rural areas is prioritized in the Growth Management Act. While Tumwater (Ordinance No. O2010-029, effective 2011) established regulations for the following:

- New agriculture uses are allowed within the Airport Related Industry, Light Industrial, Residential/Sensitive Resource, Single Family Low, Single Family Medium, and Multifamily Medium zone districts provided they are thirty acres or less in size that meet specific requirements.
- Farmers markets as a permitted use in all commercial and industrial zone districts.
- Community gardens as a permitted use in all residential, commercial, and industrial zone districts, except for the Manufactured Home Park zone district.
- Permitting small scale farm animal options such as apiaries, poultry, and rabbits within Tumwater.
- Taller fences for agricultural uses.

In 2024 the City Council approved a project to review the food system in Tumwater which included a review of regulatory barriers and prepare a plan to address next steps.

is developing to urban levels to protect the longterm sustainability of forest resource lands outside of the urban area, there are limited areas of Tumwater that are classified as forest resource

City of Tumwater 2025 Comprehensive Plan Balancing Nature and Community: Tumwater's Path to Sustainable Growth



Part 2 – Technical Information

lands.

Applying best management practices on forest land creates environmental benefits such as:

- Improving water and air quality
- Reducing soil erosion
- Decreasing damage from storms and floods
- Protecting wildlife habitat
- Providing scenic and recreational spaces

2) Forest Lands Classification

The Growth Management Act requires Tumwater to classify current forest resource lands. The classification of the forest lands is based upon the private forest land grades of the State Department of Revenue (WAC 458-40-530).

This classification system incorporates consideration of growing capacity, productivity, and soil composition of the land. These factors are calculated and expressed as a site index. Forest lands of long-term commercial significance generally have higher private forest land grades. If lower private forest land grades exist within areas of predominately higher grades, the land may be designation as forest land.

Identifying lands suitable for forest land designation in Tumwater balanced the state requirement that growth occur in urban areas against the review of the dominant species and site index to determine the land grade. Table C-1 shows land grades for Western Washington forests.

Washington State Species: Westside	Private Forest Site Index	Land Grade
	136 feet and over	1
	118 to 135 feet	2
Douglas Fir	99 to 117 feet	3
	84 to 98 feet	4
	Under 84 feet	5
	136 feet and over	1
	116 to 135 feet	2
Wostern Hemiock	98 to 115 feet	3
western Hemiock	83 to 97 feet	4
	68 to 82 feet	5
	Under 68 feet	6
Pod Aldor	117 feet and over	6
Reu Aluel	Under 117 feet	7

Table C-1. State Private Forest Land Grades.

City of Tumwater 2025 Comprehensive Plan Balancing Nature and Community: Tumwater's Path to Sustainable Growth



Part 2 – Technical Information



Washington State Species: Westside	Private Forest Site Index	Land Grade
	Marginal forest productivity	7 or 8
	Noncommercial	8

Source: WAC 458-40-530

Notes: Land Grade 1 = highest, Land Grade 8 = lowest. The marginal forest productivity in Tumwater is Land Grade 8

Forest lands are further defined by operability classes based upon characteristics of soils and slope. The criteria are as follows:

- Class 1 Favorable. Stable soils that slope less than thirty percent. Forest operations do not significantly affect soil productivity and soil erosion. Forest operations, such as road building and logging, are carried out with minimal limitations.
- Class 2 Average. Stable soils that slope less than thirty percent, but on which significant soil erosion, compaction, and displacement may occur because of forest operations.
- **Class 3 Difficult**. Soils with one or both of the following characteristics:
 - a. Stable soils that slope between 30 and
 65 percent; and
 - b. Soils that slope between zero and 65 percent but display evidence that rapid mass movement may occur as a direct result of forest operations.
- Class 4 Extreme. All soils that slope more than 65 percent.

3) Current Forest Land Use

Forestry occurs at many different scales in Tumwater. Common forestry activities include maintaining existing trees through pruning and non-native plant removal, planting new trees in parks and streetscaping, and tree removal.

Within Tumwater, there are also stands of trees which are currently used for commercial harvest. These stands of trees are harvested and replanted to harvest again in cycles to produce timber for the many wood products made, purchased, or used locally.

There are parcels within Tumwater that are designated for forestry use and are eligible for tax reduction programs through the State Department of Revenue Open Space Program. This incentive-based program promotes the retention of land used for forestry by reducing the tax rate for these parcels and restricting the use to agriculture.

Future landowners are deterred from changing the land use as doing so results in a penalty. Map C-2 and Table C-2 show the properties enrolled in these programs within Tumwater.

Thurston County Assessor public records show that seven parcels within Tumwater and its urban growth area are designated forest land. The total area of all designated forest land in Tumwater is 371.81 acres. More than half of the parcels are owned by a large timber company that manages many parcels for forest products.

There are also lands which are not identified by the assessor nor enrolled in the program where forest practices occur.

```
12 | Page
```

Part 2 – Technical Information



Table C-2. Forest Resource Land Identification.				
Parcel Number	Parcel Area	Adjacent Land Use Designation	Adjacent Current Land Use	
11718330000	37.95	Low Density Residential, Neighborhood Commercial, Open Space	Residential	
12721330000	38	Light Industrial	Residential, Commercial, Industrial	
12820340000	120	Light Industrial, Low Density Residential, Heavy Industrial	Natural Resource, Residential	
12829210000	30	Light Industrial, Low Density Residential, Heavy Industrial	Natural Resource, Residential	
12829230000	120	Light Industrial, Low Density Residential, Heavy Industrial	Natural Resource, Residential	
12829410000	16.01	Green Belt, Low Density Residential, Light Industrial	Residential, Parks, Vacant	
12832420500	9.85	Low Density Residential	Residential, Vacant, Commercial, Industrial, Utilities	
51620100100	9.59	Light Industrial	Residential, Commercial, Industrial	

Sources: Thurston County Assessor and Thurston Regional Planning Council, 2025.

4) Forest Lands Conservation

Development and urbanization are the leading causes of forest lands conversion. The Growth Management Act protects forest resources in the rural areas by directing urban development within Tumwater and its urban growth area.

Table C-2 shows the conditions for conversion of forest lands to urban land uses.

Table C-3.	Forest Lands	Designation	Considerations.
------------	--------------	-------------	-----------------

Considerations	Tumwater Conditions
The availability of transportation and other necessary public services and facilities	Services and facilities available
Forest lands are located outside the urban and suburban areas and rural settlements	Located within an urban area
Parcel size: forest lands consist of predominantly large parcels	Parcels identified are small to medium in size
The compatibility of adjacent and nearby land use and settlement patterns with forest lands of long-term commercial significance	Adjacent land uses of urban land use intensity
Property is assessed as open space or forest land	Seven parcels identified

Part 2 – Technical Information



pursuant to Chapter 84.33 RCW or Chapter 84.34 RCW	
Local economic conditions	Not supportive in the long term of forest land designation
History of land development permits issued nearby	Parcels harvested under state conversion permits are processed by Tumwater

5) Forest Lands in the Urban Area

The Growth Management Act (WAC 365-190-060(2)) locates forests of long-term commercial significance outside of urban growth areas. Forest lands in Tumwater and its urban growth area are not considered in the Conservation Element to be of long-term significance and are not planned for designation as protected resource lands.

Tumwater continues to manage urban forest resources for the many benefits they provide to residents and the larger region. Several goals and policies in the Comprehensive Plan align urban forest practices with other planning requirements and community interests.

While no forest lands of long-term significance are currently identified, the parcels of land that are currently forested are encouraged to remain forested for their environmental and open space benefits, as long as possible, before conversion to urban land uses.

6) Urban Forestry Management Plan

The Tumwater Urban Forest Management Plan establishes goals to ensure healthy trees on Tumwater's urban landscape into the future. The Plan identified a number of action items to support its goals, which include:

- Restore and enhance the community and urban forest.
- Protect and preserve the community and urban forest, which includes trees, understory, habitat, and soils.
- Balance the protection and support of the community and urban forest with other City strategic priorities, which include, in part, providing affordable housing, developing a walkable urban community, economic development, addressing climate change, and protecting endangered species.

C. Mineral Resource Lands

1) Introduction

While metallic, nonmetallic and coal are found and mined in the state, currently the most valuable mineral resource is aggregate. Gravel, sand, and crushed stone are examples of aggregate resources used for transportation, residential, commercial, and industrial construction. The state is the seventh highest sand and gravel producer in the nation. Identifying and conserving local sources for aggregate is important since hauling these materials requires large amounts of fuel due to the weight and heavy vehicle route maintenance.

14 | Page

Part 2 – Technical Information

As with other types of resource lands discussed in this plan, the identification and conservation of mineral resource lands is a requirement of the Growth Management Act. The Conservation Element identifies and classifies mineral resource lands from which the extraction of minerals currently occurs or may occur in the future as well as a strategy to ensure a future supply of these minerals is maintained.

2) Mineral Resources Lands Classification

In defining what lands qualify as mineral resource lands, the Conservation Element bases its methodology upon WAC 365-190-070(3) with modification to include consideration of environmentally sensitive areas, existing land uses, and land ownership.

Tumwater classifies mineral resource lands using the following minimum guidelines:

- <u>Geology</u> The land in question should contain deposits consisting of sand and gravel, coal, sandstone, basalt, or other igneous rock that is recoverable and marketable, based on U.S. Geological Survey maps or site-specific information prepared by a geologist, or as indicated by the State Department of Natural Resources mining permit data.
- <u>Projected Life of the Resource</u> To be designated as mineral resource land there should be sufficient mineral resource to be commercially sustainable for a number of years under market conditions present when the designation is being considered.
- <u>Proximity to Point of Use or Market</u> Consideration for the energy cost of

transporting the extracted materials to the market or end user.

- <u>Infrastructure</u> Consideration for the availability of public roads and whether there is a source of water.
- <u>Current and Future Land Use</u> Extractive industries should locate where prime natural resource deposits exist, provided these sites are separated by buffer strips from existing residential areas and restored for appropriate re-use after removing the resource material.

When classifying these areas, maps and information on the location and extent of mineral deposits provided by the State Department of Natural Resources, U.S. Geological Service and any relevant information provided by property owners should be utilized. Critical areas, other environmentally sensitive areas, and cultural resources are also important criterion to consider.

3) Mineral Resources Lands Identification

The Conservation Element identifies lands with long-term commercial significance for extracting mineral resources.

The State Department of Natural Resources maintains maps and classification systems for mineral resources which aligns with state guidance for mineral lands designation in WAC 365.190.040(4) and WAC 365.190.070(3). Aggregate Resource Maps are made using existing geologic maps, subsurface data, materials-testing data, and publicly available mining data to classify potential sources of aggregate. These classifications vary based on size, quality, and uncertainty of the resource. Table C-4 and Map C-



Part 2 – Technical Information

3 show the mineral resources in Tumwater and its

urban growth area.

Table C-4. Mineral Resource Land Identification.

Parcel Number	Parcel Area	Adjacent Land Use Designation	Adjacent Current Land Use
11706320100	23.75	Open Space	Natural Resource, Undeveloped Land
11706330000	21.85	Open Space	Natural Resource, Undeveloped Land
11706330100	42.28	Open Space	Natural Resource, Undeveloped Land
12713240100	13.03	Open Space, Light Industrial, Low Density Residential	Parks, residential, Undeveloped Land
12715120900	6.00	Light Industrial, Medium Density Residential, Low Density Residential	Natural Resource
12829120200	15.03	Light Industrial, Heavy Industrial	Commercial, Industrial, Natural Resource
12829130200	3.74	Heavy Industrial, Light Industrial Green Belt	Natural Resource, Undeveloped Land
12829130201	9.10	Heavy Industrial, Light Industrial Green Belt	Natural Resource, Undeveloped Land
12829130202	0.90	Heavy Industrial, Light Industrial Green Belt	Natural Resource, Undeveloped Land
12829130203	5.81	Heavy Industrial, Light Industrial Green Belt	Natural Resource, Undeveloped Land
12829210100	10.06	Heavy Industrial, Light Industrial	Natural Resource,
12829310000	36.57	Heavy Industrial, Light Industrial, Low Density Residential	Natural Resource, Undeveloped Land, Commercial, Industrial
12829320200	27.40	Heavy Industrial, Low Density Residential	Natural Resource, Undeveloped Land
12829320201	12.72	Heavy Industrial, Low Density Residential	Natural Resource, Undeveloped Land
12829340200	4.97	Heavy Industrial, Low Density Residential	Natural Resource, Undeveloped Land, Commercial, Industrial
63050001900	13.55	Light Industrial	Natural Resource, Commercial, Industrial, Residential, Undeveloped Land



Part 2 – Technical Information



Sources: Thurston County Assessor and Thurston Regional Planning Council, 2025.

In Tumwater, the Heavy Industrial land use designation and associated zoning district was specifically created to support the existing rock quarry use on Black Lake Boulevard. New mineral extraction uses are not permitted anywhere else in Tumwater. Mineral resource lands identified are subject to consideration of the effects of proximity to population areas and the possibility of more intense uses of land as shown in Table C-5.

 Table C-5. Considerations for Mineral Lands Designation in Tumwater.

Considerations for Mineral Lands Designation	Conditions in Tumwater
General land use patterns in the area	Urban
Availability of utilities	Available
Availability and adequacy of water supply	Available
Surrounding parcel sizes and surrounding uses	Small-medium sized parcels, land uses industrial and commercial in nature
Availability of public roads and other public services	Available
Division or zoning for urban or small lots	Yes
Accessibility and/or distance from point of use	Close to use sites
Physical and topographic characteristics of the mineral resource site	Accommodating to low operating costs
Depth of the resource	Exposed at surface
Depth of the overburden	Exposed materials
Physical properties of the resource	High grade gravel, sand, and rock
Life of the resource	10 to 100 years
Resource availability in the region	Good for sand/gravel, limited to rock
Surrounding critical areas	Site by site variation
Impact to endangered species habitat	Site by site variation
Energy costs of transporting minerals	Dependent upon location

4) Mineral Resources Lands Protection

TMC Chapter 16.16 protects legally established mineral resource extraction facilities.

Future discoveries of mineral resources, or market conditions that are conducive, may encourage the

opening of new mineral resource extraction operations. What or where these facilities would be located cannot be accurately gauged. A newly established mineral resource extraction facility must be a land use identified within the land use designations and zone districts applying to the site.

Part 2 – Technical Information

3. Critical Areas

A. Background

Tumwater's environment is comprised of both natural and built features. Lakes, mature trees, steep slopes, natural vegetation, streams, wetlands, and prairies are parts of the natural environment within Tumwater. Tumwater's history and name reflects the importance of the natural environment to the community identity.

As the need for more jobs, housing, public services, transportation, utilities, and recreation increase, the protection of the natural environment becomes more important.

Tumwater must continually assess the relationship between the natural and built environments and evaluate the potential impacts of development on the environment and the community. Maintaining a quality natural environment in Tumwater depends on coordinated actions between government, the private sector, and individuals.

The Conservation Element guides City effort to balance nature and community, creating a path to sustainable growth. It is intended to meet the objectives of the Growth management Act, the federal Endangered Species Act, State Environmental Policy Act, County-Wide Planning

Policies for Thurston County, and other applicable federal, state, and county policies.

This Element also provides guidance for reducing the risks to people, property, and the environment posed by geological and flood hazard areas. Tumwater's Appendix to the Thurston County Hazard Mitigation Plan provides additional mitigation strategies and background information about natural hazards.

TMC Title 16 promotes the maintenance, enhancement, and preservation of critical areas and environmentally sensitive natural systems by avoiding or minimizing adverse impacts from construction and development.

Under the Growth Management Act, Tumwater is required to use the best available science when reviewing and revising policies and regulations for critical areas. The plans and regulations designed to protect critical areas are not intended to deny reasonable use of private and public property, but to assure that development on or near critical areas is accomplished in a manner that is sensitive to the environmental resources of the community.

B. Wetland Areas

1) Introduction

Wetlands serve many important ecological functions. They act as natural reservoirs for flooding and stormwater runoff; protect water quality by filtering out pollutants; help stabilize shorelines; provide areas for groundwater recharge; provide fish and wildlife habitat; provide open space and recreation opportunities; and provide areas for scientific study and education.

Wetlands preservation can significantly reduce public and private costs associated with

18 | Page



Part 2 – Technical Information

downstream flooding, poor water quality, and diminished wildlife habitat.

Through the Conservation Element and associated regulations, Tumwater will:

- Preserve, protect, manage, and regulate wetlands for the purpose of promoting public health, safety and general welfare while conserving fish, wildlife, and other natural resources;
- Protect the ecological and economic benefits to the public of wetlands functions and values;
- Regulate property use and development to maintain the natural and economic benefits provided by wetlands;
- Protect private property rights consistent with the public interest; and
- Provide for protection against direct and indirect wetlands impacts by providing regulatory authority for management of wetland buffers.

It is the short-term goal of this policy to achieve no net loss of the remaining wetlands in Tumwater, defined by acreage and function. It is the longterm goal to improve the health of Tumwater's existing wetlands and create wetlands, where feasible, to increase the quantity and quality of wetlands in Tumwater.

2) Wetland Values and Benefits

Wetlands serve many important ecological functions. A summary of wetland benefits follows:

• Wetlands slow and store floodwater. Riverine wetlands and floodplains provide flat areas where floodwaters can spread out and slow down, reducing the height and velocity of floods. Floodwater trapped in wetlands may then slowly drain, reducing stream bank erosion and downstream peaks.

- Wetlands provide erosion control for shorelines by dissipating the water's energy and stabilizing shorelines with the root systems of plants commonly found in wetlands.
- Wetlands improve water quality by their ability to filter out sediments, nutrients, and toxic chemicals. Moving water carries suspended sediments and other materials. As the water enters a wetland and slows down, these sediments tend to settle down. The sediments are then trapped by wetland vegetation, which in turn reduces the amount of siltation deposited in lakes and reservoirs.
- Wetlands allow water to soak into the underlying soil, which adds to the supply of groundwater.
- Wetlands provide essential areas for waterfowl and migratory shorebirds to rest and feed.
- Wetlands provide essential escape covers and feeding, nesting, and breeding habitat for many species of fish and wildlife, including the Oregon spotted frog. Wetland plants help protect juvenile fish, thereby serving to increase the anadromous fish population.
- Wetlands furnish areas for education and research of a variety of flora and fauna that cannot be found in other



Part 2 – Technical Information



environments.

• Wetlands provide open space and recreation opportunities, including fishing, hiking, boating, and bird watching.

Existing Wetland Policies, Regulations, and Inventories

Several federal, state, and local wetland policies, regulations, and inventories currently form a patchwork for wetlands protection.

a) Federal Clean Water Act

The Federal Clean Water Act is a broad-based law covering water pollution control in general. Section 404 of the Act requires the Army Corp of Engineers to regulate the dredging and filling of waters of the United States, including the tributaries and wetlands. However, the dredging, draining or land clearing of wetlands without a nexus to waters of the United States, including their tributaries and wetlands, is not addressed by the Act.

b) State Shoreline Management Act

The State Shoreline Management Act regulates activities in shorelines of the state, which include lakes over 20 acres in size, rivers, and streams with flows more than 20 cubic feet per second (c.f.s.), and all lands within 200 feet of the ordinary high water mark and any wetlands, floodways, and/or floodplain areas associated with such waters.

The Act excludes wetlands not associated with shorelines of the state, including isolated wetlands and riparian wetlands associated with lakes less than 20 acres and streams with flows less than 20 c.f.s. It also exempts most agricultural and forest practices from permit requirements.

Tumwater adopted an updated Shoreline Master

Program in April 2014, and it was subsequently amended on December 3, 2019.

Tumwater's Shoreline Master Program requires that wetland buffers be determined by the category and function level of the wetland as stated in the version of TMC Chapter 16.28 adopted as part of the Shoreline Master Program.

c) State Hydraulics Code

Any work that uses, diverts, obstructs, or changes the natural flow or bed of any salt or freshwaters of the state requires Hydraulics Project Approval. The State Department of Fish and Wildlife administers the State Hydraulics Code through Hydraulic Project Approval process. The intent of the Code is to protect fish and fish habitat.

Wetlands outside the ordinary high-water mark and isolated wetlands without fish life are excluded. A Hydraulic Project Approval does not address impacts to wetland functions and values other than fish and fish habitat.

d) State Wetland Rating System for Western Washington

The manual is currently the definitive methodology for determining when a wetland is present and where a wetland boundary is located. It is based on the functional values present in the wetland, sensitivity to disturbance, significance, rarity, and ability to replace. The use of the most current manual during project review is consistent with using the best available science.

e) National Wetlands Inventory

Conducted on a national level using aerial photographs, the National Wetlands Inventory depicts wetland locations, approximate boundaries, and includes classification by wetland type. The inventory is available for Tumwater, but

Part 2 – Technical Information



it should not be presumed to locate every wetland area in Tumwater. Often the only reliable method for wetland identification is a site visit by a qualified wetland biologist following the process in TMC Chapter 16.28. This is typically done in conjunction with a development proposal.

f) Wetland Mapping for the Thurston Region

The Thurston Regional Planning Council has identified wetlands in Thurston County based on color infrared aerial photographs. In many cases, the results of aerial photography have been verified by field surveys. The result is digitized maps showing wetlands boundaries and types. This inventory must be supplemented with site specific field surveys to verify wetland boundaries at the time of development permit review.

g) Environmental Policy

TMC Chapter 16.04 adopts the State Environmental Policy Act with amendments. The intent of this code is to identify and if necessary, mitigate the environmental impacts associated with a variety of actions.

h) Wetlands Protection Standards

TMC Chapter 16.28 establishes standards for the protection of wetlands. Wetlands in Tumwater are regulated under this chapter. Exemptions include intentionally created wetlands, such as stormwater treatment ponds, and certain unintentionally created wetlands.

i) Protection of Trees and Vegetation

TMC Chapter 16.08 regulates the clearing of land in Tumwater, including trees and vegetation located in wetlands. The Urban Forestry Management Plan was adopted on March 2, 2021. The intent of the plan was to guide policies and implementation actions for the maintenance and improvement of the urban tree canopy in Tumwater over the next 20 years.

j) Floodplain Regulations

Floodplains are regulated by TMC Chapter 18.38. The Floodplain Overlay prohibits or strictly limits filling and development in designated floodplains, including wetlands located within these areas. This reduces the height and velocity of floods and lessens bank erosion.

4) Wetland Protection Areas Classification

The Growth Management Act requires Tumwater to classify wetlands according to their sensitivity to disturbance, rarity, functions, and irreplaceability. Tumwater uses the *Washington State Wetland Rating System for Western Washington* for classifying wetlands as outlined below, which is further identified in TMC Chapter 16.28.

5) Wetland Identification

Identification of wetlands is undertaken by the permit application at the time an application for development is made, using the *Washington State Wetland Rating System for Western Washington* in its current form, and as hereafter amended.

6) Wetland Protection Techniques

Techniques used to protect wetland areas include:

- Using the Washington State Wetland Rating System for Western Washington for wetland classification based on function and value.
- Requiring a qualified wetland biologist to



determine wetland type and boundary for development sites containing wetlands.

- Establishing wetland buffers based on the relative value of the wetland in which no development or disturbance should occur.
- Striving to achieve no net loss of wetland areas and functions.
- Striving to create wetlands in the long term, where feasible, to increase the quantity and quality of wetlands.
- Attempting to avoid impacts to wetlands altogether if practicable.
- If impact avoidance is impossible, attempting to reduce wetland impacts through mitigation.
- If impact avoidance and reduction are impossible, accomplishing wetland compensation.
- Providing education on the value of wetlands to developers and homeowners.

7) Wetland Protection

WAC 365-190-040(1) states that when critical areas, including wetland areas, cannot be readily identified, these areas should be designated by performance standards or definitions. In this way, such areas can be specifically identified during the processing of a site-specific permit or development authorization.

For the purposes of wetland protection, a performance standard based process is followed.

The wetland protection standards in TMC Chapter 16.28 classify, designate, and protect wetlands and their associated buffers from on-site and offsite activities impacts. These regulations have provisions for reasonable wetland buffer areas and the means for avoidance and reduction of wetland impacts. Attributes of the wetland protection standards in TMC Chapter 16.28 include the following.

a) Wetland Buffer Areas

Wetland buffer areas are required to be adjacent to regulated wetlands to protect wetland functions and values. All wetland buffer widths are measured from the wetland boundary as established by a field survey conducted by a qualified wetland biologist. Wetland buffers are the primary means by which wetland functions and values are protected.

Wetland buffer widths may be increased, reduced, or averaged on a case-by-case basis in accordance with best available science when an altered buffer is necessary to protect wetland functions and values in accordance with TMC Chapter 16.28.

b) Wetland and Wetland Buffer Areas – Allowed Activities

Certain limited low-intensity activities may be permitted in wetland buffer areas without a wetlands permit provided that these activities are not prohibited by any other chapter or law, and they are conducted using best management practices, such as:

- Conservation or preservation of soil, water, vegetation, fish, shellfish, and other wildlife that does not entail changing the structure or functions of the existing wetland.
- Outdoor recreational activities, including fishing, bird watching, hiking, boating, horseback riding, swimming, canoeing, and bicycling.

Part 2 – Technical Information

- The harvesting of wild crops in a manner that is not injurious to natural reproduction of such crops and provided the harvesting does not require tilling of soil, planting of crops, or alteration of the wetland by changing existing topography, water conditions, or water sources.
- The maintenance of drainage ditches.
- Education, scientific research, and use of nature trails.
- Navigation aids and boundary markers.
- Site investigative work necessary for land use application submittals such as surveys, soil logs, percolation tests and other related activities. In every case, wetland impacts should be minimized and disturbed areas should be immediately restored.
- Normal maintenance, repair, or operation of existing serviceable structures, facilities, or improved areas. Maintenance and repair do not include any modification that changes the character, scope, or size of the original structure, facility, or improved area and does not include construction of a maintenance road.
- Minor modification of existing serviceable structures within a buffer where modification does not adversely impact wetland functions.

C. Critical Aquifer Recharge Areas

1) Introduction

All of Tumwater's drinking water supply comes from underground aquifer areas. The goal of establishing Critical Aquifer Recharge Areas is to

c) Reasonable Use of Wetlands and Wetland Buffers

Following state law, if an applicant for a proposed development demonstrates that application of these regulations would deny all reasonable use of the property, conditioned development may be allowed if the applicant demonstrates that the criteria in TMC Chapter 16.28 are met.

d) Wetland Replacement Ratios

As a condition of any permit allowing alteration of wetlands or wetland buffers, the applicant should engage in the restoration, creation, or enhancement of wetlands and their buffers to offset loss of wetland function and value. It is recognized that the alteration of wetlands and/or wetland buffers is not desirable. Creation, restoration and enhancement of wetlands or wetland buffers are extremely difficult to achieve. Wetland alteration should only occur when impact avoidance and reduction are impossible.

Wetland restoration, creation and enhancement acreage replacement ratios are identified in TMC Chapter 16.28.

8) Wetland Enhancement

The Growth Management Act requires that Tumwater not only protects but enhances its critical areas. As part of that process, the Conservation Element's goals, policies, and implementation actions in Part 1 address how Tumwater will be enhancing wetland resources.

protect the functions and values of Tumwater's drinking water by preventing pollution and maintaining supply.

Tumwater's water comes from underground



Part 2 – Technical Information



aquifers, which are gravel formations below the ground that hold water. These aquifers are replenished by rain and snow as they infiltrate through the ground and are filtered by the soil and trillions of microbes. The many layers of soil, rock, and microbes that sit above the aquifers act to cleanse the water as it passes and helps prevent contamination from the surface.

Tumwater's water system draws water out of these aquifers from multiple sources located around the City. As the water seeps into the aquifers, contaminants from above-ground activities can get into the water. Critical aquifer recharge areas are areas identified where groundwater seeps through the layers and collects. Certain land uses and organic and inorganic materials must be regulated to protect the water and keep it safe.

Public drinking water supply systems are regulated by the State Department of Health under the Safe Drinking Water Act. Generally, the state regulates systems with 15 or more residential connections, known as a Group A system.

Tumwater coordinates with Thurston County Environmental Health to monitor groundwater and report on conditions, including notification of any identified hazards. Periodic inspections have also been completed every few years of businesses that use hazardous materials onsite to ensure they are handled, stored, and disposed of properly.

2) Critical Aquifer Recharge Area Classification

The Growth Management Act requires Tumwater to protect public groundwater supplies so contamination can be avoided. Drinking water depends on groundwater availability, and so the amount of water in aquifers must be monitored to ensure water sources are replenished and are not depleted.

In addition, Chapter 246-290 WAC requires water protection and standards that address vulnerable sources of drinking water, such as Tumwater's Wellhead Protection Program.

All groundwater is vulnerable. Using criteria to create classifications or categories of vulnerability helps Tumwater apply the appropriate measures for the risks involved. Vulnerability is the combined effect of hydrogeologic susceptibility to contamination based on characteristics of the aquifer and the contamination potential.

Vulnerability and hydrogeologic susceptibility to contamination can be determined by using the guidance in WAC 365-190-100(a), such as the following:

- Depth to groundwater.
- Aquifer properties such as hydraulic conductivity, gradients, and size.
- Soil, including texture, permeability, and contaminant attenuation properties.
- Characteristics of the vadose zone including permeability and attenuation properties.
- Other relevant factors.

The following have been considered to evaluate vulnerability based on the contaminant loading potential outlined in WAC 365-190-100:

- General land use.
- Waste disposal sites.

Part 2 – Technical Information



- Agriculture activities.
- Well logs and water quality test results.
- Proximity to marine shorelines.
- Other information about the potential for contamination.

The goals of Tumwater's classification strategy for critical aquifer recharge areas and wellhead protection are to maintain the quality of the groundwater effectively by preventing contamination, with particular attention to recharge areas of high susceptibility. Classification of these areas includes:

- Consideration of the degree to which the aquifer is used, now or in the future, as a potable (drinking) water source.
- Protective measures to preclude further degradation.
- Practicability of treatment measures to maintain potability.
- Availability of alternative potable water sources.
- The degree of sensitivity of contaminants entering the aquifer.

The aquifers within Tumwater provide drinking water and require an Aquifer Protection Overlay based on TMC Chapter 18.39. The intent of the Aquifer Protection Overlay is to identify, classify, and protect critical aquifer recharge areas within Tumwater and its urban growth area. This overlay imposes additional restrictions on development in order to protect public health and safety by preserving the existing and future groundwater supply for Tumwater and its urban growth area.

TMC Chapter 16.24 regulates land uses and

activities within the critical aquifer recharge areas. These areas are identified using a mapping system, Geodata, maintained by Thurston County. There are three levels of critical aquifer recharge areas, TMC Chapter 16.24 standards apply to all critical aquifer recharge area designations.

Wellhead protection areas are the surface and subsurface areas surrounding a well or well field of a public water system. These areas have higher vulnerability to contamination from potential land uses. They are classified or rated by the amount of time it would take groundwater to reach a pumping well.

Tumwater's wellhead protection areas are divided into six-month, one-year, five-year, and ten-year time of travel zones. A raindrop landing in the one-year time of travel should reach the well within one year. These areas have extra protections and regulations found in TMC Chapter 16.26. Wellhead protection areas and the Aquifer Protection Overlay are identified using Thurston County Geodata maps.

3) Critical Aquifer Protection Concerns

Concerns about ground water in Tumwater and the Thurston region, in general, include:

- Few alternative sources of drinking water exist.
- Geological conditions in the region leave aquifers unprotected and ground water extremely vulnerable to pollution.
- Septic systems, stormwater runoff, chemical spills, pesticides, and fertilizers can add contaminants to ground water.
- Though the region's ground water is



generally of good quality, it is showing increasing effects of human activities.

• Urbanization and population growth are placing increased demands on limited ground water resources.

Potential sources of ground water pollution include pesticides and fertilizers, septic systems, hazardous materials, contaminated storm water and leaking underground storage tanks.

Gas stations and other land uses that utilize hazardous chemicals are prohibited within the one year and six-month wellhead protection areas. The types of hazardous chemicals that need to be addressed are defined in TMC Chapter 16.24 Aquifer Recharge Standards and TMC Chapter 16.26 Wellhead Protection Standards and are updated based on adopted federal and state standards, whichever is more stringent.

Wellhead protection areas are the surface or subsurface areas surrounding municipal water wells or well fields through which contaminants are reasonably likely to move toward and reach such water well or well field within six months, one year, five years, and ten years.

4) Wells and Critical Aquifers in Tumwater

a) Bush Wellfield

The Bush Wellfield is located in southwest Tumwater, near George Washington Bush Middle School. The wellfield is currently Tumwater's largest producer of drinkable water.

b) Lakeland Manor Water System

The Lakeland Manor Water System is operated and maintained by Tumwater, but it is a Satellite Management Water System that is a separate system from Tumwater's main distribution system. All drinking water for Lakeland Manor comes from its own well located on 60th Avenue Southwest. Tumwater provides an emergency interconnection with Lakeland Manor, if the well were to lose power or experience another emergency, the development will receive water from Tumwater's main distribution system.

c) Palermo Wellfield

Tumwater's Palermo Wellfield is one of Tumwater's oldest and most important wellfields, and up until the early 1990s, provided close to 100 percent of Tumwater's drinking water. Palermo wells deliver high quality groundwater from six wells to customers throughout Tumwater, blending with water from Tumwater's Bush and Port Wellfields. To help keep the water clean, Tumwater has implemented aeration treatment, which is suitable for removing volatile organic compounds coming from the Southgate and Littlerock Road and Trosper Road areas.

d) Port Wellfield

Tumwater operates several wells in the general area of the Olympia Airport and Israel Road. Many of these wells and the associated water rights were acquired by Tumwater from the Port of Olympia in 1991. As with the Palermo water rights, in 2001 Tumwater negotiated with the State Department of Ecology to review the quantities associated with previously issued water rights and determine whether the amounts were correctly established. As a result, the State Department of Ecology issued a superseding permit that increased Tumwater's water right portfolio.

Part 2 – Technical Information

e) Olympia Brewery Wellfield

The Olympia Brewery Wellfield does not currently provide drinking water to the Community of Tumwater, but it is an aquifer within Tumwater that has been carbon dated to be over 3,000 years old. It is part of the Tumwater Sand unit of the Vashon recessional outwash formation.

5) Critical Aquifer Protection

Wellhead protection is a high priority. Tumwater's first wellhead protection areas were drawn in 1997 and adopted the Wellhead Protection Ordinance. There have been many changes to the wellhead protection ordinance as water protection updates were needed, and Tumwater acquired more wells.

In 2016, Tumwater completed a new Wellhead Protection Area Plan. This plan included new groundwater flow technology which more accurately drew the protection areas. The program identifies risks of contamination with potential to impact city wells. Once identified, it finds ways to reduce or eliminate those risks. The ordinance also defines what development and/or uses within the wellhead protection areas are allowed. For example, the ordinance does not allow dry cleaners to use chemical cleaning methods on site within the designated six-month and one-year wellhead protection areas.

Tumwater also has a wellhead protection program



that supports Tumwater's mission of keeping high quality drinking water. This program ensures businesses who use, store, and dispose of hazardous materials do so in the safest manner possible. City staff work with Thurston County Environmental Health to inspect and educate businesses about proper material storage and disposal.

Quarterly groundwater monitoring is conducted at wells throughout the wellhead protection areas. Tumwater's well monitoring network serves as an early warning system against issues that could impact Tumwater's drinking water. If the system detects contamination, Tumwater can take measures to lessen the impact.

To help educate and inform the public on the importance of water protection and best practices, the Water Resources & Sustainability Department maintains a website with maps, information, and a quarterly "One Water Newsletter."

The Tumwater aquifer protection classification regime measures susceptibility to pollution in terms of vulnerability. TMC Chapter 16.24 and TMC Chapter 16.26 protect areas of high vulnerability through the Aquifer Protection Overlay, which is geographically applied Citywide. In addition, these chapters maintain specific standards applied citywide.

Part 2 – Technical Information

D. Frequently Flooded Areas

1) Introduction

Protection of life and property during floods is a vital part of Tumwater's responsibility to public safety. Many of Tumwater's rivers, streams, and lakes are subject to flooding during periods of heavy rainfall.

Tumwater has had extensive research and study completed regarding frequently flooded areas within Tumwater. Since August of 1980, Tumwater has participated in the National Flood Insurance Program, as authorized by the National Flood Insurance Act of 1968, and updated its Floodplain Ordinance in 2024.

2) Frequently Flooded Areas Classification

The Growth Management Act requires Tumwater to classify frequently flooded areas known as the 100-year floodplain based on the one percent flood designations of the Federal Emergency Management Agency and the National Flood Insurance Program. In addition, the Act requires Tumwater to provide guidance for corrective actions to mitigate or cleanse those discharges that pollute the waters of the state.

Tumwater considered the following when designating and classifying frequently flooded areas:

- 1. Effects of flooding on human health and safety, and to public facilities and services.
- Available documentation, including federal, state, and local laws, regulations, and programs, local studies and maps, and federal flood insurance programs.

- Future flow floodplain, defined as the channel of the stream and that portion of the adjoining floodplain that is necessary to contain and discharge the base flood flow at buildout without measurable increase in flood heights.
- 4. The potential effects of tsunami, high tides with strong winds, sea level rise resulting from global climate change.
- 5. Greater surface runoff caused by increasing impervious surfaces.

The frequently flooded areas within Tumwater are mainly along the Deschutes River and valley. Other areas include Trosper Lake, Barnes Lake, Munn Lake, and Black Lake. City staff utilize Thurston County Geodata mapping, which was updated May 8, 2024, to match the 2024 Flood Insurance Study and Flood Insurance Maps for Thurston County, completed by Federal Emergency Management Agency and adopted by Tumwater to locate frequently flooded areas.

3) Frequently Flooded Areas Concerns

Concerns about frequently flooded areas in Tumwater include:

- Heavy rains project to increase from climate change can cause sudden river and stream rises and out-of-bank flows.
- Out-of-bank flows can cause damage to life, dwellings, and industrial, commercial, agricultural, recreational facilities, and Tumwater owned utilities such as drinking wells.
- 3. Groundwater flooding of low-lying areas



Part 2 – Technical Information



when there is higher than normal precipitation.

4) Frequently Flooded Areas Protection

Tumwater last updated its Floodplain Overlay Ordinance in 2024. Ordinance No. 02023-017 amended floodplain regulations to be more consistent with the state model ordinance for Floodplain Management under the Nation Flood Insurance Program and the Endangered Species Act.

Continued enforcement of the floodplain management ordinance allows Federal Emergency Management Agency to make federally backed flood insurance available to property owners within Tumwater. Properties mapped in the Floodplain Overlay are subject to the regulations in TMC Chapter 18.38 to protect life and property.

In 2023 Tumwater adopted the fourth edition of the Hazards Mitigation Plan. This plan includes the Tumwater Appendix and outlines in detail the extent, impacts, risks, and mitigation measures to protect against damage from floods.

5) Flood Insurance Maps

Flood Insurance Rate Maps from the Federal Emergency Management Agency clearly delineate frequently flooded areas. These maps are used to designate the Floodplain Overlay. The Flood Insurance Rate Maps and Flood Insurance Study were updated in 2024.

The Floodplain Overlay identifies and defines the special flood hazard area within Tumwater, which is the area subject to flooding by the base flood and subject to the provisions of TMC Chapter 18.38. The Floodplain Overlay has served Tumwater well in minimizing the undesirable impacts of flooding.

Frequently flooded areas are identified by the Federal Emergency Management Agency in a scientific and engineering report entitled, "Flood Insurance Study for Thurston County, Washington and Incorporated Areas," dated May 8, 2024, and any revisions thereto, with an accompanying Flood Insurance Rate Map for Thurston County, Washington and Incorporated Areas, dated May 8, 2024, and any subsequent revisions.

The methodology and detail of these studies is accepted as best available science. TMC Chapters 16.28 and 18.38 serve to designate frequently flooded areas. If allowed, any structures permitted in the designated flood areas are subject to strict development regulations. The existing regulations were put in place after careful study, and they fulfill the requirements of the Growth Management Act regarding designation, classification, and protection of frequently flooded areas.

6) Salmon Creek Groundwater Flooding

Above average rainfall caused localized flooding in Salmon Creek Basin in the rainy seasons of 1996-97 and 1998-99. Property owners experienced a range of inconveniences from high water around and under homes to failed septic systems, contaminated drinking water, and restricted access to property. Salmon Creek Basin is located in Thurston County, Washington, in the southern portion of Tumwater.

The basin is relatively flat and slopes gently toward Salmon Creek, which flows into Black River. The basin boundary encompasses

Part 2 – Technical Information

approximately 12 square miles (7,500 acres) from the Tumwater on the northern boundary, to 113th Avenue on its southernmost edge. The western boundary lies along Littlerock Road, and the eastern boundary extends just past Brooks Lane. The area of the basin is defined by the surface and groundwater sources that contribute to recharge of Salmon Creek.

A comprehensive study of the area was

E. Geologically Hazardous Areas

1) Introduction

The Conservation Element defines geologically hazardous areas as those areas susceptible to erosion, landslides, earthquakes, and other geological events, which pose a threat to public safety. This section discusses the proper design and location of development to remove or reduce incompatibility with underlying geology. Appropriate engineering, design, or construction can be used to achieve this goal of land use and geological harmony.

It must also be recognized that even the best of efforts in proper design and application of technology, at times, will not adequately reduce the risks of geological damage. In these instances, building in such extreme geologically hazardous areas should be avoided.

2) Geologically Hazardous Areas Classification

Areas in Tumwater that are prone to one or more of the following hazards are defined as geologically hazardous:

- 1. Erosion.
- 2. Landslides.
- 30 | Page

completed in late 1999. As a result, Tumwater and several other jurisdictions in Thurston County completed and adopted the Salmon Creek Comprehensive Drainage Basin Plan.

The development review process within the Salmon Creek Comprehensive Drainage Basin Plan was adopted by resolution but Tumwater and was incorporated into the Drainage Design and Erosion Control Manual for Tumwater, WA.

- 3. Earthquakes.
- 4. Volcanic hazards (slight risk in Tumwater).
- 5. Tsunami Hazard (slight risk in Tumwater).
- 6. Stream channel migration zones.
- Other geologic events, including mass wasting, debris flows, rock falls, and differential settlement.

The Conservation Element identifies areas with the above-described hazards and subsequently classifies areas within Tumwater in one of three categories:

- 1. Known or suspected risk.
- 2. No risk.
- 3. Risk unknown (because of lack of information).

Geologically Hazardous Areas Identification

The identification methodology used in the Conservation Element to define geologically hazardous areas is as follows:



Part 2 – Technical Information



a) Erosion

Identified by the U.S. Department of Agriculture Soil Conservation Service as the breakdown, detachment, transport, and redistribution of soil particles by forces of water, wind, or gravity. Erosion hazard areas include areas likely to become unstable, such as bluffs, steep slopes, and areas with unconsolidated soils.

b) Landslides

A landslide generally refers to the downhill movement of rock, soil, or debris. The term landslide can also refer to the deposit that is created by a landslide event. Landslide areas are at risk of mass movement due to a combination of geologic, topographic, and hydrologic factors. They include any areas susceptible to landslide because of any combination of bedrock, soil, slope (gradient), slope aspect, structure, hydrology, or other factors found in WAC 365-190-120(6). Landslide hazard areas are further defined in TMC Chapter 16.20.

c) Earthquakes

Earthquakes or seismic hazard areas are those which are subject to severe risk of damage because of ground shaking, slope failure, settlement, soil liquefaction, or surface faulting, debris flows, lahars, or tsunamis. Within the region, the historic damage inducement has been ground shaking which results in settlement and soil liquefaction. The amount of ground shaking is affected by earthquake magnitude, distance from the earthquake epicenter, cohesionless soils of low density, shallow ground water tables, and sub-surface geologic structure.

4) Geologically Hazardous Areas in Tumwater

a) Erosion

The two major soil groupings within Tumwater are the Alderwood-Everett and Spanaway-Nisqually series. These soil types are identified as having severe erosion hazard characteristics when disturbed. Erosion risk is dependent on-site conditions and locations outlined in TMC Chapter 16.20.

b) Landslides

Areas of slope over 15 percent and groundwater seepage exist on Tumwater Hill, the Deschutes River valley slopes, and Bush Mountain. Steep slope risk is the combination of slope, soil, and other factors. A geotechnical report is required in certain areas based on conditions outlined in TMC Chapter 16.28, drainage, and site development reviews, and building plan review.

c) Earthquakes

Tumwater is identified in the International Building Code as being located within the Zone D seismic zone map of the United States. Zone A is the lowest and Zone E is the highest. Zone D is a high-risk area for earthquakes and IBC standards for building construction set out stringent structural performance standards.

Liquefaction is primarily addressed through implementation of International Building Code requirements, local amendments, and structural engineering principles. Liquefaction issues associated with the poor soil bearing capacity in Tumwater are addressed with commonly accepted engineering principles such as "pin piles" or "aggregate piers" to ensure structures of up to approximate six stories can accommodate the impacts of a seismic event.

31 | Page

Part 2 – Technical Information



The 2024 update to the Hazard Mitigation Plan for Thurston County identifies areas of liquefaction based on the fault lines in the region. The entire Deschutes Valley from Henderson Boulevard SE to the former Olympia Brewery has high liquefaction susceptibility. Percival Creek vicinity from Trosper Road SW to Sapp Road SW have areas of moderate to high liquefaction.

d) Volcanic Hazards

According to the 2014 U.S. Geological Survey Mt.

F. Fish and Wildlife Habitat Conservation Areas

1) Introduction

Preservation of fish and wildlife habitats is critical to the protection of suitable environments for animal species and in providing a natural beauty and healthy quality of life for Tumwater and its residents.

The conservation of habitat means active land management for maintaining species within their preferred habitats and accustomed geographic distribution. These areas may include, but are not limited to, rare or vulnerable ecological systems, communities, and habitat or habitat elements including seasonal ranges, breeding habitat, winter range, and movement corridors; and areas with high relative population density or species richness. If these areas are altered, it may reduce the likelihood of the species survival.

2) Habitat Conservation Classification

The Growth Management Act requires Tumwater to classify seasonal ranges and habitats that are critical to the survival of endangered, threatened, and sensitive species. Within Tumwater, habitats and species are identified which are of local Rainier Lahar Hazard Map, Tumwater is not in a Volcanic Hazard zone. Only ash fall from a volcanic eruption is expected to affect Tumwater.

5) Development within Geologically Hazardous Areas

TMC Chapter 16.20 sets forth standards for construction in areas identified as susceptible to earthquake and landslide conditions.

importance. Tumwater must include best available science when classifying these areas according to WAC 365-195.

A listing of the types of fish and wildlife habitat areas to be protected by state-mandate in Tumwater include:

- Areas where endangered, threatened, and sensitive species have a primary association.
- Habitats and species of local importance, as determined locally.
- Naturally occurring ponds under twenty acres and their submerged aquatic beds, which provide fish or wildlife habitat.
- Waters of the state as defined in WAC Title 222.
- Lakes, ponds, streams, and rivers planted with game fish by a governmental or tribal entity.
- State natural area preserves and natural resource conservation areas.
- All areas within Tumwater meeting one or more of the criteria in this section,

Part 2 – Technical Information

regardless of any formal identification, are subject to the provisions of TMC Chapter 16.32 and should be managed consistent with the best available science, which includes the "Washington State Department of Fish and Wildlife's Management Recommendations for Priority Habitat and Species" as written or hereafter amended.

3) Habitat Protection Techniques

After classifying and designating fish and wildlife areas in Tumwater, the following protection techniques are pursued:

- Creating a system of fish and wildlife habitat with connections between larger habitat blocks and open spaces, integrating with open space corridor planning where appropriate.
- Limiting the level of human activity in such areas including presence of roads and level of recreation type after site specific analysis and planning passive or active recreation may be appropriate for certain areas and habitats.
- Protecting riparian ecosystems and salmonid habitat.
- Evaluating land uses surrounding ponds and fish and wildlife habitat areas that may negatively or positively impact these areas and functions.
- Establishing buffer areas around these areas to separate incompatible uses from habitat areas.
- Restoration of lost and impaired salmonid habitat.

4) Habitat Identification

A review of state and local records and studies on habitats and species indicates that the following habitat categories exist within Tumwater:

- Seasonal ranges and habitats with which federal and state listed endangered, threatened, and sensitive species have a primary association and which, if altered, may reduce the likelihood that the species will maintain and reproduce over the long term.
- 2. Naturally occurring ponds under twenty acres and their submerged aquatic beds which provide fish or wildlife habitat.

Naturally occurring ponds do not include ponds deliberately designed and created from dry sites, such as canals, detention facilities, wastewater treatment facilities, farm ponds, temporary construction ponds of less than three years duration and landscape amenities. However, naturally occurring ponds may include those artificial ponds intentionally created from dry areas to mitigate conversion of ponds, if permitted by a regulatory authority.

3. Waters of the state. Waters of the state are defined in WAC Title 222; the forest practices rules and regulations. Tumwater uses the water typing system established in WAC 222-16-030 to classify waters of the state.

The following factors are considered when classifying waters of the state as fish and wildlife habitats:

a. Species present in Tumwater that are

176



endangered, threatened, or sensitive, and other species of concern.

- b. Species present which are sensitive to habitat manipulation.
- c. Historic presence of priority species.
- d. Existing surrounding land uses that are incompatible with salmonid habitat.
- e. Presence and size of riparian ecosystems.
- f. Existing water rights.
- g. The intermittent nature of some of the higher classes of waters of the state.
- 4. Lakes, ponds, streams, and rivers planted with game fish. This includes game fish planted in these water bodies under the auspices of a federal, state, local, or tribal program or which supports priority fish species as identified by the State Department of Fish and Wildlife.

5) Sensitive Species Identification

The State Department of Fish and Wildlife maintains a listing of the priority habitats and species for Tumwater. This database is the reference document to be used by Tumwater in the protection of habitats and species identified within Tumwater.

6) Fish and Wildlife Habitat Protection

Given Tumwater's diversity of fish and wildlife habitats in terms of geographic location, biological sensitivity, species hierarchy, and current and future adjacent land uses, the Conservation Element proposes a regulation and protection process based upon performance standards to be applied to site-specific development.

These performance standards are to be implemented on site-specific projects through TMC Chapter 16.32 and associated development permits. If there are any conflicts between the Shoreline Master Program and the standards in TMC Chapter 16.32, the requirements of the Shoreline Master Program apply.

In addition, Tumwater's Flood Ordinance incorporates federal recommendations for protection of aquatic species. Tumwater has also upgraded the fish capture facility at the head of Tumwater Falls and is planning a new hatchery with the State Department of Fish and Wildlife to help stabilize South Sound salmon populations.

a) Threatened and Endangered Species

Tumwater has critical habitat for several federally designated, threatened, or endangered species including:

- Bull Trout (threatened).
- Oregon Spotted Frog (threatened).
- Mazama Pocket Gopher Olympia Subspecies (threatened).
- Puget Sound Chinook Salmon (threatened).
- Streaked Horned Lark (threatened).
- Taylor's Checkerspot Butterfly (endangered).
- Water howellia (threatened).

Tumwater has critical habitat for several state designated, threatened, or endangered species including:

Part 2 – Technical Information



- Oregon Spotted Frog (endangered).
- Oregon vesper sparrow (threatened).
- Streaked Horned Lark (endangered).
- Western gray squirrel (threatened).

b) Habitat Conservation Plan

The southern portion of Tumwater is located in a glacial prairie that has been historically called Bush Prairie. Most of Bush Prairie has been converted to agriculture or forestry, residences, and businesses, but part of it still remains and provides a home for the unique flora and fauna of

the South Puget Sound Prairie ecosystem.

A Habitat Conservation Plan will provide for the long-term preservation and management of three species, protected or soon to be protected under the federal Endangered Species Act that occurs in Tumwater: Olympia pocket gopher subspecies of the Mazama pocket gopher, the streaked horned lark, the Oregon spotted frog, and the Oregon vesper sparrow.

As of 2025, Tumwater in cooperation with the Port of Olympia is preparing the Bush Prairie Habitat Conservation Plan for prairie species. When that plan is completed, the Comprehensive Plan and the Tumwater Municipal Code will need to be amended to reflect the Plan.

Part 2 – Technical Information



Appendix A Foundational Documents and Best Available Science

The table below identifies the foundational planning documents that form the basis for the

Conservation Element of the Comprehensive Plan.

1. General Policy

Table C-6. General Policy Foundational Documents for the Conservation Element.

Topic Index	Supporting Plans and Materials		
	Land Use Element (2025)		
General Policy	 County-Wide Planning Policies, Thurston County (2015) 		
	 Sustainable Thurston, Thurston Regional Planning Council (2013) 		

Part 2 – Technical Information



2. Critical Areas

Tahle C-7	Critical Areas	Foundational	Documents	for the	Conservation	Flement
Tuble C 7.	cifficul Arcus	roundational	Documents	joi uic	conscivation	Licincii.

Topic Index	Supporting Plans and Materials			
	 Best Available Science for Freshwater Wetlands, State Department of Fish and Wildlife and State Department of Ecology (2005, 2013) 			
Wetland Areas	 Custom Soil Resource Report for Thurston County Area, Washington – 2016 Tumwater Soil Survey, U.S. Department of Agriculture (2016) 			
	 Priority Habitats and Species Data Base, State Department of Fish and Wildlife (Updated annually) 			
	 Priority Habitats and Species List, State Department of Fish and Wildlife (1999) 			
	Shoreline Master Program (2014)			
	 Soil Conservation Service Soil Survey of Thurston County, U.S. Department of Agriculture (1958) 			
	 Wetland Mapping for the Thurston Region, Thurston Regional Planning Council (2004) 			
Critical Aquifer Recharge Areas	Lands for Public Purposes Element			
	 Thurston County On-Site Sewage Management Plan (2014) 			
	 Wastewater Resource Management Plan, LOTT Clean Water Alliance (2015) 			
	Water System Plan (2010-2015)			
	 Wellhead Protection Plan (2010 informal update, 2016 update underway) 			
Part 2 – Technical Information



Topic Index	Supporting Plans and Materials		
	• Comprehensive Stormwater Implementation Plan (2002, 2016 Plan in development)		
	Flood Hazard Maps (2024)		
	 Flood Insurance Studies and the Flood Insurance Rate Maps, Federal Emergency Management Agency (2012 – 2016) 		
	Floodplain Overlay Ordinance (2016)		
Frequently Flooded Areas	GIS Thurston County Floodplain Mapping		
	 Littlerock-70th Avenue Annexation Area Drainage Study (Part of the Littlerock-70th Avenue Annexation in 2008) (2011) 		
	 Natural Hazards Mitigation Plan for the Thurston Region (2009) 		
	 Salmon Creek Comprehensive Drainage Basin Plan (2004) 		
	 Coastal Zone Atlas, State Department of Ecology (2014) 		
	 Custom Soil Resource Report for Thurston County Area, Washington – 2016 Tumwater Soil Survey, U.S. Department of Agriculture (2016) 		
Geologically Hazardous Areas	 Geologic Map of the Centralia Quadrangle, Washington, State Department of Natural Resources (1987) 		
	 Liquefaction Hazards Map, State Department of Natural Resources 		
	 Natural Hazards Mitigation Plan for the Thurston Region (2009) 		
	 Steep Slopes Map, State Department of Natural Resources 		

38 | Page

Part 2 – Technical Information



	 Habitat Conservation Plan (In development 2016-25)
	 Landscape Planning for Washington's Wildlife: Managing for Biodiversity in Developing Areas, State Department of Fish and Wildlife (2009)
	 Management Recommendations for Priority Habitat and Species (Multiple Documents), State Department of Fish and Wildlife (1991 – 2011)
	 Priority Habitats and Species Data Base, State Department of Fish and Wildlife (Updated annually)
	 Priority Habitats and Species List, State Department of Fish and Wildlife (1999)
	• Determination of Threatened Status for Bull Trout in the Coterminous United States, Federal Register (64):58910-58933, U.S. Fish and Wildlife Service (1999)
Fish and Wildlife Conservation Areas	 Endangered and Threatened Species: Regulations Consolidation, Final Rule, Code of Federal Regulations Volume 50 Part 223.102, National Oceanic and Atmospheric Administration (1999) (Chinook Salmon)
	 Endangered and Threatened Wildlife and Plants, Threatened Status for Oregon Spotted Frog, Final Rule, Federal Register Volume 79:51658, U.S. Fish and Wildlife Service (2014)
	• Threatened Species Status for the Olympia Pocket Gopher, Roy Prairie Pocket Gopher, Tenino Pocket Gopher, and Yelm Pocket Gopher, With Special Rule, Federal Register Volume 79:19759, U.S. Fish and Wildlife Service (2014)
	• Endangered and Threatened Wildlife and Plants, Endangered Status for Taylor's Checkerspot Butterfly and Threatened Status for the Streaked Horned Lark, Final Rule, Federal Register Volume 78:61452, U.S. Fish

39 | Page

Part 2 – Technical Information



Topic Index	Supporting Plans and Materials	
	and Wildlife Service (2013)	
	Multiple Additional ESA Documents described in the Environmental Conservation Online System	

3. Resource Lands

Table C-8.	Resource Lands Foundational	Documents for the Conservation	Element.

Topic Index	Supporting Plans and Materials			
Agricultural Lands	 Custom Soil Resource Report for Thurston County Area, Washington – 2025 Tumwater Soil Survey, U.S. Department of Agriculture (2025) Handbook No. 210, U.S. Department of Agriculture (1961) Natural Resource Conservation Service Soil Survey of Thurston County, U.S. Department of Agriculture (1958) 			
Forest Lands	 Private Forest Land Grades (WAC 458-40-530), State Department of Revenue 			
Mineral Resource Lands	 Mineral Resource Lands (WAC 365-190-070), State Department of Natural Resources Thurston County Assessor Land Use Data 			

Part 2 – Technical Information



Appendix B Open Space Taxation Act Summary

The Open Space Taxation Act Summary is

attached as a separate document.

Part 2 – Technical Information



Appendix C Tumwater Soils Report

The Custom Soils Resource Report for Tumwater is attached as a separate document.

Attachment E

Appendix B

Open Space Taxation Act

The information and instructions in this publication are to be used when applying for assessment on the basis of current use under the "open space laws," chapter 84.34 RCW and chapter 458-30 WAC.

What is the Open Space Taxation Act?

The Open Space Taxation Act, enacted in 1970, allows property owners to have their open space, farm and agricultural, and timber lands valued at their current use rather than at their highest and best use. The Act states that it is in the best interest of the state to maintain, preserve, conserve, and otherwise continue in existence adequate open space lands for the production of food, fiber, and forest crops and to assure the use and enjoyment of natural resources and scenic beauty for the economic and social wellbeing of the state and its citizens.

Lands qualifying for current use classification

The law provides three classifications:

Open space land Farm and agricultural land Timber land



Open space land is defined as any of the following:

- Any land area zoned for open space by a comprehensive official land use plan adopted by any city or county.
- 2. Any land area in which the preservation in its present use would:
 - a. Conserve and enhance natural or scenic resources.
 - b. Protect streams or water supply.
 - c. Promote conservation of soils, wetlands, beaches or tidal marshes. (As a condition of granting open space classification, the legislative body may not require public access on land classified for the purpose of promoting conservation of wetlands.)
 - d. Enhance the value to the public of neighbouring parks, forests, wildlife preserves, nature reservations or sanctuaries or other open space.
 - e. Enhance recreation opportunities.
 - f. Preserve historic sites.

- **g.** Preserve visual quality along highway, road, and street corridors or scenic vistas.
- h. Retain in its natural state tracts of land not less than one acre situated in an urban area and open to public use on such conditions as may be reasonably required by the legislative authority granting the open space classification.
- 3. Any land meeting the definition of "farm and agricultural conservation land," which means either:
 - a. Land previously classified under the farm and agricultural classification that no longer meets the criteria and is reclassified under open space land; or
 - b. "Traditional farmland," not classified, that has not been irrevocably devoted to a use inconsistent with agricultural uses, and that has a high potential for returning to commercial agriculture.



Farm and agricultural land is defined as any of the following:

- 1. Any parcel of land that is 20 or more acres, or multiple parcels of land that are contiguous and total 20 or more acres, and are:
 - a. Devoted primarily to the production of livestock or agricultural commodities for commercial purposes.
 - b. Enrolled in the federal conservation reserve program (CRP) or its successor administered by the United States Department of Agriculture.
 - c. Other commercial agricultural activities established under chapter 458-30 WAC.
- 2. Any parcel of land that is five acres or more but less than 20 acres, is devoted primarily to agricultural uses, and has produced a gross income equivalent to:
 - a. Prior to January 1, 1993, \$100 or more per acre per year for three of the five calendar years preceding the date of application for classification.
 - b. On or after January 1, 1993, \$200 or more per acre per year for three of the five calendar years preceding the date of application for classification.

- 3. Any parcel of land that is five acres or more but less than 20 acres, is devoted primarily to agricultural uses, and has standing crops with an expectation of harvest within:
 - a. Seven years and a demonstrable investment in the production of those crops equivalent to \$100 or more per acre in the current or previous calendar year.
 - b. Fifteen years for short rotation hardwoods and a demonstrable investment in the production of those crops equivalent to \$100 or more per acre in the current or previous calendar year.
- 4. For parcels of land five acres or more but less than 20 acres, "gross income from agricultural uses" includes, but is not limited to, the wholesale value of agricultural products donated to nonprofit food banks or feeding programs.
- 5. Any parcel of land less than five acres devoted primarily to agricultural uses and has produced a gross income of:
 - a. Prior to January 1, 1993, \$1,000 or more per year for three of the five calendar years preceding the date of application for classification.
 - b. On or after January 1, 1993, \$1,500 or more per year for three of the five calendar years preceding the date of application for classification.
- **6.** "Farm and agricultural land" also includes any of the following:
 - a. Incidental uses compatible with agricultural purposes, including wetland preservation, provided such use does not exceed 20 of the classified land.

- b. Land on which appurtenances necessary for production, preparation, or sale of agricultural products exist in conjunction with the lands producing such products.
- c. Any non-contiguous parcel one to five acres, that is an integral part of the farming operations.
- d. Land on which housing for employees or the principal place of residence of the farm operator or owner is sited provided the use of the housing or residence is integral to the use of the classified land for agricultural purposes, the housing or residence is on or contiguous to the classified land, and the classified land is 20 or more acres.
- e. Land that is used primarily for equestrian-related activities for which a charge is made, including, but not limited to, stabling, training, riding, clinics, schooling, shows, or grazing for feed. Depending on the number of classified acres, the land may be subject to minimum gross income requirements.
- f. Land that is primarily used for commercial horticultural purposes, including growing seedlings, trees, shrubs, vines, fruits, vegetables, flowers, herbs, and other plants in containers, whether under a structure or not. For additional criteria regarding this use, please refer to RCW 84.34.020(2)(h).

Timber land is defined as the following:

Any parcel of land five or more acresor multiple parcels of land that are contiguous and total five or more acres which is or are devoted primarily to the growth and harvest of timber for commercial purposes. Timber land means the land only and does not include a residential homesite. The term includes land used for incidental uses that are compatible with the growing and harvesting of timber but no more than 10% of the land may be used for such incidental uses.

It also includes the land which appurtenances necessary for the production, preparation, or sale of the timber products exist in conjunction with land producing these products.

The timber land classification may be unavailable in some counties. As a result of the passage of Senate Bill 6180 in 2014, counties have the option to merge their timber land classification into their designated forest land program under chapter 84.33 RCW. To determine whether your county offers the timber land classification, you may contact the county assessor or visit the Department of Revenue's website at: www.dor.wa.gov.

Who may apply?

An owner or contract vendee may apply for current use assessment. However, all owners or contract vendees must sign the application for classification, and any resulting agreement.

When may I apply?

Applications may be made for classification at any time during the year from January 1 through December 31. If approved, current use assessment will begin on January 1 following the year the application was submitted.

Where do I get the application?

Application forms for the farm and agricultural land classification are available from the county assessor's office. Application forms for the open space and timber land classifications are available from either the county assessor's office or by contacting the county legislative authority.

Where do I file the application?

An application for open space classification is filed with the county legislative authority.

An application for farm and agricultural land classification is filed with the county assessor.

An application for timber land classification is filed with the county legislative authority. Timber land applications require that a timber management plan also be filed.



Is there an application fee?

The city or county legislative authority may, at their discretion, establish a processing fee to accompany each application. This fee must be in an amount that reasonably covers the processing costs of the application.

What happens after I file my application for open space classification?

Applications for classification or reclassification as "open space land" are made to the appropriate agency or official called the "granting authority." If the land is located in the county's unincorporated area, the county legislative authority is the granting authority on the application. If the land is located within an incorporated area of the county, the application is acted upon by both the county and city legislative authorities.

If the application is subject to a comprehensive plan that has been adopted by any city or county it will be processed in the same manner in which an amendment to the comprehensive plan is processed. If the application is not subject to a comprehensive land use plan, a public hearing on the application will be conducted, but a notice announcing the hearing must be published at least 10 days prior to the hearing.

The granting authority must approve or reject the application within six months of receiving the application. In determining whether an application made for classification or reclassification should be approved or denied, the granting authority may consider the benefits to the general welfare of preserving the current use of the property. Item 6.

They may require that certain conditions be met including but not limited to the granting of easements.

If the application is approved, the granting authority will, within five calendar days of the approval date, send an agreement to the applicant for signature showing the land classification and conditions imposed. The applicant may accept or reject the agreement. If the applicant accepts, he or she must sign and return the agreement to the granting authority within 30 days after receipt.

The approval or denial of the application for classification or reclassification is a legislative determination and is reviewable only for arbitrary and capricious actions. Appeal can be made only to the superior court of the county where the application was filed.

Within 10 days of receiving notice of classification of the land from the granting authority, the assessor submits the notice to the county auditor for recording in the place and manner provided for the public recording of state tax liens on real property.

If approved, current use assessment will begin on January 1 following the year the application was submitted. The criteria for classification continue to apply after classification has been granted.



How does a public benefit rating system work?

If the county legislative authority has established a public benefit rating system (PBRS) for the open space classification, the criteria contained within the rating system governs the eligibility and valuation of the land subject to the application.

When a county creates or amends a PBRS, all classified open space land will be rated under the new PBRS. Land that no longer qualifies for classification will not be removed from classification, but will be rated according to the PBRS. Within 30 days of receiving notification of the new assessed value established by the PBRS, the owner may request removal of classification of the land without imposition of additional tax, interest, and penalty.

What happens after I file my application for farm and agricultural land classification?

Upon application for classification or reclassification, the assessor may require applicants to provide data regarding the use of the land, including, but not limited to, the productivity of typical crops, sales receipts, federal income tax returns, other related income and expense data, and any other information relevant to the application.

The application will be considered approved unless the assessor notifies the applicant in writing prior to May 1 of the year after the application was submitted. The criteria for classification continue to apply after classification has been granted.

What is an "advisory committee"?

The county legislative authority must appoint a five member committee representing the active farming community within the county. This committee will serve in an advisory capacity to the assessor in implementing assessment guidelines as established by the Department of Revenue for the assessment of open space lands, farm and agricultural lands, and timber lands.

How do I appeal a denial of my farm and agricultural land application?

The owner may appeal the assessor's denial to the board of equalization in the county where the land is located. The appeal must be filed with the board on or before July 1 of the year of the determination or within 30 days after the mailing of the notice of denial, or within a time limit of up to 60 days adopted by the county legislative authority, whichever is later.

What happens after I file my application for timber land classification?

Applications for timber land classification or reclassification are made to the county legislative authority. A timber management plan is required at the time of application or when a sale or transfer of timber land occurs and a notice of continuance is signed.

The application form requests information about forest management, restocking, fire protection, insect and disease control, weed control, and any other summary of experience and activity that supports the growth and harvest of timber for commercial purposes. The application is acted upon in a manner similar to open space land applications and within six months of receiving the application.

Approval or denial of a timber land application is a legislative determination and is reviewable only for arbitrary and capricious action. Appeal can be made only to the superior court of the county where the application was filed.

Within 10 days of receiving notice of classification of the land from the granting authority, the assessor submits the notice to the county auditor for recording in the place and manner provided for the public recording of state tax liens on real property.

If approved, current use assessment will begin on January 1 following the year the application was submitted. The criteria for classification continue to apply after classification has been granted.

How is the value of classified land determined?

The assessor is required to maintain two values for each parcel that is classified. The first is the value that would be placed on the land if it was not classified. This is commonly referred to as the "fair market value." The second is the current use land value based on its current use, not highest and best use , as classified by the granting authority.

Open space land located within a county that has adopted a public benefit rating system will be valued according to the criteria of the rating system.

In the absence of a rating system, the per acre value can be no less than the lowest per acre value of classified farm and agricultural land in the county. In determining the current use value of farm and agricultural land, the assessor considers the earning or productive capacity of comparable lands from crops grown most typically in the area averaged over not less than five years. This earning or productive capacity is the "net cash rental" and is capitalized by a "rate of interest" charged on long term loans secured by a mortgage on farm or agricultural land plus a component for property taxes.

Timber land is valued according to a schedule prepared by the Department of Revenue according to chapter 84.33 RCW. The Department of Revenue annually adjusts and certifies timber land values to be used by county assessors in preparing assessment rolls. The assessors assign the timber land values to the property based upon land grades and operability classes.

When are taxes due on classified lands?

Land classified as open space, farm and agricultural, or timber land is assessed at its current use value and placed on the assessment rolls the year after the application was submitted. Taxes on classified land are due and payable the year after the current use value was placed on the assessment rolls.

How long does the classification last?

The land continues in its classification until a request for removal is made by the owner, the use of land no longer complies, a sale or transfer to an owner that causes land to be exempt from property taxes, or the ownership has changed and the new owner has not signed a Notice of Continuance. The notice of removal is recorded with the county auditor in the same manner as the recording of state tax liens on real property. Additional tax, interest, and penalties will apply if the land is removed and the removal does not meet one of the exceptions listed in RCW 84.34.108(6).

How do I withdraw from classification?

If intending to withdraw all or a portion of the land from classification after 10 years of classification, the owner must complete a withdrawal form with the county assessor.

If a portion of the land is removed from classification, the remaining portion must meet the requirements of original classification unless the remaining land has different income or investment criteria.



What happens after I file a request to withdraw?

Upon receipt of a request for withdrawal, the assessor notifies the granting authority that originally approved the classification, and, the assessor withdraws the land from classification. The land withdrawn from classification is subject to seven years of additional tax and interest, but not a 20% penalty.

What happens if the classified land is sold or transferred?

When classified land is sold or transferred, the seller or transferor becomes liable at the time of sale for the additional tax, interest, and penalty unless the new owner(s) signs the Notice of Continuance which is attached to or shown on the real estate excise tax affidavit. The county auditor cannot accept an instrument of conveyance on any classified land unless the Notice of Continuance has been signed or the additional tax, interest, and penalty has been paid. The assessor determines if the land qualifies for continued classification

What if I want to change the use of my classified property?

An owner changing the use of land from a classified use must notify the county assessor within 60 days of this action. The assessor will remove the land from classified status and impose an additional tax equal to the difference between the tax paid on the current use value and the tax that would have been paid on the land had it not been classified. The additional tax is payable for the last seven tax years, plus interest at the same rate as charged on delinquent property taxes, plus a penalty of 20% of the total amount.

If the assessor removes my land from classification, may I appeal?

Yes, the owner may file an appeal of the removal from classification to the county board of equalization on or before July 1 of the year of the determination, or within 30 days of the date the notice was mailed by the assessor, or within a time limit of up to 60 days adopted by the county legislative authority, whichever is later.



Upon removal from classification, what taxes are due?

At the time the land is removed from classification, any taxes owing from January 1 of the removal year through the removal date, and any additional tax, applicable interest, and penalty owing are due and payable to the county treasurer within 30 days of the owner being notified.

What if the additional taxes are not paid?

Any additional tax, applicable interest, and penalty become a lien on the land at the time the land is removed from classification. This lien has priority over any other encumbrance on the land. Such a lien may be foreclosed upon expiration of the same period after delinquency in the same manner as delinguent real property taxes. If unpaid, interest is charged on the total amount due at the same rate that is applied by law to delinguent property taxes. Interest accrues from the date of the delinguency until the date the total amount is paid in full.

What is done with the additional tax, interest, and penalty paid when land is removed from classification?

Upon collection, the additional tax is distributed by the county treasurer in the same manner in which current taxes applicable to the subject land are distributed. The applicable interest and penalties are distributed to the county's current expense fund.

How do I change the classification of my property?

Item 6.

Land may be reclassified, upon request by the owner, subject toall applicable qualifications for each classification, without additional tax, interest, and penalty for the following:

- Land classified as farm and agricultural land may be reclassified to timber land; timber land may be reclassified to farm and agricultural land.
- 2. Land classified as either farm and agricultural land or timber land under chapter 84.34 RCW, or forest land under chapter 84.33 RCW may be reclassified to open space land.
- 3. Land classified as farm and agricultural land or timber land may be reclassified to forest land under chapter 84.33 RCW.
- 4. Land previously classified as farm and agricultural land may be reclassified to open space land as "farm and agricultural conservation land" and subsequently be reclassified back to farm and agricultural land.

Applications for reclassification are acted upon in the same manner as approvals for initial classification. The county assessor approves all applications for farm and agricultural classifications and reclassifications. The county legislative authority (and in some cases, the city legislative authority)approves all land classifications or reclassifications for timber land and open space land, including farm and agricultural conservation land.

Is supporting information required to change classifications?

The assessor may require an owner of classified land to submit data regarding the use of the land, productivity of typical crops, income and expense data, and similar information regarding continued eligibility.

Laws and Rules

It is helpful to read the complete laws, Revised Code of Washington, chapters 84.33 and 84.34 (RCW) and rules, Washington Administrative Code, chapter 458-30 (WAC) to understand requirements of the classifications and the tax liabilities incurred.

Need More Information?

Requirements for making application for current use classification are available at the county assessor's office or by contacting the county legislative authority.

For general information contact:

- Department of Revenue, Property Tax Division
 P. O. Box 47471
 Olympia, Washington 98504-7471
 360-534-1400
- Website: dor.wa.gov
- Telephone Information Center 360-705-6705
- For tax assistance or to request this document in an alternate format, visit dor.wa.gov or call 360-705-6705. Teletype (TTY) users may use the Washington Relay Service by calling 711.



dor.wa.gov



United States Department of Agriculture

Natural Resources Conservation Service A product of the National Cooperative Soil Survey, a joint effort of the United States Department of Agriculture and other Federal agencies, State agencies including the Agricultural Experiment Stations, and local participants

Custom Soil Resource Report for Thurston County Area, Washington



Preface

Soil surveys contain information that affects land use planning in survey areas. They highlight soil limitations that affect various land uses and provide information about the properties of the soils in the survey areas. Soil surveys are designed for many different users, including farmers, ranchers, foresters, agronomists, urban planners, community officials, engineers, developers, builders, and home buyers. Also, conservationists, teachers, students, and specialists in recreation, waste disposal, and pollution control can use the surveys to help them understand, protect, or enhance the environment.

Various land use regulations of Federal, State, and local governments may impose special restrictions on land use or land treatment. Soil surveys identify soil properties that are used in making various land use or land treatment decisions. The information is intended to help the land users identify and reduce the effects of soil limitations on various land uses. The landowner or user is responsible for identifying and complying with existing laws and regulations.

Although soil survey information can be used for general farm, local, and wider area planning, onsite investigation is needed to supplement this information in some cases. Examples include soil quality assessments (http://www.nrcs.usda.gov/wps/portal/nrcs/main/soils/health/) and certain conservation and engineering applications. For more detailed information, contact your local USDA Service Center (https://offices.sc.egov.usda.gov/locator/app?agency=nrcs) or your NRCS State Soil Scientist (http://www.nrcs.usda.gov/wps/portal/nrcs/detail/soils/contactus/? cid=nrcs142p2_053951).

Great differences in soil properties can occur within short distances. Some soils are seasonally wet or subject to flooding. Some are too unstable to be used as a foundation for buildings or roads. Clayey or wet soils are poorly suited to use as septic tank absorption fields. A high water table makes a soil poorly suited to basements or underground installations.

The National Cooperative Soil Survey is a joint effort of the United States Department of Agriculture and other Federal agencies, State agencies including the Agricultural Experiment Stations, and local agencies. The Natural Resources Conservation Service (NRCS) has leadership for the Federal part of the National Cooperative Soil Survey.

Information about soils is updated periodically. Updated information is available through the NRCS Web Soil Survey, the site for official soil survey information.

The U.S. Department of Agriculture (USDA) prohibits discrimination in all its programs and activities on the basis of race, color, national origin, age, disability, and where applicable, sex, marital status, familial status, parental status, religion, sexual orientation, genetic information, political beliefs, reprisal, or because all or a part of an individual's income is derived from any public assistance program. (Not all prohibited bases apply to all programs.) Persons with disabilities who require

alternative means for communication of program information (Braille, large print, audiotape, etc.) should contact USDA's TARGET Center at (202) 720-2600 (voice and TDD). To file a complaint of discrimination, write to USDA, Director, Office of Civil Rights, 1400 Independence Avenue, S.W., Washington, D.C. 20250-9410 or call (800) 795-3272 (voice) or (202) 720-6382 (TDD). USDA is an equal opportunity provider and employer.

Contents

Preface	2
How Soil Surveys Are Made	6
Soil Map	9
Soil Map	.10
Leaend	.11
Map Unit Legend	. 12
Map Unit Descriptions	13
Thurston County Area. Washington	. 16
1—Alderwood gravelly sandy loam, 0 to 8 percent slopes	.16
2-Alderwood gravelly sandy loam, 8 to 15 percent slopes	.17
3-Alderwood gravelly sandy loam. 15 to 30 percent slopes	.19
20—Cagev loamv sand	.21
27—Delphi very gravelly loam, 3 to 15 percent slopes	.22
30—Dystric Xerochrepts. 60 to 90 percent slopes	.23
32—Everett very gravelly sandy loam, 0 to 8 percent slopes	24
33—Everett very gravelly sandy loam, 8 to 15 percent slopes	25
34—Everett very gravelly sandy loam, 15 to 30 percent slopes	26
35—Everett very gravelly sandy loam, 30 to 50 percent slopes	28
38—Giles silt loam. 0 to 3 percent slopes	29
39—Giles silt loam, 3 to 15 percent slopes	.30
40—Giles silt loam, 15 to 30 percent slopes	.31
41—Godfrey silty clay loam	.32
45—Hvdraguents, tidal	.34
46—Indianola loamy sand. 0 to 5 percent slopes	35
47—Indianola loamy sand, 5 to 15 percent slopes	36
48—Indianola loamy sand, 15 to 30 percent slopes	38
51—Kapowsin silt loam. 3 to 15 percent slopes	.39
52—Kapowsin silt loam, 15 to 30 percent slopes	.41
53—Kapowsin silt loam, 30 to 50 percent slopes	.42
65—McKenna gravelly silt loam, 0 to 5 percent slopes	.43
69—Mukilteo muck	.44
70—Mukilteo muck, drained	.45
73-Nisqually loamy fine sand, 0 to 3 percent slopes	.46
74—Nisqually loamy fine sand, 3 to 15 percent slopes	.47
75—Norma fine sandy loam	.49
76—Norma silt loam.	.50
84—Pilchuck loamy sand	. 51
85—Pits, gravel	. 52
88—Puget silt loam	.52
89—Puyallup silt loam	.54
98—Salkum silty clay loam, 8 to 15 percent slopes	. 55
102—Schneider very gravelly loam, 20 to 40 percent slopes	56
103—Schneider very gravelly loam, 40 to 65 percent slopes	.57
104—Semiahmoo muck	. 58

106—Shalcar variant muck	59
108—Skipopa silt loam, 3 to 15 percent slopes	60
109—Spana gravelly loam	61
110—Spanaway gravelly sandy loam, 0 to 3 percent slopes	.62
115—Sultan silt loam	.63
120—Tisch silt loam	.64
125—Xerorthents, 0 to 5 percent slopes	.65
126—Yelm fine sandy loam, 0 to 3 percent slopes	66
127—Yelm fine sandy loam, 3 to 15 percent slopes	67
128—Yelm fine sandy loam, 15 to 30 percent slopes	68
129—Water	69
Soil Information for All Uses	.71
Soil Reports	.71
Land Classifications	.71
Prime and other Important Farmlands	.71
References	.75

How Soil Surveys Are Made

Soil surveys are made to provide information about the soils and miscellaneous areas in a specific area. They include a description of the soils and miscellaneous areas and their location on the landscape and tables that show soil properties and limitations affecting various uses. Soil scientists observed the steepness, length, and shape of the slopes; the general pattern of drainage; the kinds of crops and native plants; and the kinds of bedrock. They observed and described many soil profiles. A soil profile is the sequence of natural layers, or horizons, in a soil. The profile extends from the surface down into the unconsolidated material in which the soil formed or from the surface down to bedrock. The unconsolidated material is devoid of roots and other living organisms and has not been changed by other biological activity.

Currently, soils are mapped according to the boundaries of major land resource areas (MLRAs). MLRAs are geographically associated land resource units that share common characteristics related to physiography, geology, climate, water resources, soils, biological resources, and land uses (USDA, 2006). Soil survey areas typically consist of parts of one or more MLRA.

The soils and miscellaneous areas in a survey area occur in an orderly pattern that is related to the geology, landforms, relief, climate, and natural vegetation of the area. Each kind of soil and miscellaneous area is associated with a particular kind of landform or with a segment of the landform. By observing the soils and miscellaneous areas in the survey area and relating their position to specific segments of the landform, a soil scientist develops a concept, or model, of how they were formed. Thus, during mapping, this model enables the soil scientist to predict with a considerable degree of accuracy the kind of soil or miscellaneous area at a specific location on the landscape.

Commonly, individual soils on the landscape merge into one another as their characteristics gradually change. To construct an accurate soil map, however, soil scientists must determine the boundaries between the soils. They can observe only a limited number of soil profiles. Nevertheless, these observations, supplemented by an understanding of the soil-vegetation-landscape relationship, are sufficient to verify predictions of the kinds of soil in an area and to determine the boundaries.

Soil scientists recorded the characteristics of the soil profiles that they studied. They noted soil color, texture, size and shape of soil aggregates, kind and amount of rock fragments, distribution of plant roots, reaction, and other features that enable them to identify soils. After describing the soils in the survey area and determining their properties, the soil scientists assigned the soils to taxonomic classes (units). Taxonomic classes are concepts. Each taxonomic class has a set of soil characteristics with precisely defined limits. The classes are used as a basis for comparison to classify soils systematically. Soil taxonomy, the system of taxonomic classification used in the United States, is based mainly on the kind and character of soil properties and the arrangement of horizons within the profile. After the soil

scientists classified and named the soils in the survey area, they compared the individual soils with similar soils in the same taxonomic class in other areas so that they could confirm data and assemble additional data based on experience and research.

The objective of soil mapping is not to delineate pure map unit components; the objective is to separate the landscape into landforms or landform segments that have similar use and management requirements. Each map unit is defined by a unique combination of soil components and/or miscellaneous areas in predictable proportions. Some components may be highly contrasting to the other components of the map unit. The presence of minor components in a map unit in no way diminishes the usefulness or accuracy of the data. The delineation of such landforms and landform segments on the map provides sufficient information for the development of resource plans. If intensive use of small areas is planned, onsite investigation is needed to define and locate the soils and miscellaneous areas.

Soil scientists make many field observations in the process of producing a soil map. The frequency of observation is dependent upon several factors, including scale of mapping, intensity of mapping, design of map units, complexity of the landscape, and experience of the soil scientist. Observations are made to test and refine the soil-landscape model and predictions and to verify the classification of the soils at specific locations. Once the soil-landscape model is refined, a significantly smaller number of measurements of individual soil properties are made and recorded. These measurements may include field measurements, such as those for color, depth to bedrock, and texture, and laboratory measurements, such as those for content of sand, silt, clay, salt, and other components. Properties of each soil typically vary from one point to another across the landscape.

Observations for map unit components are aggregated to develop ranges of characteristics for the components. The aggregated values are presented. Direct measurements do not exist for every property presented for every map unit component. Values for some properties are estimated from combinations of other properties.

While a soil survey is in progress, samples of some of the soils in the area generally are collected for laboratory analyses and for engineering tests. Soil scientists interpret the data from these analyses and tests as well as the field-observed characteristics and the soil properties to determine the expected behavior of the soils under different uses. Interpretations for all of the soils are field tested through observation of the soils in different uses and under different levels of management. Some interpretations are modified to fit local conditions, and some new interpretations are developed to meet local needs. Data are assembled from other sources, such as research information, production records, and field experience of specialists. For example, data on crop yields under defined levels of management are assembled from farm records and from field or plot experiments on the same kinds of soil.

Predictions about soil behavior are based not only on soil properties but also on such variables as climate and biological activity. Soil conditions are predictable over long periods of time, but they are not predictable from year to year. For example, soil scientists can predict with a fairly high degree of accuracy that a given soil will have a high water table within certain depths in most years, but they cannot predict that a high water table will always be at a specific level in the soil on a specific date.

After soil scientists located and identified the significant natural bodies of soil in the survey area, they drew the boundaries of these bodies on aerial photographs and

identified each as a specific map unit. Aerial photographs show trees, buildings, fields, roads, and rivers, all of which help in locating boundaries accurately.

Soil Map

The soil map section includes the soil map for the defined area of interest, a list of soil map units on the map and extent of each map unit, and cartographic symbols displayed on the map. Also presented are various metadata about data used to produce the map, and a description of each soil map unit.

Item 6.

Item 6.

Custom Soil Resource Report Soil Map



Area of Interest (AOI) Spoil Area			MAP INFORMATION	
		88	Spoil Area	The soil surveys that comprise your AOI were mapped at 1:24,000.
Soils		0	Very Stony Spot	Please rely on the bar scale on each map sheet for map
	Soil Map Unit Polygons	00 (*)	Wet Spot	measurements.
~	Soil Map Unit Lines	× ∧	Other	Source of Map: Natural Resources Conservation Service
	Soil Map Unit Points		Special Line Features	Web Soil Survey URL: Coordinate System: Web Mercator (EPSG:3857)
Special	Point Features Blowout	Water Fea	atures	
R	Borrow Pit	\sim	Streams and Canals	Maps from the Web Soil Survey are based on the Web Mercator projection which preserves direction and shape but distorts
×	Clay Spot	Transport	tation Rails	distance and area. A projection that preserves area, such as the Albers equal-area conic projection, should be used if more
\diamond	Closed Depression	~	Interstate Highways	accurate calculations of distance or area are required.
X	Gravel Pit	~	US Routes	This product is generated from the USDA-NRCS certified data as
**	Gravelly Spot	~	Major Roads	of the version date(s) listed below.
Ø	Landfill	\approx	Local Roads	Soil Survey Area: Thurston County Area, Washington
٨.	Lava Flow	Backgrou	ind	Survey Area Data: Version 18, Aug 27, 2024
 ⊛	Marsh or swamp Mine or Quarry	Mar.	Aerial Photography	Soil map units are labeled (as space allows) for map scales 1:50,000 or larger.
6	Miscellaneous Water			
õ	Perennial Water			Date(s) aerial images were photographed: Jul 18, 2020—Aug 14, 2023
\vee	Rock Outcrop			The orthophote or other base map on which the soil lines were
+	Saline Spot			compiled and digitized probably differs from the background
0 0 0 0	Sandy Spot			imagery displayed on these maps. As a result, some minor shifting of map unit boundaries may be evident.
-	Severely Eroded Spot			
\diamond	Sinkhole			
≫	Slide or Slip			
ø	Sodic Spot			

Map Unit Legend

Map Unit Symbol	Map Unit Name	Acres in AOI	Percent of AOI
1	Alderwood gravelly sandy loam, 0 to 8 percent slopes	173.8	1.2%
2	Alderwood gravelly sandy loam, 8 to 15 percent slopes	321.7	2.2%
3	Alderwood gravelly sandy loam, 15 to 30 percent slopes	36.9	0.3%
20	Cagey loamy sand	2,116.9	14.8%
27	Delphi very gravelly loam, 3 to 15 percent slopes	116.2	0.8%
30	Dystric Xerochrepts, 60 to 90 percent slopes	20.7	0.1%
32	Everett very gravelly sandy loam, 0 to 8 percent slopes	862.6	6.0%
33	Everett very gravelly sandy loam, 8 to 15 percent slopes	169.6	1.2%
34	Everett very gravelly sandy loam, 15 to 30 percent slopes	29.4	0.2%
35	Everett very gravelly sandy loam, 30 to 50 percent slopes	111.8	0.8%
38	Giles silt loam, 0 to 3 percent slopes	16.9	0.1%
39	Giles silt loam, 3 to 15 percent slopes	107.4	0.8%
40	Giles silt loam, 15 to 30 percent slopes	8.9	0.1%
41	Godfrey silty clay loam	47.4	0.3%
45	Hydraquents, tidal	1.5	0.0%
46	Indianola loamy sand, 0 to 5 percent slopes	1,432.1	10.0%
47	Indianola loamy sand, 5 to 15 percent slopes	378.0	2.6%
48	Indianola loamy sand, 15 to 30 percent slopes	471.5	3.3%
51	Kapowsin silt loam, 3 to 15 percent slopes	127.7	0.9%
52	Kapowsin silt loam, 15 to 30 percent slopes	267.5	1.9%
53	Kapowsin silt loam, 30 to 50 percent slopes	23.0	0.2%
65	McKenna gravelly silt loam, 0 to 5 percent slopes	159.7	1.1%
69	Mukilteo muck	212.3	1.5%
70	Mukilteo muck, drained	227.8	1.6%

Map Unit Symbol	Map Unit Name	Acres in AOI	Percent of AOI
73	Nisqually loamy fine sand, 0 to 3 percent slopes	3,613.9	25.2%
74	Nisqually loamy fine sand, 3 to 15 percent slopes	14.2	0.1%
75	Norma fine sandy loam	40.5	0.3%
76	Norma silt loam	707.2	4.9%
84	Pilchuck loamy sand	36.3	0.3%
85	Pits, gravel	69.4	0.5%
88	Puget silt loam	69.2	0.5%
89	Puyallup silt loam	293.6	2.1%
98	Salkum silty clay loam, 8 to 15 percent slopes	3.6	0.0%
102	Schneider very gravelly loam, 20 to 40 percent slopes	271.5	1.9%
103	Schneider very gravelly loam, 40 to 65 percent slopes	386.2	2.7%
104	Semiahmoo muck	232.2	1.6%
106	Shalcar variant muck	11.4	0.1%
108	Skipopa silt loam, 3 to 15 percent slopes	12.7	0.1%
109	Spana gravelly loam	10.1	0.1%
110	Spanaway gravelly sandy loam, 0 to 3 percent slopes	21.3	0.1%
115	Sultan silt loam	213.4	1.5%
120	Tisch silt loam	25.9	0.2%
125	Xerorthents, 0 to 5 percent slopes	43.8	0.3%
126	Yelm fine sandy loam, 0 to 3 percent slopes	423.9	3.0%
127	Yelm fine sandy loam, 3 to 15 percent slopes	36.4	0.3%
128	Yelm fine sandy loam, 15 to 30 percent slopes	33.6	0.2%
129	Water	308.0	2.2%
Totals for Area of Interest		14,319.9	100.0%

Map Unit Descriptions

The map units delineated on the detailed soil maps in a soil survey represent the soils or miscellaneous areas in the survey area. The map unit descriptions, along with the maps, can be used to determine the composition and properties of a unit.

A map unit delineation on a soil map represents an area dominated by one or more major kinds of soil or miscellaneous areas. A map unit is identified and named according to the taxonomic classification of the dominant soils. Within a taxonomic

class there are precisely defined limits for the properties of the soils. On the landscape, however, the soils are natural phenomena, and they have the characteristic variability of all natural phenomena. Thus, the range of some observed properties may extend beyond the limits defined for a taxonomic class. Areas of soils of a single taxonomic class rarely, if ever, can be mapped without including areas of other taxonomic classes. Consequently, every map unit is made up of the soils or miscellaneous areas for which it is named and some minor components that belong to taxonomic classes other than those of the major soils.

Most minor soils have properties similar to those of the dominant soil or soils in the map unit, and thus they do not affect use and management. These are called noncontrasting, or similar, components. They may or may not be mentioned in a particular map unit description. Other minor components, however, have properties and behavioral characteristics divergent enough to affect use or to require different management. These are called contrasting, or dissimilar, components. They generally are in small areas and could not be mapped separately because of the scale used. Some small areas of strongly contrasting soils or miscellaneous areas are identified by a special symbol on the maps. If included in the database for a given area, the contrasting minor components are identified in the map unit descriptions along with some characteristics of each. A few areas of minor components may not have been observed, and consequently they are not mentioned in the descriptions, especially where the pattern was so complex that it was impractical to make enough observations to identify all the soils and miscellaneous areas on the landscape.

The presence of minor components in a map unit in no way diminishes the usefulness or accuracy of the data. The objective of mapping is not to delineate pure taxonomic classes but rather to separate the landscape into landforms or landform segments that have similar use and management requirements. The delineation of such segments on the map provides sufficient information for the development of resource plans. If intensive use of small areas is planned, however, onsite investigation is needed to define and locate the soils and miscellaneous areas.

An identifying symbol precedes the map unit name in the map unit descriptions. Each description includes general facts about the unit and gives important soil properties and qualities.

Soils that have profiles that are almost alike make up a *soil series*. Except for differences in texture of the surface layer, all the soils of a series have major horizons that are similar in composition, thickness, and arrangement.

Soils of one series can differ in texture of the surface layer, slope, stoniness, salinity, degree of erosion, and other characteristics that affect their use. On the basis of such differences, a soil series is divided into *soil phases*. Most of the areas shown on the detailed soil maps are phases of soil series. The name of a soil phase commonly indicates a feature that affects use or management. For example, Alpha silt loam, 0 to 2 percent slopes, is a phase of the Alpha series.

Some map units are made up of two or more major soils or miscellaneous areas. These map units are complexes, associations, or undifferentiated groups.

A *complex* consists of two or more soils or miscellaneous areas in such an intricate pattern or in such small areas that they cannot be shown separately on the maps. The pattern and proportion of the soils or miscellaneous areas are somewhat similar in all areas. Alpha-Beta complex, 0 to 6 percent slopes, is an example.

An *association* is made up of two or more geographically associated soils or miscellaneous areas that are shown as one unit on the maps. Because of present or anticipated uses of the map units in the survey area, it was not considered practical or necessary to map the soils or miscellaneous areas separately. The pattern and relative proportion of the soils or miscellaneous areas are somewhat similar. Alpha-Beta association, 0 to 2 percent slopes, is an example.

An *undifferentiated group* is made up of two or more soils or miscellaneous areas that could be mapped individually but are mapped as one unit because similar interpretations can be made for use and management. The pattern and proportion of the soils or miscellaneous areas in a mapped area are not uniform. An area can be made up of only one of the major soils or miscellaneous areas, or it can be made up of all of them. Alpha and Beta soils, 0 to 2 percent slopes, is an example.

Some surveys include *miscellaneous areas*. Such areas have little or no soil material and support little or no vegetation. Rock outcrop is an example.

Thurston County Area, Washington

1—Alderwood gravelly sandy loam, 0 to 8 percent slopes

Map Unit Setting

National map unit symbol: 2t625 Elevation: 50 to 800 feet Mean annual precipitation: 25 to 60 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 160 to 240 days Farmland classification: Prime farmland if irrigated

Map Unit Composition

Alderwood and similar soils: 85 percent Minor components: 15 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Alderwood

Setting

Landform: Hills, ridges Landform position (two-dimensional): Summit Landform position (three-dimensional): Crest, talf Down-slope shape: Convex, linear Across-slope shape: Convex Parent material: Glacial drift and/or glacial outwash over dense glaciomarine deposits

Typical profile

A - 0 to 7 inches: gravelly sandy loam Bw1 - 7 to 21 inches: very gravelly sandy loam Bw2 - 21 to 30 inches: very gravelly sandy loam Bg - 30 to 35 inches: very gravelly sandy loam 2Cd1 - 35 to 43 inches: very gravelly sandy loam 2Cd2 - 43 to 59 inches: very gravelly sandy loam

Properties and qualities

Slope: 0 to 8 percent
Depth to restrictive feature: 20 to 39 inches to densic material
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: About 18 to 37 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very low (about 2.7 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4s Hydrologic Soil Group: B Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Limited Depth Soils (G002XS301WA), Limited Depth Soils (G002XF303WA), Limited Depth Soils (G002XN302WA) *Other vegetative classification:* Limited Depth Soils (G002XS301WA), Limited Depth Soils (G002XF303WA), Limited Depth Soils (G002XN302WA) *Hydric soil rating:* No

Minor Components

Everett

Percent of map unit: 5 percent Landform: Moraines, eskers, kames Landform position (two-dimensional): Summit, shoulder Landform position (three-dimensional): Crest, interfluve Down-slope shape: Convex Across-slope shape: Convex Hydric soil rating: No

Mckenna

Percent of map unit: 5 percent Landform: Drainageways, depressions Landform position (three-dimensional): Dip Down-slope shape: Linear, concave Across-slope shape: Concave Hydric soil rating: Yes

Shalcar

Percent of map unit: 3 percent Landform: Depressions Landform position (three-dimensional): Dip Down-slope shape: Concave Across-slope shape: Concave Hydric soil rating: Yes

Norma

Percent of map unit: 2 percent Landform: Drainageways, depressions Landform position (three-dimensional): Dip Down-slope shape: Linear, concave Across-slope shape: Concave Hydric soil rating: Yes

2—Alderwood gravelly sandy loam, 8 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2t626 Elevation: 50 to 800 feet Mean annual precipitation: 20 to 60 inches Mean annual air temperature: 46 to 52 degrees F Frost-free period: 160 to 240 days Farmland classification: Prime farmland if irrigated

Map Unit Composition

Alderwood and similar soils: 85 percent

Minor components: 15 percent

Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Alderwood

Setting

Landform: Hills, ridges Landform position (two-dimensional): Shoulder Landform position (three-dimensional): Nose slope, talf Down-slope shape: Convex, linear Across-slope shape: Convex Parent material: Glacial drift and/or glacial outwash over dense glaciomarine deposits

Typical profile

A - 0 to 7 inches: gravelly sandy loam Bw1 - 7 to 21 inches: very gravelly sandy loam Bw2 - 21 to 30 inches: very gravelly sandy loam Bg - 30 to 35 inches: very gravelly sandy loam 2Cd1 - 35 to 43 inches: very gravelly sandy loam 2Cd2 - 43 to 59 inches: very gravelly sandy loam

Properties and qualities

Slope: 8 to 15 percent
Depth to restrictive feature: 20 to 39 inches to densic material
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: About 18 to 37 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very low (about 2.7 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4s Hydrologic Soil Group: B

Ecological site: F002XA004WA - Puget Lowlands Forest

Forage suitability group: Limited Depth Soils (G002XS301WA), Limited Depth Soils (G002XF303WA), Limited Depth Soils (G002XN302WA)

Other vegetative classification: Limited Depth Soils (G002XS301WA), Limited Depth Soils (G002XF303WA), Limited Depth Soils (G002XN302WA) *Hydric soil rating:* No

Minor Components

Indianola

Percent of map unit: 5 percent Landform: Terraces, kames, eskers Landform position (three-dimensional): Tread Down-slope shape: Linear Across-slope shape: Linear Hydric soil rating: No

Everett

Percent of map unit: 5 percent Landform: Moraines, eskers, kames Landform position (two-dimensional): Shoulder, footslope Landform position (three-dimensional): Crest, base slope Down-slope shape: Convex Across-slope shape: Convex Hydric soil rating: No

Shalcar

Percent of map unit: 3 percent Landform: Depressions Landform position (three-dimensional): Dip Down-slope shape: Concave Across-slope shape: Concave Hydric soil rating: Yes

Norma

Percent of map unit: 2 percent Landform: Drainageways, depressions Landform position (three-dimensional): Dip Down-slope shape: Linear, concave Across-slope shape: Concave Hydric soil rating: Yes

3—Alderwood gravelly sandy loam, 15 to 30 percent slopes

Map Unit Setting

National map unit symbol: 2t627 Elevation: 0 to 1,000 feet Mean annual precipitation: 25 to 60 inches Mean annual air temperature: 46 to 52 degrees F Frost-free period: 160 to 240 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Alderwood and similar soils: 85 percent Minor components: 15 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Alderwood

Setting

Landform: Hills, ridges Landform position (two-dimensional): Backslope Landform position (three-dimensional): Nose slope, side slope, talf Down-slope shape: Convex, linear Across-slope shape: Convex Parent material: Glacial drift and/or glacial outwash over dense glaciomarine deposits

Typical profile

A - 0 to 7 inches: gravelly sandy loam Bw1 - 7 to 21 inches: very gravelly sandy loam *Bw2 - 21 to 30 inches:* very gravelly sandy loam *Bg - 30 to 35 inches:* very gravelly sandy loam *2Cd1 - 35 to 43 inches:* very gravelly sandy loam *2Cd2 - 43 to 59 inches:* very gravelly sandy loam

Properties and qualities

Slope: 15 to 30 percent
Depth to restrictive feature: 20 to 39 inches to densic material
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: About 18 to 37 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very low (about 2.7 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4e Hydrologic Soil Group: B Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Limited Depth Soils (G002XS301WA), Limited Depth Soils (G002XF303WA), Limited Depth Soils (G002XN302WA) Other vegetative classification: Limited Depth Soils (G002XS301WA), Limited Depth Soils (G002XF303WA), Limited Depth Soils (G002XN302WA) Hydric soil rating: No

Minor Components

Everett

Percent of map unit: 5 percent Landform: Moraines, eskers, kames Landform position (two-dimensional): Backslope Landform position (three-dimensional): Side slope Down-slope shape: Convex Across-slope shape: Convex Hydric soil rating: No

Indianola

Percent of map unit: 5 percent Landform: Terraces, kames, eskers Landform position (three-dimensional): Tread Down-slope shape: Linear Across-slope shape: Linear Hydric soil rating: No

Shalcar

Percent of map unit: 3 percent Landform: Depressions Landform position (three-dimensional): Dip Down-slope shape: Concave Across-slope shape: Concave Hydric soil rating: Yes

Norma

Percent of map unit: 2 percent Landform: Drainageways, depressions Landform position (three-dimensional): Dip Down-slope shape: Linear, concave Across-slope shape: Concave Hydric soil rating: Yes

20—Cagey loamy sand

Map Unit Setting

National map unit symbol: 2nd8d Elevation: 330 to 980 feet Mean annual precipitation: 40 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 165 to 195 days Farmland classification: Prime farmland if irrigated

Map Unit Composition

Cagey and similar soils: 85 percent Minor components: 5 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Cagey

Setting

Landform: Terraces Parent material: Sandy glacial drift

Typical profile

H1 - 0 to 6 inches: loamy sand H2 - 6 to 28 inches: loamy sand H3 - 28 to 60 inches: fine sand

Properties and qualities

Slope: 0 to 4 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): High to very high (5.95 to 19.98 in/hr)
Depth to water table: About 18 to 30 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 4.0 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3w Hydrologic Soil Group: A Ecological site: F002XA005WA - Puget Lowlands Moist Forest Forage suitability group: Seasonally Wet Soils (G002XS201WA) Other vegetative classification: Seasonally Wet Soils (G002XS201WA) Hydric soil rating: No

Minor Components

Mckenna

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

27—Delphi very gravelly loam, 3 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2nd8m Elevation: 330 to 3,280 feet Mean annual precipitation: 50 to 75 inches Mean annual air temperature: 50 degrees F Frost-free period: 165 to 195 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Delphi and similar soils: 100 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Delphi

Setting

Landform: Till plains Parent material: Continental basal till

Typical profile

H1 - 0 to 8 inches: very gravelly loam

H2 - 8 to 13 inches: very gravelly loam

H3 - 13 to 48 inches: very gravelly silt loam

H4 - 48 to 52 inches: extremely gravelly clay loam

Properties and qualities

Slope: 3 to 15 percent
Depth to restrictive feature: 40 to 55 inches to densic material
Drainage class: Well drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: About 39 to 54 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Moderate (about 6.0 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4s Hydrologic Soil Group: B Ecological site: F002XA007WA - Puget Lowlands Wet Forest

Item 6.

Forage suitability group: Droughty Soils (G001XY402WA) *Other vegetative classification:* Droughty Soils (G001XY402WA) *Hydric soil rating:* No

30—Dystric Xerochrepts, 60 to 90 percent slopes

Map Unit Setting

National map unit symbol: 2nd8r Elevation: 0 to 3,280 feet Mean annual precipitation: 50 inches Mean annual air temperature: 50 degrees F Frost-free period: 180 days Farmland classification: Not prime farmland

Map Unit Composition

Dystric xerochrepts and similar soils: 85 percent *Minor components:* 5 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Dystric Xerochrepts

Setting

Landform: Escarpments Parent material: Colluvium and glacial till

Typical profile

H1 - 0 to 4 inches: very gravelly sandy loam *H2 - 4 to 30 inches:* very gravelly sandy loam *H3 - 30 to 34 inches:* very gravelly sandy loam

Properties and qualities

Slope: 60 to 90 percent
Depth to restrictive feature: 20 to 72 inches to densic material
Drainage class: Well drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very low (about 2.1 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 7e Hydrologic Soil Group: C Hydric soil rating: No

Minor Components

Skipopa

Percent of map unit: 5 percent *Hydric soil rating:* No

32—Everett very gravelly sandy loam, 0 to 8 percent slopes

Map Unit Setting

National map unit symbol: 2t629 Elevation: 30 to 900 feet Mean annual precipitation: 35 to 91 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 180 to 240 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Everett and similar soils: 80 percent *Minor components:* 20 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Everett

Setting

Landform: Kames, moraines, eskers Landform position (two-dimensional): Summit, shoulder Landform position (three-dimensional): Crest, interfluve Down-slope shape: Convex Across-slope shape: Convex Parent material: Sandy and gravelly glacial outwash

Typical profile

Oi - 0 to 1 inches: slightly decomposed plant material

A - 1 to 3 inches: very gravelly sandy loam

Bw - 3 to 24 inches: very gravelly sandy loam

C1 - 24 to 35 inches: very gravelly loamy sand

C2 - 35 to 60 inches: extremely cobbly coarse sand

Properties and qualities

Slope: 0 to 8 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None

Available water supply, 0 to 60 inches: Low (about 3.2 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4s Hydrologic Soil Group: A Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Droughty Soils (G002XS401WA), Droughty Soils (G002XF403WA), Droughty Soils (G002XN402WA)
Other vegetative classification: Droughty Soils (G002XS401WA), Droughty Soils (G002XF403WA), Droughty Soils (G002XN402WA) *Hydric soil rating:* No

Minor Components

Indianola

Percent of map unit: 10 percent Landform: Terraces, kames, eskers Landform position (three-dimensional): Tread Down-slope shape: Linear Across-slope shape: Linear Hydric soil rating: No

Alderwood

Percent of map unit: 10 percent Landform: Hills, ridges Landform position (two-dimensional): Summit Landform position (three-dimensional): Crest, talf Down-slope shape: Convex, linear Across-slope shape: Convex Hydric soil rating: No

33—Everett very gravelly sandy loam, 8 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2t62b Elevation: 30 to 900 feet Mean annual precipitation: 35 to 91 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 180 to 240 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Everett and similar soils: 80 percent *Minor components:* 20 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Everett

Setting

Landform: Moraines, eskers, kames Landform position (two-dimensional): Shoulder, footslope Landform position (three-dimensional): Crest, base slope Down-slope shape: Convex Across-slope shape: Convex Parent material: Sandy and gravelly glacial outwash

Typical profile

Oi - 0 to 1 inches: slightly decomposed plant material *A - 1 to 3 inches:* very gravelly sandy loam

- Bw 3 to 24 inches: very gravelly sandy loam
- C1 24 to 35 inches: very gravelly loamy sand
- C2 35 to 60 inches: extremely cobbly coarse sand

Properties and qualities

Slope: 8 to 15 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 3.2 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4s Hydrologic Soil Group: A Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Droughty Soils (G002XS401WA), Droughty Soils (G002XF403WA), Droughty Soils (G002XN402WA) Other vegetative classification: Droughty Soils (G002XS401WA), Droughty Soils (G002XF403WA), Droughty Soils (G002XN402WA) Hydric soil rating: No

Minor Components

Alderwood

Percent of map unit: 10 percent Landform: Hills, ridges Landform position (two-dimensional): Shoulder Landform position (three-dimensional): Nose slope, talf Down-slope shape: Convex, linear Across-slope shape: Convex Hydric soil rating: No

Indianola

Percent of map unit: 10 percent Landform: Terraces, kames, eskers Landform position (three-dimensional): Riser Down-slope shape: Linear Across-slope shape: Linear Hydric soil rating: No

34—Everett very gravelly sandy loam, 15 to 30 percent slopes

Map Unit Setting

National map unit symbol: 2t62c Elevation: 30 to 900 feet Mean annual precipitation: 35 to 91 inches *Mean annual air temperature:* 48 to 52 degrees F *Frost-free period:* 180 to 240 days *Farmland classification:* Farmland of statewide importance

Map Unit Composition

Everett and similar soils: 80 percent *Minor components:* 20 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Everett

Setting

Landform: Moraines, eskers, kames Landform position (two-dimensional): Backslope Landform position (three-dimensional): Side slope Down-slope shape: Convex Across-slope shape: Convex Parent material: Sandy and gravelly glacial outwash

Typical profile

Oi - 0 to 1 inches: slightly decomposed plant material

A - 1 to 3 inches: very gravelly sandy loam

Bw - 3 to 24 inches: very gravelly sandy loam

C1 - 24 to 35 inches: very gravelly loamy sand

C2 - 35 to 60 inches: extremely cobbly coarse sand

Properties and qualities

Slope: 15 to 30 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 3.2 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4e Hydrologic Soil Group: A Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Droughty Soils (G002XS401WA), Droughty Soils (G002XN402WA) Other vegetative classification: Droughty Soils (G002XS401WA), Droughty Soils (G002XN402WA) Hydric soil rating: No

Minor Components

Alderwood

Percent of map unit: 10 percent Landform: Hills, ridges Landform position (two-dimensional): Backslope Landform position (three-dimensional): Nose slope, side slope, talf Down-slope shape: Convex, linear Across-slope shape: Convex

Hydric soil rating: No

Indianola

Percent of map unit: 10 percent Landform: Terraces, kames, eskers Landform position (two-dimensional): Backslope Landform position (three-dimensional): Side slope Down-slope shape: Linear Across-slope shape: Linear Hydric soil rating: No

35—Everett very gravelly sandy loam, 30 to 50 percent slopes

Map Unit Setting

National map unit symbol: 2t62d Elevation: 30 to 900 feet Mean annual precipitation: 35 to 91 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 180 to 240 days Farmland classification: Not prime farmland

Map Unit Composition

Everett and similar soils: 80 percent *Minor components:* 20 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Everett

Setting

Landform: Moraines, eskers, kames Landform position (two-dimensional): Backslope Landform position (three-dimensional): Side slope Down-slope shape: Convex Across-slope shape: Convex Parent material: Sandy and gravelly glacial outwash

Typical profile

Oi - 0 to 1 inches: slightly decomposed plant material

A - 1 to 3 inches: very gravelly sandy loam

- Bw 3 to 24 inches: very gravelly sandy loam
- C1 24 to 35 inches: very gravelly loamy sand
- C2 35 to 60 inches: extremely cobbly coarse sand

Properties and qualities

Slope: 30 to 50 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: More than 80 inches

Frequency of flooding: None

Frequency of ponding: None *Available water supply, 0 to 60 inches:* Low (about 3.2 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 7e Hydrologic Soil Group: A Ecological site: F002XA004WA - Puget Lowlands Forest Hydric soil rating: No

Minor Components

Indianola

Percent of map unit: 10 percent Landform: Terraces, kames, eskers Landform position (two-dimensional): Backslope Landform position (three-dimensional): Side slope Down-slope shape: Linear Across-slope shape: Linear Hydric soil rating: No

Alderwood

Percent of map unit: 10 percent Landform: Hills, ridges Landform position (two-dimensional): Backslope Landform position (three-dimensional): Nose slope, side slope, talf Down-slope shape: Convex, linear Across-slope shape: Convex Hydric soil rating: No

38—Giles silt loam, 0 to 3 percent slopes

Map Unit Setting

National map unit symbol: 2ndbv Elevation: 160 to 1,640 feet Mean annual precipitation: 35 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 170 to 200 days Farmland classification: All areas are prime farmland

Map Unit Composition

Giles and similar soils: 85 percent *Minor components:* 5 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Giles

Setting

Landform: Terraces Parent material: Volcanic ash and glacial outwash

Typical profile

H1 - 0 to 10 inches: silt loam *H2 - 10 to 48 inches:* silt loam *H3 - 48 to 60 inches:* silt loam

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Well drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very high (about 14.6 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 1 Hydrologic Soil Group: B Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Soils with Few Limitations (G002XS501WA) Other vegetative classification: Soils with Few Limitations (G002XS501WA) Hydric soil rating: No

Minor Components

Yelm

Percent of map unit: 3 percent Landform: Terraces Hydric soil rating: No

Norma

Percent of map unit: 2 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

39—Giles silt loam, 3 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2ndbw Elevation: 160 to 1,640 feet Mean annual precipitation: 35 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 170 to 200 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Giles and similar soils: 85 percent

Minor components: 5 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Giles

Setting

Landform: Terraces Parent material: Volcanic ash and glacial outwash

Typical profile

H1 - 0 to 10 inches: silt loam *H2 - 10 to 48 inches:* silt loam

H3 - 48 to 60 inches: silt loam

Properties and qualities

Slope: 3 to 15 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Well drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very high (about 14.6 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3e Hydrologic Soil Group: B Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Soils with Moderate Limitations (G002XN602WA) Other vegetative classification: Soils with Moderate Limitations (G002XN602WA) Hydric soil rating: No

Minor Components

Yelm

Percent of map unit: 5 percent Hydric soil rating: No

40—Giles silt loam, 15 to 30 percent slopes

Map Unit Setting

National map unit symbol: 2nd8v Elevation: 160 to 1,640 feet Mean annual precipitation: 35 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 170 to 200 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Giles and similar soils: 85 percent *Minor components:* 5 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Giles

Setting

Landform: Escarpments Parent material: Volcanic ash and glacial outwash

Typical profile

H1 - 0 to 10 inches: silt loam *H2 - 10 to 48 inches:* silt loam *H3 - 48 to 60 inches:* silt loam

Properties and qualities

Slope: 15 to 30 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Well drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very high (about 14.6 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4e Hydrologic Soil Group: B Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Sloping to Steep Soils (G002XN702WA) Other vegetative classification: Sloping to Steep Soils (G002XN702WA) Hydric soil rating: No

Minor Components

Yelm

Percent of map unit: 5 percent Landform: Terraces Hydric soil rating: No

41—Godfrey silty clay loam

Map Unit Setting

National map unit symbol: 2nd8w Elevation: 20 to 300 feet Mean annual precipitation: 40 to 65 inches Mean annual air temperature: 50 to 54 degrees F *Frost-free period:* 150 to 200 days *Farmland classification:* Prime farmland if drained

Map Unit Composition

Godfrey, drained, and similar soils: 85 percent *Minor components:* 15 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Godfrey, Drained

Setting

Landform: Flood plains Parent material: Alluvium

Typical profile

H1 - 0 to 8 inches: silty clay loam
H2 - 8 to 52 inches: silty clay
H3 - 52 to 60 inches: silty clay

Properties and qualities

Slope: 0 to 3 percent Depth to restrictive feature: More than 80 inches Drainage class: Poorly drained Capacity of the most limiting layer to transmit water (Ksat): Very low (0.00 in/hr) Depth to water table: About 12 to 24 inches Frequency of flooding: Occasional Frequency of ponding: None Available water supply, 0 to 60 inches: High (about 9.7 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4w Hydrologic Soil Group: C/D Ecological site: F002XA008WA - Puget Lowlands Riparian Forest Forage suitability group: Seasonally Wet Soils (G002XS201WA) Other vegetative classification: Seasonally Wet Soils (G002XS201WA) Hydric soil rating: Yes

Minor Components

Sultan

Percent of map unit: 5 percent Hydric soil rating: No

Godfrey, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Newberg

Percent of map unit: 3 percent Hydric soil rating: No

Puget, undrained

Percent of map unit: 2 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

45—Hydraquents, tidal

Map Unit Setting

National map unit symbol: 2nd90 Elevation: 0 to 100 feet Mean annual precipitation: 20 to 30 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 170 to 210 days Farmland classification: Not prime farmland

Map Unit Composition

Hydraquents, tidal, and similar soils: 100 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Hydraquents, Tidal

Setting

Landform: Tidal flats Parent material: Alluvium

Typical profile

H1 - 0 to 6 inches: fine sandy loam *H2 - 6 to 60 inches:* stratified fine sandy loam to silty clay loam

Properties and qualities

Slope: 0 to 1 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Poorly drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: About 0 inches
Frequency of flooding: Very frequent
Frequency of ponding: None
Maximum salinity: Moderately saline to strongly saline (8.0 to 16.0 mmhos/cm)
Sodium adsorption ratio, maximum: 10.0
Available water supply, 0 to 60 inches: High (about 10.0 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 6w Hydrologic Soil Group: B/D Hydric soil rating: Yes

46—Indianola loamy sand, 0 to 5 percent slopes

Map Unit Setting

National map unit symbol: 2t62k Elevation: 0 to 980 feet Mean annual precipitation: 30 to 81 inches Mean annual air temperature: 48 to 50 degrees F Frost-free period: 170 to 210 days Farmland classification: Prime farmland if irrigated

Map Unit Composition

Indianola and similar soils: 85 percent *Minor components:* 15 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Indianola

Setting

Landform: Terraces, eskers, kames Landform position (three-dimensional): Tread Down-slope shape: Linear Across-slope shape: Linear Parent material: Sandy glacial outwash

Typical profile

Oi - 0 to 1 inches: slightly decomposed plant material *A - 1 to 6 inches:* loamy sand *Bw1 - 6 to 17 inches:* loamy sand *Bw2 - 17 to 27 inches:* sand *BC - 27 to 37 inches:* sand *C - 37 to 60 inches:* sand

Properties and qualities

Slope: 0 to 5 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High to very high (5.95 to 99.90 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 3.9 inches)

Interpretive groups

Land capability classification (irrigated): 4s Land capability classification (nonirrigated): 4s Hydrologic Soil Group: A Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Droughty Soils (G002XS401WA), Droughty Soils (G002XF403WA), Droughty Soils (G002XN402WA), Droughty Soils (G002XV402WA) Other vegetative classification: Droughty Soils (G002XS401WA), Droughty Soils (G002XF403WA), Droughty Soils (G002XN402WA), Droughty Soils (G002XV402WA) (G002XV402WA) Hydric soil rating: No

Minor Components

Alderwood

Percent of map unit: 5 percent Landform: Ridges, hills Landform position (two-dimensional): Summit Landform position (three-dimensional): Crest, talf Down-slope shape: Linear, convex Across-slope shape: Convex Hydric soil rating: No

Everett

Percent of map unit: 5 percent Landform: Eskers, moraines, kames Landform position (two-dimensional): Summit, shoulder Landform position (three-dimensional): Interfluve, crest Down-slope shape: Convex Across-slope shape: Convex Hydric soil rating: No

Norma

Percent of map unit: 5 percent Landform: Depressions, drainageways Landform position (three-dimensional): Dip Down-slope shape: Concave, linear Across-slope shape: Concave Hydric soil rating: Yes

47—Indianola loamy sand, 5 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2t635 Elevation: 0 to 980 feet Mean annual precipitation: 30 to 81 inches Mean annual air temperature: 48 to 50 degrees F Frost-free period: 170 to 210 days Farmland classification: Prime farmland if irrigated

Map Unit Composition

Indianola and similar soils: 85 percent Minor components: 15 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Indianola

Setting

Landform: Terraces, kames, eskers Landform position (three-dimensional): Riser Down-slope shape: Linear Across-slope shape: Linear Parent material: Sandy glacial outwash

Typical profile

Oi - 0 to 1 inches: slightly decomposed plant material *A - 1 to 6 inches:* loamy sand *Bw1 - 6 to 17 inches:* loamy sand *Bw2 - 17 to 27 inches:* sand *BC - 27 to 37 inches:* sand *C - 37 to 60 inches:* sand

Properties and qualities

Slope: 5 to 15 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High to very high (5.95 to 99.90 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 3.9 inches)

Interpretive groups

Land capability classification (irrigated): 4e Land capability classification (nonirrigated): 4s Hydrologic Soil Group: A Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Droughty Soils (G002XS401WA), Droughty Soils (G002XN402WA) Other vegetative classification: Droughty Soils (G002XS401WA), Droughty Soils (G002XN402WA) Hydric soil rating: No

Minor Components

Alderwood

Percent of map unit: 8 percent Landform: Hills, ridges Landform position (two-dimensional): Shoulder Landform position (three-dimensional): Nose slope, talf Down-slope shape: Convex, linear Across-slope shape: Convex Hydric soil rating: No

Everett

Percent of map unit: 5 percent Landform: Moraines, eskers, kames Landform position (two-dimensional): Shoulder, footslope Landform position (three-dimensional): Crest, base slope Down-slope shape: Convex Across-slope shape: Convex Hydric soil rating: No

Norma

Percent of map unit: 2 percent Landform: Drainageways, depressions Landform position (three-dimensional): Dip Down-slope shape: Linear, concave Across-slope shape: Concave Hydric soil rating: Yes

48-Indianola loamy sand, 15 to 30 percent slopes

Map Unit Setting

National map unit symbol: 2t639 Elevation: 0 to 980 feet Mean annual precipitation: 30 to 81 inches Mean annual air temperature: 48 to 50 degrees F Frost-free period: 170 to 210 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Indianola and similar soils: 85 percent Minor components: 15 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Indianola

Setting

Landform: Terraces, kames, eskers Landform position (two-dimensional): Backslope Landform position (three-dimensional): Side slope Down-slope shape: Linear Across-slope shape: Linear Parent material: Sandy glacial outwash

Typical profile

Oi - 0 to 1 inches: slightly decomposed plant material *A - 1 to 6 inches:* loamy sand *Bw1 - 6 to 17 inches:* loamy sand *Bw2 - 17 to 27 inches:* sand *BC - 27 to 37 inches:* sand *C - 37 to 60 inches:* sand

Properties and qualities

Slope: 15 to 30 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High to very high (5.95 to 99.90 in/hr)
Depth to water table: More than 80 inches

Frequency of flooding: None Frequency of ponding: None Available water supply, 0 to 60 inches: Low (about 3.9 inches)

Interpretive groups

Land capability classification (irrigated): 6e Land capability classification (nonirrigated): 4e Hydrologic Soil Group: A Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Droughty Soils (G002XS401WA), Droughty Soils (G002XN402WA) Other vegetative classification: Droughty Soils (G002XS401WA), Droughty Soils (G002XN402WA) Hydric soil rating: No

Minor Components

Alderwood

Percent of map unit: 8 percent Landform: Hills, ridges Landform position (two-dimensional): Backslope Landform position (three-dimensional): Nose slope, side slope, talf Down-slope shape: Convex, linear Across-slope shape: Convex Hydric soil rating: No

Everett

Percent of map unit: 5 percent Landform: Moraines, eskers, kames Landform position (two-dimensional): Backslope Landform position (three-dimensional): Side slope Down-slope shape: Convex Across-slope shape: Convex Hydric soil rating: No

Norma

Percent of map unit: 2 percent Landform: Drainageways, depressions Landform position (three-dimensional): Dip Down-slope shape: Linear, concave Across-slope shape: Concave Hydric soil rating: Yes

51—Kapowsin silt loam, 3 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2ndbx Elevation: 50 to 900 feet Mean annual precipitation: 30 to 50 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 220 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Kapowsin and similar soils: 85 percent *Minor components:* 8 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Kapowsin

Setting

Landform: Till plains Parent material: Compact basal till

Typical profile

H1 - 0 to 4 inches: silt loam

H2 - 4 to 22 inches: silt loam

H3 - 22 to 30 inches: gravelly loam

H4 - 30 to 34 inches: gravelly loam

Properties and qualities

Slope: 3 to 15 percent
Depth to restrictive feature: 20 to 40 inches to densic material
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: About 12 to 24 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Moderate (about 6.1 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3e Hydrologic Soil Group: C/D Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Limited Depth Soils (G002XN302WA) Other vegetative classification: Limited Depth Soils (G002XN302WA) Hydric soil rating: No

Minor Components

Norma

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Skipopa

Percent of map unit: 3 percent Other vegetative classification: Seasonally Wet Soils (G002XN202WA) Hydric soil rating: No

52—Kapowsin silt loam, 15 to 30 percent slopes

Map Unit Setting

National map unit symbol: 2ndby Elevation: 50 to 900 feet Mean annual precipitation: 30 to 50 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 220 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Kapowsin and similar soils: 85 percent *Minor components:* 10 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Kapowsin

Setting

Landform: Till plains Parent material: Compact basal till

Typical profile

- H1 0 to 4 inches: silt loam
- H2 4 to 22 inches: silt loam
- H3 22 to 30 inches: gravelly loam
- H4 30 to 34 inches: gravelly loam

Properties and qualities

Slope: 15 to 30 percent
Depth to restrictive feature: 20 to 40 inches to densic material
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: About 12 to 24 inches
Frequency of flooding: None

Frequency of ponding: None

Available water supply, 0 to 60 inches: Moderate (about 6.1 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4e Hydrologic Soil Group: C/D Ecological site: F002XA004WA - Puget Lowlands Forest Forage suitability group: Limited Depth Soils (G002XN302WA) Other vegetative classification: Limited Depth Soils (G002XN302WA) Hydric soil rating: No

Minor Components

Hoogdal

Percent of map unit: 5 percent Hydric soil rating: No

Indianola

Percent of map unit: 5 percent Hydric soil rating: No

53—Kapowsin silt loam, 30 to 50 percent slopes

Map Unit Setting

National map unit symbol: 2ndbz Elevation: 50 to 900 feet Mean annual precipitation: 30 to 50 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 220 days Farmland classification: Not prime farmland

Map Unit Composition

Kapowsin and similar soils: 85 percent *Minor components:* 5 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Kapowsin

Setting

Landform: Till plains Parent material: Compact basal till

Typical profile

H1 - 0 to 4 inches: silt loam
H2 - 4 to 22 inches: silt loam
H3 - 22 to 30 inches: gravelly loam
H4 - 30 to 34 inches: gravelly loam

Properties and qualities

Slope: 30 to 50 percent
Depth to restrictive feature: 20 to 40 inches to densic material
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: About 12 to 24 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Moderate (about 6.1 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 7e

Hydrologic Soil Group: C/D *Ecological site:* F002XA004WA - Puget Lowlands Forest *Hydric soil rating:* No

Minor Components

Hoogdal

Percent of map unit: 5 percent *Hydric soil rating:* No

65—McKenna gravelly silt loam, 0 to 5 percent slopes

Map Unit Setting

National map unit symbol: 2nd9g Elevation: 50 to 500 feet Mean annual precipitation: 30 to 60 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 180 days Farmland classification: Prime farmland if drained

Map Unit Composition

Mckenna and similar soils: 85 percent *Minor components:* 15 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Mckenna

Setting

Landform: Depressions, drainageways Parent material: Glacial drift

Typical profile

H1 - 0 to 9 inches: gravelly silt loam H2 - 9 to 13 inches: gravelly silt loam H3 - 13 to 36 inches: very gravelly loam H4 - 36 to 40 inches: very gravelly loam

Properties and qualities

Slope: 0 to 5 percent
Depth to restrictive feature: 20 to 39 inches to densic material
Drainage class: Poorly drained
Capacity of the most limiting layer to transmit water (Ksat): Very low to moderately low (0.00 to 0.06 in/hr)
Depth to water table: About 0 to 6 inches
Frequency of flooding: None
Frequency of ponding: Frequent
Available water supply, 0 to 60 inches: Low (about 5.4 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 6w

Hydrologic Soil Group: D *Ecological site:* F002XA007WA - Puget Lowlands Wet Forest *Forage suitability group:* Wet Soils (G002XS101WA) *Other vegetative classification:* Wet Soils (G002XS101WA) *Hydric soil rating:* Yes

Minor Components

Bellingham, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XN102WA) Hydric soil rating: Yes

Norma

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Skipopa

Percent of map unit: 5 percent Other vegetative classification: Seasonally Wet Soils (G002XN202WA) Hydric soil rating: No

69—Mukilteo muck

Map Unit Setting

National map unit symbol: 2nd9l Elevation: 0 to 1,000 feet Mean annual precipitation: 40 to 70 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 250 days Farmland classification: Prime farmland if drained

Map Unit Composition

Mukilteo, undrained, and similar soils: 90 percent *Minor components:* 10 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Mukilteo, Undrained

Setting

Landform: Depressions Parent material: Herbaceous organic material

Typical profile

Oa - 0 to 6 inches: muck *Oe - 6 to 60 inches:* mucky peat

Properties and qualities

Slope: 0 to 2 percent

Custom Soil Resource Report

Depth to restrictive feature: More than 80 inches Drainage class: Very poorly drained Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr) Depth to water table: About 0 inches Frequency of flooding: None Frequency of ponding: Frequent Available water supply, 0 to 60 inches: Very high (about 26.9 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 5w Hydrologic Soil Group: B/D Ecological site: R002XA003WA - Puget Lowlands Bogs and Fens Forage suitability group: Wet Soils (G002XS101WA) Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Minor Components

Shalcar

Percent of map unit: 10 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

70-Mukilteo muck, drained

Map Unit Setting

National map unit symbol: 2ndc5 Elevation: 0 to 1,000 feet Mean annual precipitation: 40 to 70 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 250 days Farmland classification: Prime farmland if drained

Map Unit Composition

Mukilteo, drained, and similar soils: 90 percent *Minor components:* 10 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Mukilteo, Drained

Setting

Landform: Depressions Parent material: Herbaceous organic material

Typical profile

Oa - 0 to 6 inches: muck *Oe2 - 6 to 60 inches:* mucky peat

45

Properties and qualities

Slope: 0 to 2 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Very poorly drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: About 0 to 24 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very high (about 26.9 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 5w Hydrologic Soil Group: B/D Ecological site: R002XA003WA - Puget Lowlands Bogs and Fens Forage suitability group: Seasonally Wet Soils (G002XS201WA) Other vegetative classification: Seasonally Wet Soils (G002XS201WA) Hydric soil rating: Yes

Minor Components

Shalcar

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Mukilteo, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

73—Nisqually loamy fine sand, 0 to 3 percent slopes

Map Unit Setting

National map unit symbol: 2ndc8 Elevation: 160 to 1,310 feet Mean annual precipitation: 40 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 150 to 200 days Farmland classification: Prime farmland if irrigated

Map Unit Composition

Nisqually and similar soils: 85 percent Minor components: 5 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Nisqually

Setting

Landform: Terraces Parent material: Sandy glacial outwash

Typical profile

H1 - 0 to 5 inches: loamy fine sand H2 - 5 to 31 inches: loamy fine sand H3 - 31 to 60 inches: loamy sand

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 4.9 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3s Hydrologic Soil Group: A Ecological site: R002XA006WA - Puget Lowlands Prairie Forage suitability group: Droughty Soils (G002XS401WA) Other vegetative classification: Droughty Soils (G002XS401WA) Hydric soil rating: No

Minor Components

Yelm

Percent of map unit: 3 percent Hydric soil rating: No

Norma

Percent of map unit: 2 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

74—Nisqually loamy fine sand, 3 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2ndc9 Elevation: 160 to 1,310 feet Mean annual precipitation: 40 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 150 to 200 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Nisqually and similar soils: 85 percent *Minor components:* 5 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Nisqually

Setting

Landform: Terraces Parent material: Sandy glacial outwash

Typical profile

H1 - 0 to 5 inches: loamy fine sand H2 - 5 to 31 inches: loamy fine sand H3 - 31 to 60 inches: loamy sand

Properties and qualities

Slope: 3 to 15 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 4.9 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3e Hydrologic Soil Group: A Ecological site: R002XA006WA - Puget Lowlands Prairie Forage suitability group: Droughty Soils (G002XS401WA) Other vegetative classification: Droughty Soils (G002XS401WA) Hydric soil rating: No

Minor Components

Yelm

Percent of map unit: 3 percent Hydric soil rating: No

Norma

Percent of map unit: 2 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

75—Norma fine sandy loam

Map Unit Setting

National map unit symbol: 2ndcb Elevation: 0 to 1,000 feet Mean annual precipitation: 35 to 60 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 200 days Farmland classification: Prime farmland if drained

Map Unit Composition

Norma, fine sandy loam, and similar soils: 90 percent *Minor components:* 10 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Norma, Fine Sandy Loam

Setting

Landform: Depressions, drainageways Parent material: Alluvium

Typical profile

H1 - 0 to 7 inches: fine sandy loam *H2 - 7 to 25 inches:* fine sandy loam *H3 - 25 to 60 inches:* sandy loam

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Poorly drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: About 0 to 12 inches
Frequency of flooding: None
Frequency of ponding: Frequent
Available water supply, 0 to 60 inches: Moderate (about 8.4 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 5w Hydrologic Soil Group: A/D Ecological site: F002XA007WA - Puget Lowlands Wet Forest Forage suitability group: Wet Soils (G002XS101WA) Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Minor Components

Norma, silt loam Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Alderwood

Percent of map unit: 5 percent Hydric soil rating: No

76—Norma silt loam

Map Unit Setting

National map unit symbol: 2ndcc Elevation: 0 to 1,000 feet Mean annual precipitation: 35 to 60 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 200 days Farmland classification: Prime farmland if drained

Map Unit Composition

Norma, silt loam, and similar soils: 90 percent Minor components: 10 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Norma, Silt Loam

Setting

Landform: Depressions, drainageways Parent material: Alluvium

Typical profile

H1 - 0 to 8 inches: silt loam

- H2 8 to 30 inches: sandy loam
- H3 30 to 60 inches: sandy loam

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Poorly drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: About 0 to 12 inches
Frequency of flooding: None
Frequency of ponding: Frequent

Available water supply, 0 to 60 inches: Moderate (about 8.9 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 5w Hydrologic Soil Group: B/D Ecological site: F002XA007WA - Puget Lowlands Wet Forest Forage suitability group: Wet Soils (G002XS101WA) *Other vegetative classification:* Wet Soils (G002XS101WA) *Hydric soil rating:* Yes

Minor Components

Norma, fine sandy loam

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Alderwood

Percent of map unit: 5 percent Hydric soil rating: No

84—Pilchuck loamy sand

Map Unit Setting

National map unit symbol: 2nd9t Elevation: 70 to 1,970 feet Mean annual precipitation: 35 to 60 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 160 to 210 days Farmland classification: Prime farmland if irrigated

Map Unit Composition

Pilchuck and similar soils: 85 percent *Minor components:* 15 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Pilchuck

Setting

Landform: Flood plains Parent material: Alluvium

Typical profile

H1 - 0 to 6 inches: loamy sand H2 - 6 to 60 inches: fine sand

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat excessively drained
Capacity of the most limiting layer to transmit water (Ksat): High to very high (5.95 to 19.98 in/hr)
Depth to water table: About 24 to 48 inches
Frequency of flooding: Occasional
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 4.1 inches)

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3w Hydrologic Soil Group: A Ecological site: F002XA008WA - Puget Lowlands Riparian Forest Forage suitability group: Droughty Soils (G002XS401WA) Other vegetative classification: Droughty Soils (G002XS401WA) Hydric soil rating: No

Minor Components

Puget, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Sultan

Percent of map unit: 5 percent *Hydric soil rating:* No

Newberg

Percent of map unit: 5 percent Hydric soil rating: No

85—Pits, gravel

Map Unit Composition Pits: 100 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Pits

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 8 Hydric soil rating: No

88—Puget silt loam

Map Unit Setting

National map unit symbol: 2nd9y Elevation: 10 to 650 feet Mean annual precipitation: 35 to 55 inches Mean annual air temperature: 48 to 50 degrees F Frost-free period: 160 to 180 days Farmland classification: Prime farmland if drained

Map Unit Composition

Puget, drained, and similar soils: 85 percent *Minor components:* 15 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Puget, Drained

Setting

Landform: Flood plains Parent material: Alluvium

Typical profile

H1 - 0 to 9 inches: silt loam H2 - 9 to 60 inches: silt loam

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Poorly drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high (0.20 to 0.57 in/hr)
Depth to water table: About 12 to 35 inches
Frequency of flooding: Occasional
Frequency of ponding: None
Available water supply, 0 to 60 inches: High (about 12.0 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3w Hydrologic Soil Group: C Ecological site: F002XA008WA - Puget Lowlands Riparian Forest Forage suitability group: Soils with Few Limitations (G002XS501WA) Other vegetative classification: Soils with Few Limitations (G002XS501WA) Hydric soil rating: Yes

Minor Components

Puget, undrained

Percent of map unit: 5 percent Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Newberg

Percent of map unit: 5 percent Hydric soil rating: No

Semiahmoo, undrained

Percent of map unit: 3 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Sultan

Percent of map unit: 2 percent Hydric soil rating: No

89—Puyallup silt loam

Map Unit Setting

National map unit symbol: 2nd9z Elevation: 70 to 1,970 feet Mean annual precipitation: 35 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 170 to 200 days Farmland classification: All areas are prime farmland

Map Unit Composition

Puyallup and similar soils: 85 percent *Minor components:* 10 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Puyallup

Setting

Landform: Terraces, flood plains Parent material: Alluvium

Typical profile

H1 - 0 to 10 inches: silt loam H2 - 10 to 19 inches: fine sandy loam H3 - 19 to 60 inches: sand

Properties and qualities

Slope: 0 to 3 percent
 Depth to restrictive feature: 14 to 20 inches to strongly contrasting textural stratification
 Drainage class: Well drained
 Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high

(0.57 to 1.98 in/hr) Depth to water table: More than 80 inches Frequency of flooding: Occasional Frequency of ponding: None Available water supply, 0 to 60 inches: Low (about 3.2 inches)

Interpretive groups

Land capability classification (irrigated): 3w Land capability classification (nonirrigated): 3w Hydrologic Soil Group: B Ecological site: F002XA008WA - Puget Lowlands Riparian Forest Forage suitability group: Droughty Soils (G002XS401WA) Other vegetative classification: Droughty Soils (G002XS401WA) Hydric soil rating: No

Minor Components

Newberg

Percent of map unit: 5 percent Hydric soil rating: No

Semiahmoo, undrained

Percent of map unit: 3 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Sultan

Percent of map unit: 2 percent Hydric soil rating: No

98—Salkum silty clay loam, 8 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2ndcq Elevation: 200 to 1,000 feet Mean annual precipitation: 40 to 70 inches Mean annual air temperature: 48 to 50 degrees F Frost-free period: 150 to 210 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Salkum and similar soils: 95 percent Minor components: 5 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Salkum

Setting

Landform: Terraces Parent material: Highly weatherd glacial drift

Typical profile

H1 - 0 to 12 inches: silty clay loam H2 - 12 to 51 inches: silty clay H3 - 51 to 60 inches: silty clay

Properties and qualities

Slope: 8 to 15 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Well drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high (0.20 to 0.57 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None

Available water supply, 0 to 60 inches: High (about 9.5 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3e Hydrologic Soil Group: B Ecological site: F002XA005WA - Puget Lowlands Moist Forest Forage suitability group: Soils with Moderate Limitations (G001XY602WA) Other vegetative classification: Soils with Moderate Limitations (G001XY602WA) Hydric soil rating: No

Minor Components

Scamman

Percent of map unit: 5 percent Landform: Terraces Other vegetative classification: Seasonally Wet Soils (G003XF203WA) Hydric soil rating: No

102—Schneider very gravelly loam, 20 to 40 percent slopes

Map Unit Setting

National map unit symbol: 2nd7p Elevation: 50 to 1,800 feet Mean annual precipitation: 60 to 75 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 200 days Farmland classification: Not prime farmland

Map Unit Composition

Schneider and similar soils: 100 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Schneider

Setting

Landform: Mountains

Typical profile

H1 - 0 to 6 inches: very gravelly loam

- H2 6 to 32 inches: very gravelly silt loam
- H3 32 to 55 inches: extremely gravelly silt loam
- H4 55 to 59 inches: unweathered bedrock

Properties and qualities

Slope: 20 to 40 percent

Depth to restrictive feature: 40 to 60 inches to lithic bedrock

Drainage class: Well drained

Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)

Depth to water table: More than 80 inches

Frequency of flooding: None *Frequency of ponding:* None *Available water supply, 0 to 60 inches:* Low (about 4.6 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 6e Hydrologic Soil Group: B Ecological site: F002XA005WA - Puget Lowlands Moist Forest Hydric soil rating: No

103—Schneider very gravelly loam, 40 to 65 percent slopes

Map Unit Setting

National map unit symbol: 2nd7q Elevation: 50 to 1,800 feet Mean annual precipitation: 60 to 75 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 200 days Farmland classification: Not prime farmland

Map Unit Composition

Schneider and similar soils: 100 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Schneider

Setting

Landform: Mountains

Typical profile

H1 - 0 to 6 inches: very gravelly loam

- H2 6 to 32 inches: very gravelly silt loam
- H3 32 to 55 inches: extremely gravelly silt loam

H4 - 55 to 59 inches: unweathered bedrock

Properties and qualities

Slope: 40 to 65 percent
Depth to restrictive feature: 40 to 60 inches to lithic bedrock
Drainage class: Well drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: More than 80 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Low (about 4.6 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 7e Hydrologic Soil Group: B Ecological site: F002XA005WA - Puget Lowlands Moist Forest Hydric soil rating: No

104—Semiahmoo muck

Map Unit Setting

National map unit symbol: 2nd7r Elevation: 10 to 1,300 feet Mean annual precipitation: 4 to 70 inches Mean annual air temperature: 46 to 50 degrees F Frost-free period: 125 to 250 days Farmland classification: Prime farmland if drained

Map Unit Composition

Semiahmoo, drained, and similar soils: 85 percent Minor components: 15 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Semiahmoo, Drained

Setting

Landform: Flood plains *Parent material:* Herbaceous organic material

Typical profile

Oa1 - 0 to 6 inches: muck *Oa2 - 6 to 60 inches:* muck

Properties and qualities

Slope: 0 to 1 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Very poorly drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: About 0 to 12 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Very high (about 26.9 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 5w Hydrologic Soil Group: B/D Ecological site: R002XA003WA - Puget Lowlands Bogs and Fens Forage suitability group: Wet Soils (G002XS101WA) Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Minor Components

Shalcar variant

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Semiahmoo, undrained

Percent of map unit: 5 percent Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Puget, undrained

Percent of map unit: 3 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Sultan

Percent of map unit: 2 percent Hydric soil rating: No

106—Shalcar variant muck

Map Unit Setting

National map unit symbol: 2nd7t Elevation: 70 to 980 feet Mean annual precipitation: 40 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 150 to 200 days Farmland classification: Prime farmland if drained

Map Unit Composition

Shalcar variant and similar soils: 85 percent Minor components: 15 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Shalcar Variant

Setting

Landform: Flood plains Parent material: Organic material over alluvium

Typical profile

Oa1 - 0 to 6 inches: muck *Oa2 - 6 to 20 inches:* muck *H3 - 20 to 60 inches:* clay

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Very poorly drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately low to moderately high (0.06 to 0.20 in/hr)
Depth to water table: About 0 inches
Frequency of flooding: Occasional
Frequency of ponding: Frequent
Available water supply, 0 to 60 inches: Very high (about 15.4 inches)

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 5w Hydrologic Soil Group: C/D Ecological site: R002XA003WA - Puget Lowlands Bogs and Fens Forage suitability group: Wet Soils (G002XS101WA) Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Minor Components

Sultan

Percent of map unit: 5 percent Hydric soil rating: No

Puget, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Semiahmoo, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

108—Skipopa silt loam, 3 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2nd7w Elevation: 490 to 980 feet Mean annual precipitation: 30 to 50 inches Mean annual air temperature: 48 to 50 degrees F Frost-free period: 160 to 200 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Skipopa and similar soils: 90 percent *Minor components:* 10 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Skipopa

Setting

Landform: Terraces Parent material: Volcanic ash over glaciolacustrine deposits

Typical profile

H1 - 0 to 8 inches: silt loam H2 - 8 to 18 inches: silt loam
H3 - 18 to 60 inches: clay

Properties and qualities

Slope: 3 to 15 percent Depth to restrictive feature: 10 to 20 inches to abrupt textural change Drainage class: Somewhat poorly drained Capacity of the most limiting layer to transmit water (Ksat): Very low (0.00 in/hr) Depth to water table: About 12 to 24 inches Frequency of flooding: None Frequency of ponding: None Available water supply, 0 to 60 inches: Low (about 5.3 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4w Hydrologic Soil Group: D Ecological site: F002XA007WA - Puget Lowlands Wet Forest Forage suitability group: Seasonally Wet Soils (G002XN202WA) Other vegetative classification: Seasonally Wet Soils (G002XN202WA) Hydric soil rating: No

Minor Components

Yelm

Percent of map unit: 10 percent Hydric soil rating: No

109—Spana gravelly loam

Map Unit Setting

National map unit symbol: 2nd7x Elevation: 330 to 1,640 feet Mean annual precipitation: 25 to 45 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 200 days Farmland classification: Prime farmland if drained

Map Unit Composition

Spana and similar soils: 100 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Spana

Setting

Landform: Drainageways, outwash plains Parent material: Glacial outwash

Typical profile

H1 - 0 to 22 inches: gravelly loam H2 - 22 to 26 inches: gravelly loam H3 - 26 to 38 inches: gravelly loam

H4 - 38 to 60 inches: extremely gravelly sandy loam

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Somewhat poorly drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: About 12 to 35 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: Moderate (about 6.3 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3w Hydrologic Soil Group: B Ecological site: R002XA006WA - Puget Lowlands Prairie Forage suitability group: Soils with Few Limitations (G002XS501WA) Other vegetative classification: Soils with Few Limitations (G002XS501WA) Hydric soil rating: No

110—Spanaway gravelly sandy loam, 0 to 3 percent slopes

Map Unit Setting

National map unit symbol: 2ndb6 Elevation: 330 to 1,310 feet Mean annual precipitation: 35 to 65 inches Mean annual air temperature: 50 degrees F Frost-free period: 150 to 200 days Farmland classification: Prime farmland if irrigated

Map Unit Composition

Spanaway and similar soils: 100 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Spanaway

Setting

Landform: Terraces, outwash plains Parent material: Volcanic ash over gravelly outwash

Typical profile

H1 - 0 to 15 inches: gravelly sandy loam
H2 - 15 to 20 inches: very gravelly loam
H3 - 20 to 60 inches: extremely gravelly sand

Properties and qualities

Slope: 0 to 3 percent Depth to restrictive feature: More than 80 inches Drainage class: Somewhat excessively drained Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr) Depth to water table: More than 80 inches Frequency of flooding: None Frequency of ponding: None Available water supply, 0 to 60 inches: Low (about 3.8 inches)

Interpretive groups

Land capability classification (irrigated): 3s Land capability classification (nonirrigated): 3s Hydrologic Soil Group: A Ecological site: R002XA006WA - Puget Lowlands Prairie Forage suitability group: Droughty Soils (G002XS401WA) Other vegetative classification: Droughty Soils (G002XS401WA) Hydric soil rating: No

115—Sultan silt loam

Map Unit Setting

National map unit symbol: 2ndbc Elevation: 0 to 150 feet Mean annual precipitation: 35 to 55 inches Mean annual air temperature: 48 to 50 degrees F Frost-free period: 150 to 200 days Farmland classification: All areas are prime farmland

Map Unit Composition

Sultan and similar soils: 85 percent *Minor components:* 8 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Sultan

Setting

Landform: Flood plains Parent material: Alluvium

Typical profile

H1 - 0 to 7 inches: silt loam H2 - 7 to 60 inches: silt loam

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): Moderately high to high (0.57 to 1.98 in/hr)
Depth to water table: About 24 to 48 inches
Frequency of flooding: Occasional
Frequency of ponding: None
Available water supply, 0 to 60 inches: High (about 11.4 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 3w

Item 6.

Hydrologic Soil Group: C Ecological site: F002XA008WA - Puget Lowlands Riparian Forest Forage suitability group: Seasonally Wet Soils (G002XS201WA) Other vegetative classification: Seasonally Wet Soils (G002XS201WA) Hydric soil rating: No

Minor Components

Godfrey, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Puget, undrained

Percent of map unit: 3 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

120—Tisch silt loam

Map Unit Setting

National map unit symbol: 2nd82 Elevation: 50 to 1,000 feet Mean annual precipitation: 20 to 60 inches Mean annual air temperature: 48 to 52 degrees F Frost-free period: 150 to 250 days Farmland classification: Prime farmland if drained

Map Unit Composition

Tisch, drained, and similar soils: 80 percent *Minor components:* 20 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Tisch, Drained

Setting

Landform: Depressions, drainageways Parent material: Alluvium, volcanic ash, and diatomaceous earth

Typical profile

H1 - 0 to 11 inches: silt loam *H2 - 11 to 50 inches:* silt loam *Oa - 50 to 60 inches:* muck

Properties and qualities

Slope: 0 to 3 percent *Depth to restrictive feature:* More than 80 inches *Drainage class:* Very poorly drained Capacity of the most limiting layer to transmit water (Ksat): Moderately high (0.20 to 0.57 in/hr) Depth to water table: About 0 to 12 inches Frequency of flooding: None Frequency of ponding: None Available water supply, 0 to 60 inches: Very high (about 25.3 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 5w Hydrologic Soil Group: C/D Ecological site: F002XA007WA - Puget Lowlands Wet Forest Forage suitability group: Wet Soils (G002XS101WA) Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Minor Components

Tisch, undrained

Percent of map unit: 5 percent Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Everson, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Dupont, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Mckenna

Percent of map unit: 4 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Giles

Percent of map unit: 1 percent Hydric soil rating: No

125—Xerorthents, 0 to 5 percent slopes

Map Unit Setting

National map unit symbol: 2nd87 Elevation: 0 to 2,620 feet Mean annual precipitation: 30 to 60 inches Mean annual air temperature: 39 to 50 degrees F *Frost-free period:* 150 to 200 days *Farmland classification:* Not prime farmland

Map Unit Composition

Xerorthents and similar soils: 100 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Xerorthents

Setting

Landform: Tidal flats *Parent material:* Sandy and loamy cut and fill material

Typical profile

H1 - 0 to 60 inches: variable

Properties and qualities

Slope: 0 to 5 percent Depth to restrictive feature: More than 80 inches Drainage class: Somewhat excessively drained Depth to water table: About 24 inches Frequency of flooding: None Frequency of ponding: None

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 7s Hydric soil rating: No

126—Yelm fine sandy loam, 0 to 3 percent slopes

Map Unit Setting

National map unit symbol: 2nd88 Elevation: 80 to 980 feet Mean annual precipitation: 30 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 170 to 200 days Farmland classification: All areas are prime farmland

Map Unit Composition

Yelm and similar soils: 85 percent Minor components: 13 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Yelm

Setting

Landform: Outwash terraces Parent material: Glacial outwash

Typical profile

H1 - 0 to 8 inches: fine sandy loam *H2 - 8 to 46 inches:* fine sandy loam

H3 - 46 to 60 inches: loamy sand

Properties and qualities

Slope: 0 to 3 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: About 18 to 36 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: High (about 11.0 inches)

Interpretive groups

Land capability classification (irrigated): 3w Land capability classification (nonirrigated): 3w Hydrologic Soil Group: B Ecological site: F002XA005WA - Puget Lowlands Moist Forest Forage suitability group: Seasonally Wet Soils (G002XS201WA) Other vegetative classification: Seasonally Wet Soils (G002XS201WA) Hydric soil rating: No

Minor Components

Everson, undrained

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Norma

Percent of map unit: 5 percent Landform: Depressions Other vegetative classification: Wet Soils (G002XS101WA) Hydric soil rating: Yes

Skipopa

Percent of map unit: 3 percent Other vegetative classification: Seasonally Wet Soils (G002XN202WA) Hydric soil rating: No

127—Yelm fine sandy loam, 3 to 15 percent slopes

Map Unit Setting

National map unit symbol: 2nd89 Elevation: 80 to 980 feet Mean annual precipitation: 30 to 60 inches Mean annual air temperature: 50 degrees F Frost-free period: 170 to 200 days Farmland classification: Farmland of statewide importance

Map Unit Composition

Yelm and similar soils: 85 percent Minor components: 3 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Yelm

Setting

Landform: Outwash terraces Parent material: Glacial outwash

Typical profile

H1 - 0 to 8 inches: fine sandy loam H2 - 8 to 46 inches: fine sandy loam H3 - 46 to 60 inches: loamy sand

Properties and qualities

Slope: 3 to 15 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: About 18 to 36 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: High (about 11.0 inches)

Interpretive groups

Land capability classification (irrigated): 4e Land capability classification (nonirrigated): 3e Hydrologic Soil Group: B Ecological site: F002XA005WA - Puget Lowlands Moist Forest Forage suitability group: Soils with Moderate Limitations (G002XS601WA) Other vegetative classification: Soils with Moderate Limitations (G002XS601WA) Hydric soil rating: No

Minor Components

Skipopa

Percent of map unit: 3 percent Other vegetative classification: Seasonally Wet Soils (G002XN202WA) Hydric soil rating: No

128—Yelm fine sandy loam, 15 to 30 percent slopes

Map Unit Setting

National map unit symbol: 2nd8b Elevation: 80 to 980 feet Mean annual precipitation: 30 to 60 inches Mean annual air temperature: 50 degrees F *Frost-free period:* 170 to 200 days *Farmland classification:* Farmland of statewide importance

Map Unit Composition

Yelm and similar soils: 85 percent Minor components: 2 percent Estimates are based on observations, descriptions, and transects of the mapunit.

Description of Yelm

Setting

Landform: Outwash terraces Parent material: Glacial outwash

Typical profile

H1 - 0 to 8 inches: fine sandy loam H2 - 8 to 46 inches: fine sandy loam H3 - 46 to 60 inches: loamy sand

Properties and qualities

Slope: 15 to 30 percent
Depth to restrictive feature: More than 80 inches
Drainage class: Moderately well drained
Capacity of the most limiting layer to transmit water (Ksat): High (1.98 to 5.95 in/hr)
Depth to water table: About 18 to 36 inches
Frequency of flooding: None
Frequency of ponding: None
Available water supply, 0 to 60 inches: High (about 11.0 inches)

Interpretive groups

Land capability classification (irrigated): None specified Land capability classification (nonirrigated): 4e Hydrologic Soil Group: B Ecological site: F002XA005WA - Puget Lowlands Moist Forest Forage suitability group: Sloping to Steep Soils (G002XS701WA) Other vegetative classification: Sloping to Steep Soils (G002XS701WA) Hydric soil rating: No

Minor Components

Hoogdal

Percent of map unit: 2 percent Hydric soil rating: No

129—Water

Map Unit Composition

Water: 100 percent *Estimates are based on observations, descriptions, and transects of the mapunit.*

Description of Water

Setting

Landform: Alluvial cones

Soil Information for All Uses

Soil Reports

The Soil Reports section includes various formatted tabular and narrative reports (tables) containing data for each selected soil map unit and each component of each unit. No aggregation of data has occurred as is done in reports in the Soil Properties and Qualities and Suitabilities and Limitations sections.

The reports contain soil interpretive information as well as basic soil properties and qualities. A description of each report (table) is included.

Land Classifications

This folder contains a collection of tabular reports that present a variety of soil groupings. The reports (tables) include all selected map units and components for each map unit. Land classifications are specified land use and management groupings that are assigned to soil areas because combinations of soil have similar behavior for specified practices. Most are based on soil properties and other factors that directly influence the specific use of the soil. Example classifications include ecological site classification, farmland classification, irrigated and nonirrigated land capability classification, and hydric rating.

Prime and other Important Farmlands

This table lists the map units in the survey area that are considered important farmlands. Important farmlands consist of prime farmland, unique farmland, and farmland of statewide or local importance. This list does not constitute a recommendation for a particular land use.

In an effort to identify the extent and location of important farmlands, the Natural Resources Conservation Service, in cooperation with other interested Federal, State, and local government organizations, has inventoried land that can be used for the production of the Nation's food supply.

Prime farmland is of major importance in meeting the Nation's short- and long-range needs for food and fiber. Because the supply of high-quality farmland is limited, the U.S. Department of Agriculture recognizes that responsible levels of government, as well as individuals, should encourage and facilitate the wise use of our Nation's prime farmland.

Prime farmland, as defined by the U.S. Department of Agriculture, is land that has the best combination of physical and chemical characteristics for producing food. feed, forage, fiber, and oilseed crops and is available for these uses. It could be cultivated land, pastureland, forestland, or other land, but it is not urban or built-up land or water areas. The soil quality, growing season, and moisture supply are those needed for the soil to economically produce sustained high yields of crops when proper management, including water management, and acceptable farming methods are applied. In general, prime farmland has an adequate and dependable supply of moisture from precipitation or irrigation, a favorable temperature and growing season, acceptable acidity or alkalinity, an acceptable salt and sodium content, and few or no rocks. The water supply is dependable and of adequate quality. Prime farmland is permeable to water and air. It is not excessively erodible or saturated with water for long periods, and it either is not frequently flooded during the growing season or is protected from flooding. Slope ranges mainly from 0 to 6 percent. More detailed information about the criteria for prime farmland is available at the local office of the Natural Resources Conservation Service.

For some of the soils identified in the table as prime farmland, measures that overcome a hazard or limitation, such as flooding, wetness, and droughtiness, are needed. Onsite evaluation is needed to determine whether or not the hazard or limitation has been overcome by corrective measures.

A recent trend in land use in some areas has been the loss of some prime farmland to industrial and urban uses. The loss of prime farmland to other uses puts pressure on marginal lands, which generally are more erodible, droughty, and less productive and cannot be easily cultivated.

Unique farmland is land other than prime farmland that is used for the production of specific high-value food and fiber crops, such as citrus, tree nuts, olives, cranberries, and other fruits and vegetables. It has the special combination of soil quality, growing season, moisture supply, temperature, humidity, air drainage, elevation, and aspect needed for the soil to economically produce sustainable high yields of these crops when properly managed. The water supply is dependable and of adequate quality. Nearness to markets is an additional consideration. Unique farmland is not based on national criteria. It commonly is in areas where there is a special microclimate, such as the wine country in California.

In some areas, land that does not meet the criteria for prime or unique farmland is considered to be *farmland of statewide importance* for the production of food, feed, fiber, forage, and oilseed crops. The criteria for defining and delineating farmland of statewide importance are determined by the appropriate State agencies. Generally, this land includes areas of soils that nearly meet the requirements for prime farmland and that economically produce high yields of crops when treated and managed according to acceptable farming methods. Some areas may produce as high a yield as prime farmland if conditions are favorable. Farmland of statewide importance may include tracts of land that have been designated for agriculture by State law.

In some areas that are not identified as having national or statewide importance, land is considered to be *farmland of local importance* for the production of food, feed, fiber, forage, and oilseed crops. This farmland is identified by the appropriate local agencies. Farmland of local importance may include tracts of land that have been designated for agriculture by local ordinance.

Report—Prime and other Important Farmlands

Prime and other Important Farmlands–Thurston County Area, Washington			
Map Symbol	Map Unit Name	Farmland Classification	
1	Alderwood gravelly sandy loam, 0 to 8 percent slopes	Prime farmland if irrigated	
2	Alderwood gravelly sandy loam, 8 to 15 percent slopes	Prime farmland if irrigated	
3	Alderwood gravelly sandy loam, 15 to 30 percent slopes	Farmland of statewide importance	
20	Cagey loamy sand	Prime farmland if irrigated	
27	Delphi very gravelly loam, 3 to 15 percent slopes	Farmland of statewide importance	
30	Dystric Xerochrepts, 60 to 90 percent slopes	Not prime farmland	
32	Everett very gravelly sandy loam, 0 to 8 percent slopes	Farmland of statewide importance	
33	Everett very gravelly sandy loam, 8 to 15 percent slopes	Farmland of statewide importance	
34	Everett very gravelly sandy loam, 15 to 30 percent slopes	Farmland of statewide importance	
35	Everett very gravelly sandy loam, 30 to 50 percent slopes	Not prime farmland	
38	Giles silt loam, 0 to 3 percent slopes	All areas are prime farmland	
39	Giles silt loam, 3 to 15 percent slopes	Farmland of statewide importance	
40	Giles silt loam, 15 to 30 percent slopes	Farmland of statewide importance	
41	Godfrey silty clay loam	Prime farmland if drained	
45	Hydraquents, tidal	Not prime farmland	
46	Indianola loamy sand, 0 to 5 percent slopes	Prime farmland if irrigated	
47	Indianola loamy sand, 5 to 15 percent slopes	Prime farmland if irrigated	
48	Indianola loamy sand, 15 to 30 percent slopes	Farmland of statewide importance	
51	Kapowsin silt loam, 3 to 15 percent slopes	Farmland of statewide importance	
52	Kapowsin silt loam, 15 to 30 percent slopes	Farmland of statewide importance	
53	Kapowsin silt loam, 30 to 50 percent slopes	Not prime farmland	
65	McKenna gravelly silt loam, 0 to 5 percent slopes	Prime farmland if drained	
69	Mukilteo muck	Prime farmland if drained	
70	Mukilteo muck, drained	Prime farmland if drained	
73	Nisqually loamy fine sand, 0 to 3 percent slopes	Prime farmland if irrigated	
74	Nisqually loamy fine sand, 3 to 15 percent slopes	Farmland of statewide importance	
75	Norma fine sandy loam	Prime farmland if drained	
76	Norma silt loam	Prime farmland if drained	
84	Pilchuck loamy sand	Prime farmland if irrigated	
85	Pits, gravel	Not prime farmland	
88	Puget silt loam	Prime farmland if drained	
89	Puyallup silt loam	All areas are prime farmland	
98	Salkum silty clay loam, 8 to 15 percent slopes	Farmland of statewide importance	
102	Schneider very gravelly loam, 20 to 40 percent slopes	Not prime farmland	
103	Schneider very gravelly loam, 40 to 65 percent slopes	Not prime farmland	
104	Semiahmoo muck	Prime farmland if drained	
106	Shalcar variant muck	Prime farmland if drained	
108	Skipopa silt loam, 3 to 15 percent slopes	Farmland of statewide importance	
109	Spana gravelly loam	Prime farmland if drained	

Prime and other Important Farmlands–Thurston County Area, Washington			
Map Symbol	Map Unit Name	Farmland Classification	
110	Spanaway gravelly sandy loam, 0 to 3 percent slopes	Prime farmland if irrigated	
115	Sultan silt loam	All areas are prime farmland	
120	Tisch silt loam	Prime farmland if drained	
125	Xerorthents, 0 to 5 percent slopes	Not prime farmland	
126	Yelm fine sandy loam, 0 to 3 percent slopes	All areas are prime farmland	
127	Yelm fine sandy loam, 3 to 15 percent slopes	Farmland of statewide importance	
128	Yelm fine sandy loam, 15 to 30 percent slopes	Farmland of statewide importance	
129	Water	Not prime farmland	

References

American Association of State Highway and Transportation Officials (AASHTO). 2004. Standard specifications for transportation materials and methods of sampling and testing. 24th edition.

American Society for Testing and Materials (ASTM). 2005. Standard classification of soils for engineering purposes. ASTM Standard D2487-00.

Cowardin, L.M., V. Carter, F.C. Golet, and E.T. LaRoe. 1979. Classification of wetlands and deep-water habitats of the United States. U.S. Fish and Wildlife Service FWS/OBS-79/31.

Federal Register. July 13, 1994. Changes in hydric soils of the United States.

Federal Register. September 18, 2002. Hydric soils of the United States.

Hurt, G.W., and L.M. Vasilas, editors. Version 6.0, 2006. Field indicators of hydric soils in the United States.

National Research Council. 1995. Wetlands: Characteristics and boundaries.

Soil Survey Division Staff. 1993. Soil survey manual. Soil Conservation Service. U.S. Department of Agriculture Handbook 18. http://www.nrcs.usda.gov/wps/portal/ nrcs/detail/national/soils/?cid=nrcs142p2_054262

Soil Survey Staff. 1999. Soil taxonomy: A basic system of soil classification for making and interpreting soil surveys. 2nd edition. Natural Resources Conservation Service, U.S. Department of Agriculture Handbook 436. http://www.nrcs.usda.gov/wps/portal/nrcs/detail/national/soils/?cid=nrcs142p2_053577

Soil Survey Staff. 2010. Keys to soil taxonomy. 11th edition. U.S. Department of Agriculture, Natural Resources Conservation Service. http://www.nrcs.usda.gov/wps/portal/nrcs/detail/national/soils/?cid=nrcs142p2 053580

Tiner, R.W., Jr. 1985. Wetlands of Delaware. U.S. Fish and Wildlife Service and Delaware Department of Natural Resources and Environmental Control, Wetlands Section.

United States Army Corps of Engineers, Environmental Laboratory. 1987. Corps of Engineers wetlands delineation manual. Waterways Experiment Station Technical Report Y-87-1.

United States Department of Agriculture, Natural Resources Conservation Service. National forestry manual. http://www.nrcs.usda.gov/wps/portal/nrcs/detail/soils/ home/?cid=nrcs142p2 053374

United States Department of Agriculture, Natural Resources Conservation Service. National range and pasture handbook. http://www.nrcs.usda.gov/wps/portal/nrcs/ detail/national/landuse/rangepasture/?cid=stelprdb1043084

United States Department of Agriculture, Natural Resources Conservation Service. National soil survey handbook, title 430-VI. http://www.nrcs.usda.gov/wps/portal/nrcs/detail/soils/scientists/?cid=nrcs142p2_054242

United States Department of Agriculture, Natural Resources Conservation Service. 2006. Land resource regions and major land resource areas of the United States, the Caribbean, and the Pacific Basin. U.S. Department of Agriculture Handbook 296. http://www.nrcs.usda.gov/wps/portal/nrcs/detail/national/soils/? cid=nrcs142p2_053624

United States Department of Agriculture, Soil Conservation Service. 1961. Land capability classification. U.S. Department of Agriculture Handbook 210. http://www.nrcs.usda.gov/Internet/FSE_DOCUMENTS/nrcs142p2_052290.pdf















DISCLAIMER: The City of Tumwater does not warrant, guarantee, or accept any liability for the accuracy, precision, or completeness of any information shown hereon or for any inferences made therefrom.

0 0.25 0.5

1 Miles